UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

\times	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT
	OF 1934

For the quarterly period ended September 30, 2021

	OR	
☐ TRANSITION REPORT PURSUANT TO SECTOF 1934	ΓΙΟΝ 13 OR 15(d)	OF THE SECURITIES EXCHANGE ACT
For the transition Commission Fi	period from ile Number 001-37368	to
ADAPTIMMUNE (Exact name of Regist	THERAPEU trant as specified in its of	
England and Wales (State or other jurisdiction of incorporation or organization)		Not Applicable (I.R.S. Employer Identification No.)
Abingdon, Ox Unit	venue, Milton Park (fordshire OX14 4RX) ed Kingdom ncipal executive offices)
	1235 430000 e number, including are	a code)
Securities registered pursuant to Section 12(b) of the Act:		
Title of each class	Trading Symbol	Name of each exchange on which registered
American Depositary Shares, each representing 6 Ordinary Shares, par value £0.001 per share	ADAP	The Nasdaq Global Select Market
Indicate by check mark whether the registrant (1) has filed all rep 1934 during the preceding 12 months (or for such shorter period that the requirements for the past 90 days. \boxtimes Yes \square No		
Indicate by check mark whether the registrant has submitted elec 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 files). \square Yes \square No		
Indicate by check mark whether the registrant is a large accelerate or an emerging growth company. See the definitions of "large accelerated growth company" in Rule 12b-2 of the Exchange Act.		
Large accelerated filer⊠ Non-accelerated filer⊓		Accelerated filer
Non-accelerated filer⊔		Smaller reporting company ☐ Emerging growth company☐

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).

Yes As of November 2, 2021, the number of outstanding ordinary shares par value £0.001 per share of the Registrant is937,224,360.

any new or revised financial accounting standard provided pursuant to Section 13(a) of the Exchange Act.

TABLE OF CONTENTS

PARTI —	- FINANCIAL INFORMATION	4
Item 1.	Financial Statements:	4
	Unaudited Condensed Consolidated Balance Sheets as of Setpember 30, 2021 and December 31, 2020	4
	Unaudited Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2021 and 2020	5
	<u>Unaudited Condensed Consolidated Statements of Comprehensive Loss for the three and nine months ended September 30, 2021 and 2020</u>	6
	<u>Unaudited Condensed Consolidated Statements of Change in Equity for the three and nine months ended September 30, 2021 and 2020</u>	7
	Unaudited Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2021 and 2020	9
	Notes to the Unaudited Condensed Consolidated Financial Statements	10
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	18
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	30
Item 4.	Controls and Procedures	31
PART II –	OTHER INFORMATION	31
Item 1.	<u>Legal Proceedings</u>	31
Item 1A.	Risk Factors	31
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	31
Item 3.	Defaults Upon Senior Securities	31
Item 4.	Mine Safety Disclosures	31
Item 5.	Other Information	31
Item 6.	<u>Exhibits</u>	32
Signatures		33

General information

In this Quarterly Report on Form 10-Q ("Quarterly Report"), "Adaptimmune," the "Group," the "Company," "we," "us" and "our" refer to Adaptimmune Therapeutics plc and its consolidated subsidiaries, except where the context otherwise requires.

Information Regarding Forward-Looking Statements

This Quarterly Report contains forward-looking statements that involve risks and uncertainties, as well as assumptions that, if they never materialize or prove incorrect, could cause our results to differ materially from those expressed or implied by such forward-looking statements. We make such forward-looking statements pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995 and other federal securities laws. All statements other than statements of historical facts contained in this Quarterly Report are forward-looking statements. In some cases, you can identify forward-looking statements by words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect" or the negative of these words or other comparable terminology.

Any forward-looking statements in this Quarterly Report reflect our current views with respect to future events or to our future financial performance and involve known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by these forward-looking statements. Factors that may cause actual results to differ materially from current expectations include, among other things, those listed under Part II, Item 1A. Risk Factors of this Quarterly Report and under Part I, Item 1A. Risk Factors of our Annual Report on Form 10-K for the year ended December 31, 2020. Given these uncertainties, you should not place undue reliance on these forward-looking statements. Except as required by law, we assume no obligation to update or revise these forward-looking statements for any reason, even if new information becomes available in the future.

This Quarterly Report also contains estimates, projections and other information concerning our industry, our business, and the markets for certain diseases, including data regarding the estimated size of those markets, and the incidence and prevalence of certain medical conditions. Information that is based on estimates, forecasts, projections, market research or similar methodologies is inherently subject to uncertainties and actual events or circumstances may differ materially from events and circumstances reflected in this information. Unless otherwise expressly stated, we obtained this industry, business, market and other data from reports, research surveys, studies and similar data prepared by third parties, industry, medical and general publications, government data and similar sources

PART I — FINANCIAL INFORMATION

Item 1. Financial Statements.

ADAPTIMMUNE THERAPEUTICS PLC UNAUDITED CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, except share data)

	Sep	September 30, 2021		ecember 31, 2020
Assets				
Current assets				
Cash and cash equivalents	\$	42,918	\$	56,882
Marketable securities - available-for-sale debt securities		197,202		311,335
Accounts receivable, net of allowance for doubtful accounts of \$0 and \$0		1,641		139
Other current assets and prepaid expenses		58,689		29,796
Total current assets		300,450		398,152
Restricted cash		1,717		4,602
Operating lease right-of-use assets, net of accumulated amortization		21,481		18,880
Property, plant and equipment, net of accumulated depreciation of \$35,087 (2020: \$31,097)		28,689		27,778
Intangibles, net of accumulated amortization		1,191		1,730
Total assets	\$	353,528	\$	451,142
Liabilities and stockholders' equity				
Current liabilities				
Accounts payable	\$	4,784	\$	6,389
Operating lease liabilities, current	·	2,267	•	2,773
Accrued expenses and other accrued liabilities		27,984		27,079
Deferred revenue, current		6,102		2,832
Total current liabilities		41,137		39,073
Operating lease liabilities, non-current		23,704		20,938
Deferred revenue, non-current		47,040		49,260
Other liabilities, non-current		664		644
Total liabilities		112,545		109,915
Stockholders' equity				
Common stock - Ordinary shares par value £0.001, 1,240,853,520 authorized and 937,049,820				
issued and outstanding (2020: 1,038,249,630 authorized and 928,754,958 issued and outstanding)		1.337		1.325
Additional paid in capital		954,732		935,706
Accumulated other comprehensive loss		(10,098)		(10,048)
Accumulated deficit		(704,988)		(585,756)
Total stockholders' equity		240,983		341,227
Total liabilities and stockholders' equity	\$	353,528	\$	451,142

ADAPTIMMUNE THERAPEUTICS PLC UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (In thousands, except share and per share data)

	Three months ended September 30,			Nine month Septembe				
		2021		2020		2021		2020
Revenue	\$	1,203	\$	1,193	\$	4,732	\$	2,456
Operating expenses								
Research and development		(28,211)		(24,067)		(81,585)		(65,791)
General and administrative		(15,173)		(13,001)		(42,529)		(32,557)
Total operating expenses		(43,384)		(37,068)		(124,114)		(98,348)
Operating loss		(42,181)		(35,875)		(119,382)		(95,892)
Interest income		225		2,147		916		4,024
Other income (expense), net		(237)		(1,689)		(184)		(1,501)
Loss before income taxes		(42,193)		(35,417)		(118,650)		(93,369)
Income taxes		(208)		(15)		(582)		(110)
Net loss attributable to ordinary shareholders	\$	(42,401)	\$	(35,432)	\$	(119,232)	\$	(93,479)
·								
Net loss per ordinary share								
Basic and diluted	\$	(0.05)	\$	(0.04)	\$	(0.13)	\$	(0.11)
Weighted average shares outstanding:								
Basic and diluted		936,600,648		928,022,057		933,992,708		829,973,177

ADAPTIMMUNE THERAPEUTICS PLC UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE LOSS (In thousands)

	 Three month		Nine mont Septemb	
	2021	2020	2021	2020
Net loss	\$ (42,401)	\$ (35,432)	\$ (119,232)	\$ (93,479)
Other comprehensive (loss) income, net of tax				
Foreign currency translation adjustments, net of tax of \$0, \$0, \$0 and \$0	15,564	(15,522)	8,386	3,583
Foreign currency gains (losses) on intercompany loan of a long-term investment nature,				
net of tax of \$0, \$0, \$0 and \$0	(15,310)	15,698	(8,351)	(5,061)
Unrealized holding gains (losses) on available-for-sale debt securities, net of tax of \$0,				
\$0, \$0 and \$0	33	211	(85)	324
Reclassification adjustment for gains on available-for-sale debt securities included in net				
loss, net of tax of \$0 and \$0	_	(76)		(76)
Total comprehensive loss for the period	\$ (42,114)	\$ (35,121)	\$ (119,282)	\$ (94,709)

ADAPTIMMUNE THERAPEUTICS PLC UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CHANGE IN EQUITY (In thousands, except share data)

	Common stock	Common stock	1	Additional paid in capital	ccumulated other mprehensive loss	A	ccumulated deficit	5	Total tockholders' equity
Balance as of 1 January 2021	928,754,958	\$ 1,325	\$	935,706	\$ (10,048)	\$	(585,756)	\$	341,227
Net loss	_	_		_	_		(37,763)		(37,763)
Other comprehensive loss	_	_		_	(176)		_		(176)
Issuance of shares upon exercise of stock options	4,062,210	6		529	_		_		535
Share-based compensation expense	_	_		5,334	_		_		5,334
Balance as of March 31, 2021	932,817,168	1,331		941,569	(10,224)		(623,519)		309,157
Net loss	_	_		_	_		(39,068)		(39,068)
Other comprehensive loss	_	_		_	(161)		_		(161)
Issuance of shares upon exercise of stock options	350,628	1		42	_		_		43
Issue of shares under At The Market sales agreement, net of									
expenses	3,069,330	4		2,515	_		_		2,519
Share-based compensation expense	_	_		5,449	_		_		5,449
Balance as of June 30, 2021	936,237,126	1,336		949,575	(10,385)		(662,587)		277,939
Net loss	_	_		_	_		(42,401)		(42,401)
Issuance of shares upon exercise of stock options	812,694	1		128	_				129
Issue of shares under At The Market sales agreement, net of									
expenses	_	_		10	_		_		10
Other comprehensive income	_	_		_	287		_		287
Share-based compensation expense	_	_		5,019	_		_		5,019
Balance as of September 30, 2021	937,049,820	\$ 1,337	\$	954,732	\$ (10,098)	\$	(704,988)	\$	240,983

ADAPTIMMUNE THERAPEUTICS PLC UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CHANGE IN EQUITY (In thousands, except share data)

	Common stock	Common stock	Additional paid in capital	co	other omprehensive loss) income	A	ccumulated deficit	Total stockholders' equity
Balance as of 1 January 2020	631,003,568	\$ 943	\$ 585,623	\$	(7,264)	\$	(455,664)	\$ 123,638
Net loss	_	_	_		_		(28,167)	(28,167)
Other comprehensive loss	_	_	_		(2,326)		_	(2,326)
Issuance of shares upon exercise of stock options	4,610,772	6	888		_		_	894
Issuance of shares upon completion of public offering, net of								
issuance costs	144,900,000	190	90,360		_		_	90,550
Share-based compensation expense	_	_	1,448		_		_	1,448
Balance as of March 31, 2020	780,514,340	1,139	678,319		(9,590)		(483,831)	186,037
Net loss	_	_	_				(29,880)	(29,880)
Other comprehensive income	_	_	_		785		_	785
Issuance of shares upon exercise of stock options	5,704,606	7	4,174		_		_	4,181
Issuance of shares upon completion of public offering, net of								
issuance costs	141,450,000	178	243,660		_		_	243,838
Share-based compensation expense	_	_	2,624		_		_	2,624
Balance as of June 30, 2020	927,668,946	1,324	928,777		(8,805)		(513,711)	407,585
Net loss					` _		(35,432)	(35,432)
Issuance of shares upon exercise of stock options	856,464	1	461		_		_	462
Other comprehensive loss		_	_		176		_	176
Share-based compensation expense	_	_	3,280		_		_	3,280
Balance as of September 30, 2020	928,525,410	\$ 1,325	\$ 932,518	\$	(8,494)	\$	(549,143)	\$ 376,206

ADAPTIMMUNE THERAPEUTICS PLC UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands)

		led		
		2021		2020
Cash flows from operating activities				
Net loss	\$	(119,232)	\$	(93,479)
Adjustments to reconcile net loss to net cash used in operating activities:				
Depreciation		4,333		5,151
Share-based compensation expense		15,802		7,352
Unrealized foreign exchange gains		(213)		(1,102)
Amortization on available-for-sale debt securities		4,094		2,798
Other		2,239		737
Changes in operating assets and liabilities:				
Increase/(decrease) in receivables and other operating assets		(31,809)		3,345
Decrease in non-current operating assets		_		2,291
Decrease in payables and other current liabilities		(109)		(117)
Increase in deferred revenue		1,696		48,649
Net cash used in operating activities		(123,199)		(24,375)
Cash flows from investing activities				
Acquisition of property, plant and equipment		(4,558)		(1,174)
Acquisition of intangibles		(181)		(496)
Maturity or redemption of marketable securities		190,393		78,915
Investment in marketable securities		(81,363)		(363,777)
Net cash provided by (used in) investing activities		104,291		(286,532)
Cash flows from financing activities				
Proceeds from issuance of common stock from offerings, net of commissions and issuance costs		2,529		334,388
Proceeds from exercise of stock options		707		5,541
Net cash provided by financing activities		3,236		339,929
Effect of currency exchange rate changes on cash, cash equivalents and restricted cash		(1,177)		(1,023)
Net (decrease) increase in cash, cash equivalents and restricted cash		(16,849)		27,999
Cash, cash equivalents and restricted cash at start of period		61,484		54,908
Cash, cash equivalents and restricted cash at end of period	\$	44,635	\$	82,907

ADAPTIMMUNE THERAPEUTICS PLC NOTES TO UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Note 1 — General

Adaptimmune Therapeutics plc is registered in England and Wales. Its registered office is 60 Jubilee Avenue, Milton Park, Abingdon, Oxfordshire, OX14 4RX, United Kingdom. Adaptimmune Therapeutics plc and its subsidiaries (collectively "Adaptimmune" or the "Company") is a clinical-stage biopharmaceutical company primarily focused on providing novel cell therapies to people with cancer. We are a leader in the development of T-cell therapies for solid tumors. The Company's proprietary platform enables it to identify cancer targets, find and develop cell therapy candidates active against those targets and produce therapeutic candidates for administration to patients.

The Company is subject to a number of risks similar to other biopharmaceutical companies in the early stage of clinical development including, but not limited to, the need to obtain adequate additional funding, possible failure of preclinical programs or clinical programs, the need to obtain marketing approval for its cell therapies, competitors developing new technological innovations, the need to successfully commercialize and gain market acceptance of its cell therapies, the need to develop a reliable commercial manufacturing process, the need to commercialize any cell therapies that may be approved for marketing, and protection of proprietary technology. If the Company does not successfully commercialize any of its cell therapies, it will be unable to generate product revenue or achieve profitability. The Company had an accumulated deficit of \$704,988,000 as of September 30, 2021.

Note 2 — Summary of Significant Accounting Policies

(a) Basis of presentation

The condensed consolidated financial statements of Adaptimmune Therapeutics plc and its subsidiaries and other financial information included in this Quarterly Report are unaudited and have been prepared in accordance with generally accepted accounting principles in the United States of America ("U.S. GAAP") and are presented in U.S. dollars. All significant intercompany accounts and transactions between the Company and its subsidiaries have been eliminated on consolidation.

The unaudited condensed consolidated financial statements presented in this Quarterly Report should be read in conjunction with the consolidated financial statements and accompanying notes included in the Company's Annual Report on Form 10-K filed with the SEC on February 25, 2021 (the "Annual Report"). The balance sheet as of December 31, 2020 was derived from audited consolidated financial statements included in the Company's Annual Report but does not include all disclosures required by U.S. GAAP. The Company's significant accounting policies are described in Note 2 to those consolidated financial statements.

Certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted from these interim financial statements. However, these interim financial statements include all adjustments, consisting only of normal recurring adjustments, which are, in the opinion of management, necessary to fairly state the results of the interim period. The interim results are not necessarily indicative of results to be expected for the full year.

(b) Use of estimates in interim financial statements

The preparation of interim financial statements, in conformity with U.S. GAAP and SEC regulations, requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosure of contingent assets and liabilities at the date of the interim financial statements and reported amounts of revenues and expenses during the reporting period. Estimates and assumptions are primarily made in relation to valuation allowances relating to deferred tax assets, revenue recognition, and estimation of the incremental borrowing rate for operating leases. If actual results differ from the Company's estimates, or to the extent these estimates are adjusted in future periods, the Company's results of operations could either benefit from, or be adversely affected by, any such change in estimate.

(c) Fair value measurements

The Company is required to disclose information on all assets and liabilities reported at fair value that enables an assessment of the inputs used in determining the reported fair values. The fair value hierarchy prioritizes valuation inputs based on the observable nature of those inputs. The hierarchy defines three levels of valuation inputs:

- Level 1 Quoted prices in active markets for identical assets or liabilities
- Level 2 Inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly
- Level 3 Unobservable inputs that reflect the Company's own assumptions about the assumptions market participants would use in pricing the asset or liability

The carrying amounts of the Company's cash and cash equivalents, restricted cash, accounts receivable, accounts payable and accrued expenses approximate fair value because of the short-term nature of these instruments. The fair value of marketable securities, which are measured at fair value on a recurring basis is detailed in Note 6, Fair value measurements.

(d) New accounting pronouncements

Recently adopted

Convertible instruments and contracts in an entity's own stock

On January 1, 2021, the Company adopted ASU 2020-06 -Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity (Subtopic 815-40)—Accounting for Convertible Instruments and Contracts in an Entity's Own Equity, which simplifies the accounting for convertible instruments and contracts in an entity's own stock. The guidance was adopted on a modified retrospective basis, whereby the guidance is applied to transactions outstanding as of the beginning of the fiscal year in which the guidance is adopted. The guidance has not had a material impact on the Company's condensed consolidated financial statements.

To be adopted in future periods

Measurement of credit losses on financial instruments

In June 2016, the FASB issued ASU 2016-13 - Financial Instruments - Credit losses, which replaces the incurred loss impairment methodology for financial instruments in current GAAP with a methodology that reflects expected credit losses and requires consideration of a broader range of reasonable and supportable information to inform credit loss estimates. The guidance is effective for the fiscal year beginning January 1, 2020, including interim periods within that fiscal year. In November 2019, the FASB issued ASU 2019-10 which resulted in the postponement of the effective date of the new guidance for eligible smaller reporting companies (as defined by the SEC), including the Company, at that time to the fiscal year beginning January 1, 2023; however, earlier adoption is permitted, and the Company may choose to implement the guidance in an earlier fiscal year. The guidance must be adopted using a modified-retrospective approach and a prospective transition approach is required for debt securities for which an other-than-temporary impairment had been recognized before the effective date. The Company is currently evaluating the impact of the guidance on its condensed consolidated financial statements.

Note 3 — Revenue

The Company has two contracts with customers: a collaboration and license agreement with GlaxoSmithKline ("GSK") and a collaboration agreement with Astellas.

Revenue comprises the following categories (in thousands):

	Three mo Septem	nths endo	ed		onths en ember 30	
	2021		2020	2021		2020
Development revenue	\$ 1,203	\$	1,193	\$ 4,732	\$	2,456
	\$ 1,203	\$	1,193	\$ 4,732	\$	2,456

The aggregate amount of the transaction price that is allocated to performance obligations that are unsatisfied or partially satisfied under the agreement as of September 30, 2021 was \$76,322,000. Of this amount, \$15,000,000 is allocated to the rights granted for each of the two independent Astellas targets, which will be recognised at a point-in-time upon commencement of the licenses in the event of nomination of the targets. The remaining amounts relate to our co-development with Astellas and GSK, which will be recognized as development progresses.

Future development, regulatory and sales milestones under both agreements are not considered probable as of September 30, 2021 and have not been included in the transaction price. Reimbursement of the research funding over the co-development period under the Astellas agreement is variable consideration and included in the transaction price as of September 30, 2021 to the extent that it is probable that a significant reversal in the amount of cumulative revenue recognized will not occur.

The Company received a milestone payment of \$4.2 million in the nine months ended September 30, 2021 following achievement of a development milestone for the third target under the GSK Collaboration and License Agreement. As a result of the inclusion of this amount in the transaction price, \$1,029,000 of revenue was recognized in the three and nine months ended September 30, 2021 from performance obligations partially satisfied in previous periods.

Of the revenue recognized in the nine months ended September 30, 2021, \$1,056,000 was included in the deferred income balance at January 1, 2021.

On September 3, 2021, the Company entered into a Strategic Collaboration and License Agreement with Genentech, Inc. and F. Hoffman-La Roche Ltd, which became effective on October 19, 2021 upon expiry or termination of all applicable waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976. The agreement is further described in Note 14.

Note 4 — Loss per share

The dilutive effect of 115,924,296 and 91,263,299 stock options outstanding as of September 30, 2021 and 2020 respectively have been excluded from the diluted loss per share calculation for the three and nine months ended September 30, 2021 and 2020 because they would have an antidilutive effect on the loss per share for the period.

Note 5 — Accumulated other comprehensive loss

The Company reports foreign currency translation adjustments and the foreign exchange gain or losses arising on the revaluation of intercompany loans of a long-term investment nature within Other comprehensive (loss) income. Unrealized gains and losses on available-for-sale debt securities are also reported within Other comprehensive (loss) income until a gain or loss is realized, at which point they are reclassified to Other (expense) income, net in the Condensed Consolidated Statement of Operations.

The following table shows the changes in Accumulated other comprehensive (loss) income (in thousands):

	tr	cumulated foreign currency anslation justments	Accum unrea gains (lo available debt sec	llized sses) on -for-sale	cor	Total ecumulated other nprehensive oss) income
Balance at January 1, 2021	\$	(10,158)	\$	110	\$	(10,048)
Foreign currency translation adjustments		(3,001)				(3,001)
Foreign currency gains on intercompany loan of a long-term investment nature, net of tax of \$\sqrt{9}\$		3,048		_		3,048
Unrealized holding losses on available-for-sale debt securities, net of tax of \$0				(223)		(223)
Balance at March 31, 2021		(10,111)		(113)		(10,224)
Foreign currency translation adjustments		(4,177)		`		(4,177)
Foreign currency gains on intercompany loan of a long-term investment nature, net of tax of \$\sqrt{9}\$		3,911		_		3,911
Unrealized holding gains on available-for-sale debt securities, net of tax of \$0		_		105		105
Balance at June 30, 2021		(10,377)		(8)		(10,385)
Foreign currency translation adjustments		15,564				15,564
Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$\sqrt{9}\$		(15,310)		_		(15,310)
Unrealized holding gains on available-for-sale debt securities, net of tax of \$0		_		33		33
Balance at September 30, 2021	\$	(10,123)	\$	25	\$	(10,098)
	tr ad	cumulated foreign currency anslation justments	Accum unrea gains (lo available debt sec	dized sses) on -for-sale curities	com (los	Total cumulated other prehensive ss) income
Balance at January 1, 2020	tr	foreign currency anslation justments	unrea gains (lo available	llized sses) on -for-sale	com	cumulated other aprehensive sss) income
Foreign currency translation adjustments	tr ad	foreign currency anslation justments (7,302) 17,911	unrea gains (lo available debt sec	dized sses) on -for-sale curities	com (los	cumulated other aprehensive sss) income (7,264) 17,911
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0	tr ad	foreign currency anslation justments	unrea gains (lo available debt sec	dized sses) onfor-sale curities 38 —	com (los	cumulated other aprehensive ss) income (7,264) 17,911 (19,651)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0	tr ad	foreign currency anslation justments (7,302) 17,911 (19,651)	unrea gains (lo available debt sec	dized sses) on -for-sale curities 38 - (586)	com (los	(7,264) 17,911 (19,651) (586)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020	tr ad	(7,302) 17,911 (19,651) (9,042)	unrea gains (lo available debt sec	38	com (los	cumulated other prehensive ss) income (7,264) 17,911 (19,651) (586) (9,590)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments	tr ad	(7,302) 17,911 (19,651) (9,042)	unrea gains (lo available debt sec	38	com (los	cumulated other prehensive sss) income (7,264) 17,911 (19,651) (586) (9,590) 1,194
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0	tr ad	(7,302) 17,911 (19,651) (9,042)	unrea gains (lo available debt sec	38	com (los	(7,264) 17,911 (19,651) (586) (9,590) 1,194 (1,108)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding gains on available-for-sale debt securities, net of tax of \$0	tr ad	(7,302) 17,911 (19,651) (9,042) 1,194 (1,108)	unrea gains (lo available debt sec	38	com (los	(7,264) 17,911 (19,651) (586) (9,590) 1,194 (1,108) 699
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding gains on available-for-sale debt securities, net of tax of \$0 Balance at June 30, 2020	tr ad	(7,302) 17,911 (19,651) (9,042) 1,194 (1,108) (8,956)	unrea gains (lo available debt sec	38	com (los	(7,264) 17,911 (19,651) (586) (9,590) 1,194 (1,108) (699 (8,805)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding gains on available-for-sale debt securities, net of tax of \$0 Balance at June 30, 2020 Foreign currency translation adjustments	tr ad	(7,302) 17,911 (19,651) (9,042) 1,194 (1,108) (8,956) (15,522)	unrea gains (lo available debt sec	38	com (los	(7,264) 17,911 (19,651) (586) (9,590) 1,194 (1,108) 699 (8,805) (15,522)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding gains on available-for-sale debt securities, net of tax of \$0 Balance at June 30, 2020	tr ad	(7,302) 17,911 (19,651) (9,042) 1,194 (1,108) (8,956)	unrea gains (lo available debt sec	38	com (los	(7,264) 17,911 (19,651) (586) (9,590) 1,194 (1,108) (699 (8,805)
Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding losses on available-for-sale debt securities, net of tax of \$0 Balance at March 31, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Unrealized holding gains on available-for-sale debt securities, net of tax of \$0 Balance at June 30, 2020 Foreign currency translation adjustments Foreign currency losses on intercompany loan of a long-term investment nature, net of tax of \$0 Reclassification from accumulated other comprehensive income of gains on available-for-sale debt	tr ad	(7,302) 17,911 (19,651) — (9,042) 1,194 (1,108) — (8,956) (15,522)	unrea gains (lo available debt sec	38	com (los	(7,264) 17,911 (19,651) (586) (9,590) 1,194 (1,108) 699 (8,805) (15,522) 15,698

Note 6 — Fair value measurements

Assets and liabilities measured at fair value on a recurring basis based on Level 1, Level 2, and Level 3 fair value measurement criteria as of September 30, 2021 are as follows (in thousands):

		Fair v	alue measurements	susing
	September 30, 2021	Level 1	Level 2	Level 3
Assets:		·	<u> </u>	
Corporate debt securities	\$ 197,202	\$ 197,202	_	_
	\$ 197,202	\$ 197,202	<u> </u>	\$ —

The Company estimates the fair value of available-for-sale debt securities with the aid of a third party valuation service, which uses actual trade and indicative prices sourced from third-party providers on a daily basis to estimate the fair value. If observed market prices are not available (for example securities with short maturities and infrequent secondary market trades), the securities are priced using a valuation model maximizing observable inputs, including market interest rates.

Note 7 — Marketable securities – available-for-sale debt securities

As of September 30, 2021, the Company has the following investments in marketable securities (in thousands):

	Remaining Contractual Maturity	Amortized Cost	Gross Unrealized Gains	Gross Unrealized Losses	Aggregate Estimated Fair Value
Available-for-sale debt securities:					
Corporate debt securities	Less than 3 months	\$ 9,228	\$ 5	\$ (1)	\$ 9,232
Corporate debt securities	3 months to 1 year	135,761	104	(29)	135,836
Corporate debt securities	1 year to 2 years	52,189	14	(69)	52,134
		\$ 197,178	\$ 123	\$ (99)	\$ 197,202

The aggregate fair value (in thousands) and number of securities held by the Company (including those classified as cash equivalents) in an unrealized loss position as of September 30, 2021 and 31 December, 2020 are as follows:

		September 30, 2021	1	December 31, 2020					
	Fair marke value of investments an unrealize loss position	investments n in an d unrealized	Unrealized losses	Fair market value of investments in an unrealized loss position	Number of investments in an unrealized loss position	Unrealized losses			
Marketable securities:									
Corporate debt securities	\$ 85,10	5 15	\$ (99)	\$ 157,985	30	\$ (158)			

As of September 30, 2021, the securities in an unrealized loss position are not considered to be other than temporarily impaired because the impairments are not severe and have been for a short duration. No securities have been in an unrealized loss position for more than one year. The Company does not intend to sell the debt securities in an unrealized loss position and believes that it has the ability to hold the debt securities to maturity.

Table of Contents

Note 8 — Other current assets

Other current assets consisted of the following (in thousands):

	Sep	tember 30, 2021	De	cember 31, 2020
Corporate tax receivable	\$	43,446	\$	20,585
Prepayments		9,970		6,314
Clinical materials		1,285		2,086
Other current assets		3,988		811
	\$	58,689	\$	29,796

Note 9 — Operating leases

The Company has operating leases in relation to property for office and research facilities

On August 13, 2021, the Company modified the lease of 60 Jubilee Avenue, Milton Park, Abingdon, Oxfordshire, UK (the "60 Jubilee Avenue lease") and on August 20, 2021, the Company modified the lease of 39 Innovation Drive, Milton Park, Abingdon, Oxfordshire, UK (the "39 Innovation Drive lease"). The effect of the modifications extended the first break option exercisable by the Company and has resulted in a change in the lease term for both leases. The modification to the 39 Innovation Drive lease also amended the lease payments for that lease. The modifications did not result in the identification of a separate contract.

Upon modification, the lease liability has been remeasured using the current estimate of the Company's incremental borrowing rate and the amount of the remeasurement of the lease liability has been recognized as an adjustment to the corresponding right-of-use asset. The effect of the modification was to increase the lease liability and the corresponding right-of-use asset by \$4,290,000.

The modification also removed a bank guarantee, which resulted in a reduction in restricted cash of \$2,739,000. The Company paid \$,736,000 to the lessor as a rent deposit.

The following table shows the weighted-average remaining lease term and the weighted-average discount rate as at September 30, 2021 and 2020:

	September 3	ου,
	2021	2020
Weighted-average remaining lease term - operating leases	8.0 years	6.5 years
Weighted-average discount rate - operating leases	6.8%	7.2%

The maturities of operating lease liabilities as of September 30, 2021 are as follows (in thousands):

	Opera	ting leases
2021	\$	978
2022		3,932
2023		4,050
2024		3,986
2025		4,034
after 2025		16,968
Total lease payments		33,948
Less: Imputed interest		7,977
Present value of lease liability	\$	25,971

The maximum lease term without activation of termination options is to 2041.

Note 10 — Accrued expenses and other current liabilities

Accrued expenses and other current liabilities consisted of the following (in thousands):

	Sep	tember 30, 2021		December 31, 2020
Accrued clinical and development expenditure	\$	13,661	\$	13,081
Accrued employee expenses		10,219		11,825
Other accrued expenditure		3,907		2,126
Other		197		47
	\$	27,984		27,079

Note 11 — Contingencies and commitments

On January 7, 2021, the Company entered into an agreement with a third party, whereby the third party is responsible for the development, manufacture, submission of regulatory filings and commercialization of a companion diagnostic for the detection of the MAGE-A4 biomarker. The Company shall compensate the third party for its performance of activities under the agreement based on milestone payments and reimbursement of direct expenses. The agreement is non-exclusive and the third party can sell the companion diagnostic to other parties. Once the companion diagnostic is approved and launched, the Company guarantees a minimum revenue to the third party. The agreement can be terminated by the Company and the third party upon 60 days' notice, if certain conditions are met.

Note 12 - Share-based compensation

The following table shows the total share-based compensation expense included in the unaudited consolidated statements of operations (in thousands):

	Three months ended					Nine m	onths ended			
		September 30,				September 30,				
		2021		2020	-	2021		2020		
Research and development	\$	2,177	\$	1,219	\$	7,064	\$	3,126		
General and administrative		2,842		2,061		8,738		4,226		
	\$	5,019	\$	3,280	\$	15,802	\$	7,352		

The following table shows information about share options and options which have a nominal exercise price (similar to restricted stock units (RSUs)) granted:

	Three n	ended		ended			
	Sept	30,	September 30,				
	2021 2020				2021		2020
Number of options over ordinary shares granted	4,170,230		1,882,966		19,891,334		14,851,182
Weighted average fair value of ordinary shares options	\$ 0.51	\$	1.25	\$	0.70	\$	0.57
Number of additional options with a nominal exercise price granted	1,348,920		571,320		16,008,168		7,410,136
Weighted average fair value of options with a nominal exercise price	\$ 0.70	\$	1.70	\$	0.98	\$	0.78

Note 13 - Stockholders' equity

On August 10, 2020 the Company entered into a sales agreement with Cowen and Company, LLC ("Cowen") (the "Sales Agreement") under which we may from time to time issue and sell American Depositary Shares ("ADSs") representing our ordinary shares through Cowen in at-the-market ("ATM") offerings for an aggregate offering price of up to \$200 million. In the three months ended June 30, 2021, the Company sold511,555 ADSs representing 3,069,330 ordinary shares resulting in net proceeds to the Company of \$2,519,000 after deducting commissions payable under the Sales Agreement and issuance costs. As of September 30, 2021, \$197,360,000 remained available for sale under the Sales Agreement.

Note 14 — Subsequent Events

On September 3, 2021, Adaptimmune Limited, a wholly owned subsidiary of Adaptimmune Therapeutics Plc, entered into a Strategic Collaboration and License Agreement (the "Agreement") with Genentech, Inc. ("Genentech") and F. Hoffman-La Roche Ltd.

Under the Agreement, Genentech and Adaptimmune (each, a "party" and together, the "parties") will collaborate to develop two types of allogeneic T-cell therapies: (i) "off-the-shelf" $\alpha\beta$ T-cell therapies directed to up to five collaboration targets and (ii) personalized therapies utilizing $\alpha\beta$ T-cell receptors (TCRs) isolated from a patient, with such therapies being administered to the same patient. The parties will collaborate to perform a research program, initially during an eight year period (which may be extended for up to two additional two year terms at Genentech's election upon payment of an extension fee for each two-year term), to develop the cell therapies, following which Genentech will determine whether to further develop and commercialize such therapies. Under the Agreement, Adaptimmune exclusively licenses Genentech certain intellectual property rights it controls to enable Genentech to research, develop, manufacture and commercialize (i) "off-the-shelf" T-cell therapies directed to the collaboration targets and (ii) personalized T-cell therapies developed within the scope of the Agreement, and Genentech is solely responsible for the clinical development and commercialization of any cell therapies arising from the collaboration. Adaptimmune will manufacture and supply cell therapies for Phase 1 trials of "off-the-shelf" T-cell therapies unless Genentech decides to assume responsibility for such manufacturing.

Under the Agreement, Adaptimmune is also subject to certain restrictions on its ability to further develop and commercialize certain cell therapies. In particular restrictions apply in relation to its ability to develop cell therapy products to nominated targets and to develop competing personalized cell therapies. This restriction does not prevent Adaptimmune from developing cell therapies to other targets or cell therapies containing different types of receptors.

Under the terms of the Agreement, Adaptimmune will receive \$150 million as an upfront payment, which is anticipated to be received in the fourth quarter of 2021. Adaptimmune may also receive:

- \$150 million in additional payments spread over a period of 5 years from the effective date of the Agreement, unless the agreement is earlier terminated;
- Research milestones of up to \$50 million:
- Development milestones of up to \$100 million in relation to the development of "off-the-shelf" T-cell therapies per collaboration target (unless
 Adaptimmune exercises its right to opt-in to receive a profit share) and up to \$200 million in relation to the development of personalized T-cell therapies;
- Commercialisation milestones of up to \$1.1 billion for off-the-shelf T-cell therapies (unless Adaptimmune exercises its right to opt-in to receive a profit share and assuming "off-the-shelf" T-cell therapies are developed to 5 targets) and for personalized T-cell therapies;
- Net sales milestones of up to \$1.5 billion for "off-the-shelf" T-cell therapies (unless Adaptimmune exercises its right to opt-in to receive a profit share and assuming "off-the-shelf" T-cell therapies are developed to 5 targets) and for personalized T-cell therapies.

In addition, Adaptimmune will receive tiered royalties on net sales in the mid-single to low-double digits.

Adaptimmune also has a right to opt-in to receive a profit share and to co-promote "off-the-shelf" T-cell therapies. If Adaptimmune elects to opt in, then Adaptimmune will be eligible to share 50 percent of profits and losses from U.S. sales on such products and to receive up to \$800 million in ex-U.S. regulatory and sales-based milestone payments, as well as royalties on ex-U.S. net sales.

The parties can terminate the Agreement in the event of material breach or insolvency of the other party. Genentech is entitled to terminate the Agreement in its entirety, on a product-by-product basis or collaboration target by collaboration target basis on provision of 180 days notice. Either party may terminate the Agreement on written notice in the event that the US Federal Trade Commission or US Department of Justice seeks a preliminary injunction under applicable antitrust laws against the parties or where HSR clearance has not occurred within 180 days of the effective date of the Agreement. The Agreement became effective on October 19, 2021 upon expiry of all applicable waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis of our financial condition and results of operations should be read in conjunction with our unaudited condensed consolidated financial statements and related notes included in this Quarterly Report. The following discussion and analysis of our financial condition and results of operations should be read in conjunction with our consolidated financial statements and related notes appearing elsewhere in this Quarterly Report and the audited consolidated financial statements and notes thereto and management's discussion and analysis of financial condition and results of operations for the year ended December 31, 2020, included in our Annual Report on Form 10-K that was filed with the SEC on February 25, 2021. Some of the information contained in this discussion and analysis or set forth elsewhere in this Quarterly Report, including information with respect to our plans and strategy for our business, includes forward-looking statements that involve risks and uncertainties. As a result of many factors, including those factors set forth in the "Risk Factors" section of this Quarterly Report and our Annual Report on Form 10-K for the year ended December 31, 2020, our actual results could differ materially from the results described in, or implied by, these forward-looking statements.

Overview

We are a clinical-stage biopharmaceutical company focused on providing novel cell therapies to people with cancer. We are a leader in the development of T-cell therapies for solid tumors and have reported clinical responses (per RECIST 1.1) in seven solid tumor indications.

Table of Contents

Our proprietary platform enables us to identify cancer targets, find and develop cell therapy candidates active against those targets and produce therapeutic candidates for administration to patients. Our cell therapy candidates include Specific Peptide Enhanced Affinity Receptor ("SPEAR") T-cells, which use genetically engineered T-cell receptors; next generation Tumor Infiltrating Lymphocytes ("TiLs") where a patient's own T-cells are co-administered with our next generation technology, and HLA-independent TCRs ("HiTs") where surface proteins are targeted independently of the peptide-HLA complex.

We have multiple clinical trials ongoing:

- SPEARHEAD-1 Phase 2 Trial with afamitresgene autoleucel ("afami-cel"): A registration directed Phase 2 clinical trial is underway in synovial sarcoma and myxoid round cell liposarcoma ("MRCLS") indications in which the MAGE-A4 antigen is expressed. Enrollment in Cohort 1 is complete, and Cohort 2 is currently recruiting. Initial data from Cohort 1 of this trial was presented at the American Society of Clinical Oncology (ASCO) on June 4, 2021. As presented at ASCO 2021, the Company reported an overall response rate for patients with at least one scan (evaluated by RECIST 1.1 per investigator assessment) of 39.3% (13 out of 33 patients), with an overall response rate of 41.4% (12/29 patients) for synovial sarcoma and 25.0% (1/4 patients) for MRCLS. Of the 29 patients with synovial sarcoma, the disease control rate was 86.2% (25/29 patients) (defined as either response or stable disease). Data from cohort 1 of this trial is intended to support the filing of a Biologics License Application (BLA) in 2022 and, upon approval from the U.S. Food and Drug Administration ("FDA"), the Company plans to commercially launch afami-cel. The EMA and the FDA have also agreed to the Company's pediatric investigational plans for afami-cel.
- SURPASS Phase 1 Trial with ADP-A2M4CD8: Enrollment is ongoing in a Phase 1 trial for our next generation SPEAR T-cells, ADP-A2M4CD8, focusing on treatment of patients with lung, gastroesophageal, head and neck, ovarian and bladder cancers in which the MAGE-A4 antigen is expressed. This next generation SPEAR T-cell utilizes the same engineered T-cell receptor as afami-cel, but with the addition of a CD8α homodimer. The addition of the CD8α homodimer has been shown in vitro to increase helper cell response and SPEAR T-cell potency. Data from the SURPASS trial was presented at the European Society for Medical Oncology (ESMO) meeting in September 2021. As of the August 2, 2021 data cut-off, initial efficacy and durability data were encouraging with an overall response rate of 36% and disease control rate of 86% in evaluable patients. ADP-A2M4CD8 was reported to be well tolerated in patients as of date of data cut-off. We will also evaluate ADP-A2M4CD8 in combination with a checkpoint inhibitor in patients with selected indications.
- SURPASS-2 Phase 2 Trial with ADP-A2M4CD8: Based on the responses seen in the Phase 1 clinical trial using afami-cel and initial responses seen in the SURPASS trial, a Phase 2 clinical trial with ADP-A2M4CD8 in esophageal and esophagogastric junction ("EGJ") cancers was initiated during the third quarter of 2021.
- SURPASS-3 Phase 2 Trial with ADP-A2M4CD8: Based on the initial responses seen in the SURPASS Phase 1 trial in patients with ovarian cancer, including a complete response which remained ongoing at the time of the data cut-off on August 2, 2021, we are planning to initiate a Phase 2 clinical trial with ADP-A2M4CD8 in ovarian cancer in 2022.
- ADP-A2AFP Phase 1 Trial: We continue treating patients in our Phase 1 trial designed to evaluate the safety and anti-tumor activity of our alpha fetoprotein ("AFP") specific therapeutic candidate for the treatment of hepatocellular carcinoma ("HCC"). A further cohort has also been initiated for patients with tumors other than HCC that express the AFP antigen. Data from the trial was presented at the International Liver Cancer Association (ILCA) meeting in September 2021. As of April 5, 2021, 13 patients with advanced HCC had received ADP-A2AFP in cohort 3 and expansion phases of the trial. The best overall responses per RECIST v 1.1 included 1 complete response (reported previously), 6 patients with stable disease and 4 patients with progressive disease. The disease control rate for patients with at least one scan was 64%. Given we are now approaching the original study objective of treating approximately 25 patients, we have ceased screening in the trial.
- SPEARHEAD-2 Phase 2 Trial with afami-cel: Adaptimmune is in the process of evaluating its strategy for the use of checkpoint inhibitors in combination with its cell therapies, including the evaluation of ADP-A2M4CD8 with a checkpoint inhibitor, and as a result we have decided to cease enrolment in the Spearhead-2 trial.

Table of Contents

We have an active preclinical pipeline of cell therapy candidates with the aim of delivering five new cell therapies to the clinic in the next five years. The pipeline includes new autologous SPEAR T-cells, SPEAR T-cells addressing alternative HLA-types, next generation SPEAR T-cells, next-generation TILs and HiTs. Preclinical data presented at the American Society of Gene & Cell Therapy (ASGCT) in May 2021 from the Company's HiT mesothelin program validated that human T-cells expressing a TCR targeting mesothelin, independent of HLA recognition, can kill human tumor cells in vitro; and showed that the HiT works as well, or better than, an in-house developed T-cell receptor fusion construct ("TRuC") targeting mesothelin in preclinical studies.

We are also developing allogeneic or "off-the-shelf" cell therapies utilizing a proprietary allogeneic platform.

- During Q3 2021 we announced a strategic collaboration with Genentech Inc ("Genentech") and F. Hoffman-La Roche Ltd.to research, develop, and commercialize allogeneic T-cell therapies (the "Genentech Collaboration"). The collaboration covers the research and development of "off-the-shelf" cell therapies for up to five shared cancer targets ("off the shelf" products) and the development of a novel allogeneic personalized cell therapy platform. The effectiveness of the agreement was subject to the expiration of the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, which has now expired. Under the terms of the agreement, we will receive an upfront payment of \$150 million and additional payments of \$150 million over five years, unless the agreement is earlier terminated. In addition, we may be eligible to receive research, development, regulatory and commercial milestones payments potentially exceeding \$3 billion in aggregate value. We will also receive tiered royalties on net sales in the mid-single to low-double digits. We have the right to opt in to a 50/50 U.S. profit/cost share on "off-the-shelf" products. If we elect to opt in, then we will be eligible to share 50 percent of profits and losses from U.S. sales on such products and are eligible to receive ex-U.S. regulatory and sales-based milestone payments, as well as royalties on ex-U.S. net sales.
- We also have a strategic collaboration program ongoing with Astellas (through its wholly owned subsidiary Universal Cells) in relation to up to three targets with the aim of co-developing T-cell therapy candidates directed to those targets and utilizing our allogeneic platform for "off-the-shelf" cell therapies. The first target subject to the collaboration is the mesothelin target to which a HiT cell therapy is being developed and a second target has been nominated by Astellas.

We also have a number of development and research collaborations including our collaboration with GSK for the development, manufacture and commercialization of TCR therapeutic candidates for up to five programs, a clinical and preclinical alliance agreement with MD Anderson Cancer Center and research collaborations with Alpine, Noile-Immune, and the Center for Cancer Immune Therapy (CCIT).

Financial Operations Overview

Revenue

The Company has three contracts with customers: the GSK Collaboration and License Agreement, the Astellas Collaboration Agreement and the Genentech Collaboration Agreement.

The GSK Collaboration Agreement

The GSK Collaboration and License Agreement consists of multiple performance obligations. GSK nominated its third target under the Collaboration and License Agreement in 2019, and the Company received \$3.2 million following the nomination of the target. The Company received a milestone payment of \$4.2 million in the nine months ended September 30, 2021 following achievement of a development milestone. These amounts are being recognized as revenue as development progresses.

The Astellas Collaboration Agreement

In January 2020, the Company entered into a collaboration agreement with Astellas. The Company received \$50.0 million as an upfront payment after entering into the agreement. Under the agreement the parties will agree on up to three targets and will co-develop T-cell therapies directed to those targets pursuant to an agreed research plan. For each target, Astellas will fund co-development up until completion of a Phase 1 trial for products directed to such target. In addition, Astellas was also granted the right to develop, independently of Adaptimmune, allogeneic T-cell therapy candidates directed to two targets selected by Astellas. Astellas will have sole rights to develop and commercialize products resulting from these two targets.

The agreement consists of the following performance obligations: (i) research services and rights granted under the co-exclusive license for each of the three co-development targets and (ii) the rights granted for each of the two independent Astellas targets. The revenue allocated to the co-development targets is recognized as the development of products directed to the targets progresses up until completion of a Phase 1 trial. The revenue allocated to each of the research licenses for the targets being independently developed by Astellas will be recognized when the associated license commences, which is upon designation of a target by Astellas.

The Genentech Collaboration Agreement

On September 3, 2021, the Company entered into the Genentech Collaboration Agreement, which became effective on October 19, 2021 upon expiry or termination of all applicable waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 and is fully described in Note 14 to the financial statements.

Research and Development Expenses

Research and development expenditures are expensed as incurred. Research and development expenses consist principally of the following:

- salaries for research and development staff and related expenses, including benefits;
- costs for production of preclinical compounds and drug substances by contract manufacturers;
- fees and other costs paid to contract research organizations in connection with additional preclinical testing and the performance of clinical trials;
- costs associated with the development of a process tomanufacture and supply our lentiviral vector and cell therapies for use in clinical trials;
- costs to develop manufacturing capability at our U.S. facility for manufacture of cell therapies for use in clinical trials;
- · costs relating to facilities, materials and equipment used in research and development;
- costs of acquired or in-licensed research and development which does not have alternative future use;
- costs of developing assays and diagnostics;
- an allocation of indirect costs clearly related to research and development;
- amortization and depreciation of property, plant and equipment and intangible assets used to develop our cells therapies; and
- share-based compensation expenses.

These expenses are partially offset by:

reimbursable tax and expenditure credits from the U.K. government.

As a company that carries out extensive research and development activities, we benefit from the U.K. research and development tax credit regime for small and medium sized companies ("SME R&D Tax Credit Scheme"), whereby our principal research subsidiary company, Adaptimmune Limited, is able to surrender the trading losses that arise from its research and development activities for a payable tax credit of up to approximately 33.4% of eligible research and development expenditures. Qualifying expenditures largely comprise employment costs for research staff, consumables and certain internal overhead costs incurred as part of research projects for which we do not receive income. Subcontracted research expenditures are eligible for a cash rebate of up to approximately 21.7%. A large proportion of costs in relation to our pipeline research, clinical trials management and manufacturing development activities, all of which are being carried out by Adaptimmune Limited, are eligible for inclusion within these tax credit cash rebate claims.

Expenditures incurred in conjunction with our collaboration agreements are not qualifying expenditures under the SME R&D Tax Credit Scheme but certain of these expenditures can be reimbursed through the U.K. research and development expenditure credit scheme (the "RDEC Scheme"). Under the RDEC Scheme tax relief is given at 13% of allowable R&D costs, which may result in a payable tax credit at an effective rate of approximately 10.5% of allowable R&D costs for the year ended December 31, 2021.

Our research and development expenses may vary substantially from period to period based on the timing of our research and development activities, which depends upon the timing of initiation of clinical trials and the rate of enrollment of patients in clinical trials. The duration, costs, and timing of clinical trials and development of our cell therapies will depend on a variety of factors, including:

- the scope, rate of progress, and expense of our ongoing as well as any additional clinical trials and other research and development activities;
- uncertainties in clinical trial enrollment rates;
- future clinical trial results;
- · significant and changing government regulation;
- the timing and receipt of any regulatory approvals; and
- supply and manufacture of lentiviral vector and cell therapies for clinical trials.

A change in the outcome of any of these variables may significantly change the costs and timing associated with the development of that SPEAR T-cell. For example, if the FDA, or another regulatory authority, requires us to conduct clinical trials beyond those that we currently anticipate will be required for regulatory approval, or if we experience significant delays in enrollment in any of our clinical trials, we could be required to expend significant additional financial resources and time on the completion of clinical development.

General and Administrative Expenses

Our general and administrative expenses consist principally of:

- salaries for employees other than research and development staff, including benefits;
- business development expenses, including travel expenses;
- professional fees for auditors, lawyers and other consulting expenses;
- costs of facilities, communication, and office expenses;
- cost of establishing commercial operations;
- · information technology expenses;
- amortization and depreciation of property, plant and equipment and intangible assets not related to research and development activities; and
- share-based compensation expenses.

Other (Expense) Income, Net

Other (expense) income, net primarily comprises foreign exchange (losses) gains. We are exposed to foreign exchange rate risk because we currently operate in the United Kingdom and United States. Our expenses are generally denominated in the currency in which our operations are located, which are the United Kingdom and United States. However, our U.K.-based subsidiary incurs significant research and development costs in U.S. dollars and, to a lesser extent, Euros. Our U.K. subsidiary has an intercompany loan balance in U.S. dollars payable to the ultimate parent company, Adaptimmune Therapeutics plc, which is considered of a long-term investment nature as repayment is not planned or anticipated in the foreseeable future. It is Adaptimmune Therapeutics plc's intent not to request payment of the intercompany loan for the foreseeable future. The foreign exchange gains or losses arising on the revaluation of intercompany loans of a long-term investment nature are reported within other comprehensive (loss) income, net of tax.

Our results of operations and cash flows will be subject to fluctuations due to changes in foreign currency exchange rates, which could harm our business in the future. We seek to minimize this exposure by maintaining currency cash balances at levels appropriate to meet forthcoming expenditure in U.S. dollars and pounds sterling. To date, we have not used hedging contracts to manage exchange rate exposure, although we may do so in the future.

Taxation

We are subject to corporate taxation in the United Kingdom and the United States. We incur tax losses and tax credit carryforwards in the United Kingdom. No deferred tax assets are recognized on our U.K. losses and tax credit carryforwards because there is currently no indication that we will make sufficient taxable profits to utilize these tax losses and tax credit carryforwards. On June 10, 2021, the U.K. 2021 Finance Bill received Royal Assent. Under this bill, the rate of U.K. corporation tax will increase to 25% in 2023, with lower rates and tapered relief to be applied to companies with profits below £250,000.

We benefit from reimbursable tax credits in the United Kingdom through the SME R&D Tax Credit Scheme as well as the RDEC Scheme which are presented as a deduction to research and development expenditure.

Our subsidiary in the United States has generated taxable profits due to a Service Agreement between our U.S. and U.K. operating subsidiaries and is subject to U.S. federal corporate income tax of 21%. Due to its activity in the United States, and the sourcing of its revenue, the U.S. subsidiary is not currently subject to any state or local income taxes. The Company also benefits from the U.S Research Tax Credit and Orphan Drug Credit.

In the future, if we generate taxable income in the United Kingdom, we may benefit from the United Kingdom's "patent box" regime, which would allow certain profits attributable to revenues from patented products to be taxed at a rate of 10%. As we have many different patents covering our products, future upfront fees, milestone fees, product revenues, and royalties may be taxed at this favorably low tax rate.

U.K. Value Added Tax ("VAT") is charged on all qualifying goods and services by VAT-registered businesses. An amount of 20% of the value of the goods or services is added to all relevant sales invoices and is payable to the U.K. tax authorities. Similarly, VAT paid on purchase invoices paid by Adaptimmune Limited and Adaptimmune Therapeutics plc is reclaimable from the U.K. tax authorities.

Critical Accounting Policies and Significant Judgments and Estimates

The preparation of our unaudited condensed consolidated financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosure of contingent assets and liabilities, and the revenues and expenses incurred during the reported periods. We base our estimates on historical experience and on various other factors that we believe are relevant under the circumstances, the results of which form the basis for making judgments about the carrying value of assets and liabilities that are not apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions. The accounting policies considered to be critical to the judgments and estimates used in the preparation of our financial statements are disclosed in the Management's Discussion and Analysis of Financial Condition and Results of Operations included in our Annual Report on Form 10-K for the year ended December 31, 2020.

Results of Operations

Comparison of Three Months Ended September 30, 2021 and 2020

The following table summarizes the results of our operations for the three months ended September 30, 2021 and 2020, together with the changes to those items (in thousands):

	Three months ended September 30,						
	2021			2020		Increase/do	ecrease
Revenue	\$	1,203	\$	1,193	\$	10	1 %
Research and development expenses		(28,211)		(24,067)		(4,144)	17 %
General and administrative expenses		(15,173)		(13,001)		(2,172)	<u>17</u> %
Total operating expenses		(43,384)		(37,068)		(6,316)	17 %
Operating loss		(42,181)		(35,875)		(6,306)	18 %
Interest income		225		2,147		(1,922)	(90)%
Other expense, net		(237)		(1,689)		1,452	(86)%
Loss before income taxes		(42,193)		(35,417)		(6,776)	19 %
Income taxes		(208)		(15)		(193)	1,287 %
Loss for the period	\$	(42,401)	\$	(35,432)	\$	(6,969)	20 %

Revenue

Revenue was \$1.2 million in the three months ended September 30, 2021 compared to \$1.2 million for the three months ended September 30, 2020. Revenue varies depending on the progress of development activities and the amounts we expect to receive under contracts with customers.

Research and Development Expenses

Research and development expenses increased by 17% to \$28.2 million for the three months ended September 30, 2021 from \$24.1 million for the three months ended September 30, 2020.

Our research and development expenses comprise the following (in thousands):

	Three mo Septem				
	2021 2020			Increase/o	ecrease
Salaries, materials, equipment, depreciation of property, plant and equipment and other employee-related					_
$costs^{(1)}$	\$ 20,270	\$	15,901	\$ 4,369	27 %
Subcontracted expenditure	10,111		9,636	475	5 %
Manufacturing facility expenditure	2,594		2,079	515	25 %
Share-based compensation expense	2,177		1,219	958	79 %
Reimbursements receivable for research and development tax and expenditure credits	(6,941)		(4,768)	(2,173)	46 %
	\$ 28,211	\$	24,067	\$ 4,144	17 %

(1) These costs are not analyzed by project since employees may be engaged in multiple projects simultaneously.

The net increase in our research and development expenses of \$4.1 million for the three months ended September 30, 2021 compared to the same period in 2020 was primarily due to the following:

- an increase of \$4.4 million in salaries, materials, equipment, depreciation of property, plant and equipment and other employee-related costs, which
 is mainly driven by an increase in the average number of employees engaged in research and development;
- an increase of \$1.0 million in share-based compensation expense primarily due to higher option grants in 2021 because of an increase in the number of employees and higher fair value of options being expensed; and
- an increase of \$2.2 million in reimbursements receivable for research and development tax and expenditure credits, which is driven by an increase in qualifying costs identified.

Our subcontracted costs for the three months ended September 30, 2021 were \$10.1 million, compared to \$9.6 million in the same period of 2020. This includes \$7.7 million of costs directly associated with our afami-cel, ADP-A2M4CD8 and ADP-A2AFP SPEAR T-cells and \$2.4 million of other development costs.

Our research and development expenses are highly dependent on the phases and progression of our research projects and will fluctuate depending on the outcome of ongoing clinical trials. We expect that our research and development expenses will increase in future periods as we continue to invest in our translational sciences and other research and development capabilities.

General and Administrative Expenses

General and administrative expenses increased by 17% to \$15.2 million for the three months ended September 30, 2021 from \$13.0 million in the same period in 2020. The net increase in our general and administrative expenses of \$2.2 million for the three months ended September 30, 2021 compared to the same period in 2020 was largely due to:

- an increase of \$0.9 million in salaries, depreciation of property, plant and equipment and other employee-related costs, as a result of an increase
 in the average number of employees in the three months ended September 30, 2021 compared to the same period in 2020; and
- an increase of \$0.8 million in share based compensation, because of an increase in the number of employees and higher fair value of options being expensed.

We expect that our general and administrative expenses will increase in the future as we expand our operations and move towards commercial launch.

Income Taxes

Income taxes increased to a charge \$208,000 for the three months ended September 30, 2021 from a charge of \$15,000 for the three months ended September 30, 2020. Income taxes arise in the United States due to our U.S. subsidiary generating taxable profits. We incur losses in the United Kingdom.

Comparison of Nine Months Ended September 30, 2021 and 2020

The following table summarizes the results of our operations for the nine months ended September 30, 2021 and 2020, together with the changes to those items (in thousands):

	Nine months ended								
	Se	ptembe	er 30,						
	2021 2020				Increase/o	lecrease			
Revenue	\$ 4,7	32	\$ 2,456	\$	2,276	93 %			
Research and development expenses	(81,5	85)	(65,791)		(15,794)	24 %			
General and administrative expenses	(42,5	29)	(32,557)		(9,972)	31 %			
Total operating expenses	(124,1	14)	(98,348)		(25,766)	26 %			
Operating loss	(119,3	82)	(95,892)		(23,490)	24 %			
Interest income	9	16	4,024		(3,108)	(77)%			
Other (expense) income, net	(1	84)	(1,501)		1,317	(88)%			
Loss before income taxes	(118,6	50)	(93,369)		(25,281)	27 %			
Income taxes	(5	82)	(110)		(472)	429 %			
Loss for the period	\$ (119,2	32)	\$ (93,479)	\$	(25,753)	28 %			

Revenue

Revenue was \$4.7 million in the nine months ended September 30, 2021 compared to \$2.5 million for the nine months ended September 30, 2020. Revenue has increased primarily due to an increase in development activities under our collaboration agreements. Revenue varies depending on the progress of development activities and the amounts we expect to receive under contracts with customers (to the extent that it is probable that a significant reversal in the amount of cumulative revenue recognized will not occur).

Research and Development Expenses

Research and development expenses increased by 24% to \$81.6 million for the nine months ended September 30, 2021 from \$65.8 million for the nine months ended September 30, 2020.

Our research and development expenses comprise the following (in thousands):

		Nine mor Septem					
	2021 2020				Increase/d	ecrease	
Salaries, materials, equipment, depreciation of property, plant and equipment and other employee-related							
$costs^{(1)}$	\$	58,694	\$	46,192	\$ 12,502	27 %	
Subcontracted expenditure		32,353		24,234	8,119	34 %	
Manufacturing facility expenditure		7,135		5,133	2,002	39 %	
Share-based compensation expense		7,064		3,126	3,938	126 %	
In-process research and development costs		151		784	(633)	(81)%	
Reimbursements receivable for research and development tax and expenditure credits		(23,812) (13,678)			(10,134)	74 %	
	\$	81,585	\$	65,791	\$ 15,794	24 %	

(1) These costs are not analyzed by project since employees may be engaged in multiple projects simultaneously.

The net increase in our research and development expenses of \$15.8 million for the nine months ended September 30, 2021 compared to the same period in 2020 was primarily due to the following:

- an increase of \$12.5 million in salaries, materials, equipment, depreciation of property, plant and equipment and other employee-related costs, which is mainly driven by an increase in the average number of employees engaged in research and development;
- an increase of \$8.1 million in subcontracted expenditure due to increases in costs related to the development of a companion diagnostic assay and an increase in clinical trial costs as we prepare for a Phase 2 clinical trial with ADP-A2M4CD8 in Esopogeal Gastric Junction (EGJ) cancers
- an increase of \$3.9 million in share-based compensation expense primarily because of an increase in the number of employees and higher fair value of options being expensed; and
- an increase of \$10.1 million in reimbursements receivable for research and development tax and expenditure credits, which is driven by an increase in qualifying costs identified.

Our subcontracted costs for the nine months ended September 30, 2021 were \$32.4 million, compared to \$24.2 million in the same period of 2020. This includes \$24.6 million of costs directly associated with our afami-cel, ADP-A2M4CD8 and ADP-A2AFP SPEAR T-cells and \$7.7 million of other development costs.

Our research and development expenses are highly dependent on the phases and progression of our research projects and will fluctuate depending on the outcome of ongoing clinical trials. We expect that our research and development expenses will increase in future periods as we continue to invest in our translational sciences and other research and development capabilities.

General and Administrative Expenses

General and administrative expenses increased by 31% to \$42.5million for the nine months ended September 30, 2021 from \$32.5 million in the same period in 2020. Our general and administrative expenses consist of the following (in thousands):

	Nine months ended September 30,						
	2021		2020		Increase/decrease		ase
Salaries, depreciation of property, plant and equipment and other		<u>.</u>					
employee-related costs	\$	21,434	\$	17,992	\$	3,442	19 %
Other corporate costs		13,810		11,535		2,275	20 %
Share-based compensation expense		8,738		4,226		4,512	107 %
Reimbursements		(1,453)		(1,196)		(257)	21 %
	\$	42,529	\$	32,557	\$	9,972	31 %

The net increase in our general and administrative expenses of \$10.0 million for the nine months ended September 30, 2021 compared to the same period in 2020 was largely due to:

- an increase of \$3.4 million in salaries, depreciation of property, plant and equipment and other employee-related costs, as a result of an increase in the average number of employees in the nine months ended September 30, 2021 compared to the same period in 2020;
- an increase of \$2.3 million in other corporate costs, primarily due to increases in employee-related costs and professional fees in the nine months ended September 30, 2021 compared to the same period in 2020; and
- an increase of \$4.5 million in share based compensation, due to option forfeits in the nine months ended September 30, 2020, and because of an increase in the number of employees and higher fair value of options being expensed in the nine months ended September 30, 2021.

We expect that our general and administrative expenses will increase in the future as we expand our operations and move towards commercial launch.

Income Taxes

Income taxes increased to a charge of \$582,000 for the nine months ended September 30, 2021 from a charge of \$110,000 for the nine months ended September 30, 2020. Income taxes arise in the United States due to our U.S. subsidiary generating taxable profits. We incur losses in the United Kingdom.

Liquidity and Capital Resources

Sources of Funds

Since our inception, we have incurred significant net losses and negative cash flows from operations. We financed our operations primarily through sales of equity securities, cash receipts under our collaboration arrangements and research and development tax and expenditure credits. From inception through to September 30, 2021, we have raised:

- \$857.1 million, net of issuance costs, through the issuance of shares;
- \$208.0 million through collaborative arrangements with GSK and Astellas; and
- \$59.2 million in the form of reimbursable U.K. research and development tax credits and receipts from the U.K. RDEC Scheme.

We use a non-GAAP measure, Total Liquidity, which is defined as the total of cash and cash equivalents and marketable securities, to evaluate the funds available to us in the near-term. A description of Total Liquidity and reconciliation to cash and cash equivalents, the most directly comparable U.S. GAAP measure, are provided below under "Non-GAAP measures".

As of September 30, 2021, we had cash and cash equivalents of \$42.9 million and Total Liquidity of \$240.1 million. In addition, under the terms of the Agreement with Genentech, Adaptimmune is entitled to receive \$150 million as an upfront payment, which is anticipated to be received in the fourth quarter of 2021. We regularly assess Total Liquidity against our activities and make decisions regarding prioritization of those activities and deployment of Total Liquidity. We believe that our Total Liquidity, together with the upfront and exclusivity payments under the Strategic Collaboration and License Agreement with Genentech, will fund the Company's current operations based upon our currently anticipated research and development activities, planned capital spending, and planned commercialization costs into early 2024. This belief is based on estimates that are subject to risks and uncertainties and may change if actual results differ from management's estimates.

Cash Flows

The following table summarizes the results of our cash flows for the nine months ended September 30, 2021 and 2020 (in thousands).

	Nine months ended		
	 September 30,		
	 2021		2020
Net cash used in operating activities	\$ (123,199)	\$	(24,375)
Net cash provided by (used in) investing activities	104,291		(286,532)
Net cash provided by financing activities	3,236		339,929
Cash, cash equivalents and restricted cash	44,635		82,907

Operating Activities

Net cash used in operating activities was \$123.2 million for the nine months ended September 30, 2021 compared to \$24.3 million for the nine months ended September 30, 2020. The receipt of the \$50.0 million upfront payment from Astellas in January 2020 and the receipt of the R&D tax credit of \$18.7m in July 2020, resulted in lower net cash used in operating activities for the nine months ended September 30, 2020. Excluding the impact of this, the net cash used in operating activities for the nine months ended September 30, 2021 increased due to an increase in operating expenditure.

Net cash used in operating activities of \$123.2 million for the nine months ended September 30, 2021 comprised a net loss of \$119.2 million and a net cash outflow of \$30.2 million from changes in operating assets and liabilities, offset by non-cash items of \$26.2 million. The changes in operating assets and liabilities include the impact of a \$16.8 million increase in reimbursements receivable for research and development tax credits. The non-cash items consisted of depreciation expense on plant and equipment of \$4.3 million, share-based compensation expense of \$15.8 million, amortization on available-for-sale debt securities of \$4.1 million, and other items of \$2.2 million, which were offset by unrealized foreign exchange gains of \$0.2 million.

Investing Activities

Net cash provided by investing activities was \$104.3 million for the nine months ended September 30, 2021 compared to net cash used in investing activities of \$286.5 million for the nine months ended September 30, 2020. The net cash provided by (used in) investing activities for the respective periods consisted primarily of:

• purchases of property and equipment of \$4.6 million and \$1.2 million for the nine months ended September 30, 2021 and 2020, respectively;

• cash outflows from investment in marketable securities of \$81.4 million and \$363.8 million for the nine months ended September 30, 2021 and 2020, respectively, and cash inflows from maturity or redemption of marketable securities of \$190.4 million and \$78.9 million for the nine months ended September 30, 2021 and 2020, respectively.

The Company invests surplus cash and cash equivalents in marketable securities. In the nine months ended September 30, 2021, the investments in marketable securities were reduced to fund the Company's ongoing operations. In the nine months ended September 30, 2020, the Company increased its investments in marketable securities with proceeds from its public offerings.

Financing Activities

Net cash provided by financing activities was \$3.2 million and \$339.9 million for the nine months ended September 30, 2021 and 2020, respectively. The net cash provided by financing activities in the nine months ended September 30, 2021 consisted of net proceeds of \$2.5 million from shares issued in an atthe-market offering, net of commissions and issuance costs, and proceeds of \$0.7 million from share option exercises. For the nine months ended September 30, 2020, the net cash provided by financing activities consisted of net proceeds from public offerings of \$334.4 million and proceeds from share option exercises of \$5.5 million.

Non-GAAP Measures

Total Liquidity (a non-GAAP financial measure)

Total Liquidity (a non-GAAP financial measure) is the total of cash and cash equivalents and marketable securities. Each of these components appears in the condensed consolidated balance sheet. The U.S. GAAP financial measure most directly comparable to Total Liquidity is cash and cash equivalents as reported in the condensed consolidated financial statements, which reconciles to Total Liquidity as follows (in thousands):

	Ser	September 30,		December 31,	
		2021		2020	
Cash and cash equivalents	\$	42,918	\$	56,882	
Marketable securities - available-for-sale debt securities		197,202		311,335	
Total Liquidity	\$	240,120	\$	368,217	

We believe that the presentation of Total Liquidity provides useful information to investors because management reviews Total Liquidity as part of its management of overall liquidity, financial flexibility, capital structure and leverage. The definition of Total Liquidity includes investments, which are highly-liquid and available to use in our current operations, such as marketable securities.

Safe Harbor

See the section titled "Information Regarding Forward-Looking Statements" at the beginning of this Quarterly Report.

Item 3. Quantitative and Qualitative Disclosures about Market Risk.

There have been no material changes to the Company's market risk during the three months ended September 30, 2021. For a discussion of the Company's exposure to market risk, please refer to the Company's market risk disclosures set forth in Part II, Item 7A, "Quantitative and Qualitative Disclosures About Market Risk" in our Annual Report on Form 10-K for the year ended December 31, 2020.

Item 4. Controls and Procedures.

Evaluation of Disclosure Controls and Procedures

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e)) under the Securities and Exchange Act of 1934, as amended ("Exchange Act") as of September 30, 2021. Based on such evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that, as of September 30, 2021, our disclosure controls and procedures were effective to ensure that information required to be disclosed by us in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms, and is accumulated and communicated to our management, including our Chief Executive and Chief Financial Officer, or persons performing similar functions, as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Control over Financial Reporting

No changes in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) occurred during the quarter ended September 30, 2021 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings.

As of September 30, 2021, we were not a party to any material legal proceedings.

Item 1A. Risk Factors.

Our business has significant risks. You should carefully consider the risk factors set out in Part I, Item 1A "Risk Factors" of our Annual Report on Form 10-K for the year ended December 31, 2020 and the disclosures set out in this Quarterly Report, including our condensed consolidated financial statements and the related notes, before making an investment decision regarding our securities. The risks and uncertainties described are those significant or material risk factors currently known and specific to us that we believe are relevant to our business, results of operations and financial condition. Additional risks and uncertainties not currently known to us or that we now deem immaterial may also impair our business, results of operations and financial condition.

As of and for the period ended September 30, 2021, there have been no material changes from the risk factors previously disclosed by us in Part I, Item 1A. Risk Factors of our Annual Report on Form 10-K for the year ended December 31, 2020.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

None.

Item 3. Defaults Upon Senior Securities.

None.

Item 4. Mine Safety Disclosures.

Not applicable.

Item 5. Other Information.

None.

Item 6. Exhibits.

The following exhibits are either provided with this Quarterly Report on Form 10-Q or are incorporated herein by reference:

Exhibit Number	Description of Exhibit
10.1†**	Strategic Collaboration and License Agreement among Adaptimmune Limited, on the one hand, and Genentech, Inc. and F. Hoffman-La Roche Ltd, on the other hand, made and entered into as of September 3, 2021.
31.1**	Certification of Principal Executive Officer pursuant to Rule 13a-14(a) and Rule 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2**	Certification of Principal Financial Officer pursuant to Rule 13a-14(a) and Rule 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1**	Certification of Principal Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2**	Certification of Principal Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101**	The following financial information from Adaptimmune Therapeutics plc's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2021, formatted in iXBRL (Inline eXtensible Business Reporting Language): (i) Unaudited Condensed Consolidated Balance Sheets as of September, 2021 and December 31, 2020, (ii) Unaudited Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2021 and 2020, (iii) Unaudited Condensed Consolidated Statements of Comprehensive (Loss) Income for the three and nine months ended September 30, 2021 and 2020, (iv) Unaudited Condensed Consolidated Statements of Change in Equity for the three and nine months ended September 30, 2021 and 2020, (v) Unaudited Condensed Consolidated Statements of Cash Flows for nine months ended September 30, 2021 and 2020 and (vi) Notes to the Unaudited Condensed Consolidated Financial Statements.
104**	Cover Page Interactive data File (formatted in Inline XBRL and contained in Exhibit 101)

Previously filed.

^{**} Filed herewith.

[†] Certain portions of this exhibit have been omitted because they are not material and they are the type of information that the Registrant treats as private or confidential.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

ADAPTIMMUNE THERAPEUTICS PLC

Date: November 4, 2021 /s/ Adrian Rawcliffe

Adrian Rawcliffe Chief Executive Officer

/s/ Gavin Wood Date: November 4, 2021

Gavin Wood Chief Financial Officer

CERTAIN IDENTIFIED INFORMATION CONTAINED IN THIS EXHIBIT, MARKED BY [***], HAS BEEN EXCLUDED FROM THIS EXHIBIT BECAUSE IT IS BOTH NOT MATERIAL AND IS THE TYPE THAT THE REGISTRANT TREATS AS PRIVATE OR CONFIDENTIAL.

STRATEGIC COLLABORATION AND LICENSE AGREEMENT

AMONG

ADAPTIMMUNE LIMITED,

on the one hand,

AND

GENENTECH, INC.

AND

F. HOFFMANN-LA ROCHE LTD,

on the other hand

AS OF SEPTEMBER 3, 2021

CONFIDENTIAL

TABLE OF CONTENTS

ARTICLE 1	<u>DEFINITIONS</u>	<u>1</u>
ARTICLE 2	<u>GOVERNANCE</u>	<u>28</u>
<u>2.1.</u>	Joint Research Committee	<u>28</u>
<u>2.2.</u>	Joint Project Committees	<u>29</u>
<u>2.3.</u>	JIPC	<u>30</u>
<u>2.4.</u>	Joint Development Committee	<u>30</u>
<u>2.5.</u>	Meetings	<u>30</u>
<u>2.6.</u>	Dissolution of the JPC and JRC; Functionally Distinct Determinations	<u>32</u>
<u>2.7.</u>	Alliance Managers	<u>32</u>
<u>2.8.</u>	<u>Limitations on Authority</u>	<u>32</u>
<u>2.9.</u>	<u>Escalation</u>	<u>33</u>
<u>2.10.</u>	Final Resolution	<u>33</u>
ARTICLE 3	RESEARCH PROGRAM	<u>33</u>
<u>3.1.</u>	Research Program	<u>33</u>
<u>3.2.</u>	Research Plan	<u>33</u>
<u>3.3.</u>	Research Term	<u>34</u>
<u>3.4.</u>	Conduct of Research Program	<u>34</u>
<u>3.5.</u>	Cell Line Development	<u>35</u>
<u>3.6.</u>	Nomination	<u>35</u>
<u>3.7.</u>	[***]	<u>44</u>
<u>3.8.</u>	Subcontractors	<u>36</u>
<u>3.9.</u>	Reports; Records	<u>36</u>
<u>3.10.</u>	Research Efforts	<u>37</u>
ARTICLE 4	<u>DEVELOPMENT</u>	<u>37</u>
<u>4.1.</u>	<u>General</u>	<u>37</u>
<u>4.2.</u>	Development Efforts	<u>37</u>
<u>4.3.</u>	Progress Reports	<u>37</u>
<u>4.4.</u>	Development Funding	<u>38</u>
<u>4.5.</u>	Regulatory	<u>38</u>
<u>4.6.</u>	Safety Data Exchange	<u>38</u>
<u>4.7.</u>	[***]	<u>47</u>
ARTICLE 5	COMMERCIALIZATION	<u>38</u>

<u>5.1.</u>	General	<u>38</u>
<u>5.2.</u>	Commercialization Efforts	38
<u>5.3.</u>	<u>Licensed Products</u>	<u>39</u>
ARTICLE 6	ADAPTIMMUNE OPT-IN RIGHT	39
<u>6.1.</u>	Adaptimmune Opt-In	<u>39</u>
ARTICLE 7	<u>LICENSES</u>	<u>43</u>
<u>7.1.</u>	License from Adaptimmune to GNE	43
<u>7.2.</u>	License from GNE to Adaptimmune	<u>44</u>
<u>7.3.</u>	Third Party IP Licenses	<u>45</u>
<u>7.4.</u>	No Additional Licenses	<u>46</u>
<u>7.5.</u>	GNE and Roche	<u>46</u>
<u>7.6.</u>	Pre-Existing License Terms	<u>46</u>
<u>7.7.</u>	[***]	<u>60</u>
ARTICLE 8	MANUFACTURING, SUPPLY AND TECHNOLOGY TRANSFER	<u>46</u>
<u>8.1.</u>	Phase I Supply by Adaptimmune	<u>46</u>
<u>8.2.</u>	GNE Supply	<u></u> <u>47</u>
<u>8.3.</u>	Provision of Cell Lines	<u>48</u>
<u>8.4.</u>	<u>Technology Transfer</u>	<u>48</u>
ARTICLE 9	<u>EXCLUSIVITY</u>	<u>48</u>
<u>9.1.</u>	Adaptimmune Exclusivity Requirements	<u>48</u>
<u>9.2.</u>	Change of Control	<u>49</u>
ARTICLE 10) FINANCIAL TERMS	<u>50</u>
<u>10.1.</u>	<u>Upfront Payment</u>	<u>50</u>
<u>10.2.</u>	Research Term Extension Fee	<u>50</u>
<u>10.3.</u>	Exclusivity and Research Milestone Payments	<u>50</u>
<u>10.4.</u>	Development Event Payments	<u>52</u>
<u>10.5.</u>	Net Sales Event Payments	<u>52</u>
<u>10.6.</u>	Notice of Achievement; Payment	<u>55</u>
<u>10.7.</u>	Royalties	<u>55</u>
<u>10.8.</u>	Royalty Reductions	<u>56</u>
<u>10.9.</u>	Economic Effect of Opt-In; Profit-Share	<u> <u>59</u></u>
<u>10.10</u>	. Right of Negotiation	<u>60</u>
ARTICLE 1	I PAYMENT TERMS; REPORTS; AUDITS	<u>60</u>
11.1	Timing of Royalty Payment	60

<u>11.2.</u>	Royalty Report	<u>60</u>
<u>11.3.</u>	Mode of Payment	<u>60</u>
<u>11.4.</u>	<u>Currency of Payments</u>	<u>60</u>
<u>11.5.</u>	Blocked Currency	<u>61</u>
<u>11.6.</u>	<u>Taxes</u>	<u>61</u>
<u>11.7.</u>	Records; Inspection	<u>62</u>
ARTICLE 1	2 INTELLECTUAL PROPERTY; OWNERSHIP	<u>63</u>
<u>12.1.</u>	Definitions and Ownership	<u>63</u>
<u>12.2.</u>	Disclosure; Inventorship; Assignment and Cooperation; Joint Ownership	<u>64</u>
<u>12.3.</u>	Patent Prosecution	<u>65</u>
<u>12.4.</u>	Enforcement Rights for Infringement by Third Parties	<u>66</u>
<u>12.5.</u>	Third Party Infringement Claims	<u>68</u>
<u>12.6.</u>	Trademarks	<u>70</u>
<u>12.7.</u>	<u>Unified Patent Court (Europe)</u>	<u>70</u>
<u>12.8.</u>	Common Interest Disclosures	<u>70</u>
<u>12.9.</u>	Patent Term Extensions	<u>71</u>
12.10	D. Patent Listings	<u>71</u>
ARTICLE 1	3 CONFIDENTIALITY	<u>71</u>
<u>13.1.</u>	Non-Use and Non-Disclosure of Confidential Information	<u>71</u>
<u>13.2.</u>	Exclusions Regarding Confidential Information	<u>71</u>
<u>13.3.</u>	Authorized Disclosures of Confidential Information	<u>72</u>
<u>13.4.</u>	Terms of this Agreement	<u>73</u>
<u>13.5.</u>	Termination of Prior Agreements	<u>73</u>
<u>13.6.</u>	No License	<u>73</u>
ARTICLE 1	4 PUBLICITY; PUBLICATIONS; USE OF NAME	<u>73</u>
<u>14.1.</u>	Publicity and Disclosures	<u>73</u>
<u>14.2.</u>	Disclosures by Adaptimmune	<u>74</u>
<u>14.3.</u>	<u>Disclosures by GNE</u>	<u>74</u>
<u>14.4.</u>	Approved Disclosures	<u>74</u>
<u>14.5.</u>	Filing of Agreement	<u>74</u>
<u>14.6.</u>	Scientific Publications	<u>74</u>
<u>14.7.</u>	No Right to Use Names	<u></u>
ARTICLE 1	5 REPRESENTATIONS, WARRANTIES AND COVENANTS	<u></u>
<u>15.1.</u>	Mutual Representations and Warranties	<u>76</u>

<u>15.2.</u>	Adaptimmune Additional Representations and Warranties	<u></u>
<u>15.3.</u>	GNE Additional Representations and Warranties	<u></u>
<u>15.4.</u>	Additional Adaptimmune Covenants	<u></u>
<u>15.5.</u>	Additional GNE Covenants	<u>79</u>
<u>15.6.</u>	<u>Disclaimers</u>	80
ARTICLE 1	6 INDEMNIFICATION	<u>80</u>
<u>16.1.</u>	<u>Indemnification</u>	<u>80</u>
<u>16.2.</u>	<u>Procedure</u>	<u>81</u>
<u>16.3.</u>	<u>Insurance</u>	<u>82</u>
<u>16.4.</u>	Limitation of Damages	<u>83</u>
ARTICLE 1	7 TERM; TERMINATION	<u>83</u>
<u>17.1.</u>	<u>Term</u>	<u>83</u>
<u>17.2.</u>	Termination by Either Party for Material Breach	<u>83</u>
<u>17.3.</u>	Termination by Either Party for Insolvency or Bankruptcy	<u>83</u>
<u>17.4.</u>	GNE Termination	<u>84</u>
<u>17.5.</u>	Termination for Discontinuation Event	<u>104</u>
<u>17.6.</u>	Effects of Termination	<u>84</u>
ARTICLE 1	8 HSR FILING; TERMINATION UPON HSR DENIAL	<u>87</u>
ARTICLE 19	9 DISPUTE RESOLUTION	<u>87</u>
<u>19.1.</u>	<u>Disputes</u>	<u>87</u>
<u>19.2.</u>	<u>Arbitration</u>	<u>88</u>
<u>19.3.</u>	Subject Matter Exclusions	<u>89</u>
<u>19.4.</u>	Continued Performance	<u>89</u>
ARTICLE 20	0 MISCELLANEOUS	<u>89</u>
<u>20.1.</u>	Applicable Law	<u>89</u>
<u>20.2.</u>	<u>Notices</u>	<u>89</u>
<u>20.3.</u>	Assignment	<u>90</u>
<u>20.4.</u>	Certain Employees of the Other Party	<u>90</u>
<u>20.5.</u>	Force Majeure	<u>90</u>
<u>20.6.</u>	Independent Contractors	<u>91</u>
<u>20.7.</u>	<u>Integration</u>	<u>91</u>
<u>20.8.</u>	Amendment; Waiver	<u>91</u>
<u>20.9.</u>	Further Assurance	<u>91</u>
<u>20.10</u>). Severability	<u>91</u>

20.11. No Third Party Rights	<u>91</u>
20.12. Construction	<u>91</u>
20.13. Actions of Affiliates	<u>91</u>
20.14. Interpretation	<u>91</u>
20.15. Counterparts	<u>92</u>

List of Schedules and Exhibits

Schedule 1.50	Excluded Targets
Schedule 1.51	Existing Upstream License Agreements
Schedule 1.80	Initial Collaboration Targets
Schedule 1.123	Specifications
Schedule 2.10	Expert Determination of [***]
Schedule 3.8	Approved Subcontractors
Schedule 6.1.7	Baseball-Style Arbitration
Schedule 7.6	Pre-Existing License Terms
Schedule 8.3.3	[***]
Schedule 10.4.1(a)	No Adaptimmune Opt-In
Schedule 10.4.1(b)	Adaptimmune Opt-In
Schedule 10.4.2	Collaboration Personalised T-Cell Therapy Development Milestones
Schedule 14.1	Press Release
Exhibit A	Adaptimmune Differentiation Platform
Exhibit B	Research Plan
Exhibit C	Adaptimmune Licensed Patent Rights
Exhibit D	Adaptimmune Royalty Patent Rights
Exhibit E	Example US Net Profit and Net Loss Calculation in Section 10.9.1
	-V-

STRATEGIC COLLABORATION AND LICENSE AGREEMENT

This Strategic Collaboration and License Agreement ("Agreement") is made and entered into, as of September 3, 2021 ("Execution Date"), by and among Adaptimmune Limited, having its principal place of business at 60 Jubilee Avenue, Milton Park, Abingdon, Oxfordshire OX14 4RX, United Kingdom ("Adaptimmune"), on the one hand, and Genentech, Inc., a Delaware corporation, having its principal place of business at 1 DNA Way, South San Francisco, California 94080, United States ("GNE"), and F. Hoffmann-La Roche Ltd, having its principal place of business at Grenzacherstrasse 124, CH 4070 Basel, Switzerland ("Roche"), on the other hand. GNE and Adaptimmune are sometimes referred to herein individually as a "Party" and collectively as the "Parties." The term "Party" or "Parties" shall not include Roche unless explicitly stated below.

BACKGROUND

WHEREAS, Adaptimmune is a biotechnology company that is engaged in the research and development of Cell Therapies (as defined below), including T-Cell receptor (" TCR") based Cell Therapies (a " T-Cell Therapy") for pharmaceutical therapy use.

WHEREAS, Adaptimmune is in the process of developing an allogeneic Manufacturing process for the generation of Allogeneic T-Cells (as defined below) from induced pluripotent stem cells.

WHEREAS, GNE is developing certain Receptors (as defined below), including TCRs and $\alpha\beta$ Receptors (as defined below), for use in T-Cell Therapies.

WHEREAS, the Parties desire to collaborate to develop "off-the shelf" T-Cell Therapies and personalised T-Cell Therapies using $\alpha\beta$ Receptors and $\alpha\beta$ Allogeneic T-Cells.

WHEREAS, GNE desires to obtain an exclusive license and other rights from Adaptimmune to Research, Develop, Manufacture and Commercialize Collaboration Off-the-Shelf T-Cell Therapies and Collaboration Personalised T-Cell Therapies (each defined below), and Adaptimmune agrees to grant GNE such an exclusive license and other rights in exchange for certain agreed to upfront and other payments.

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, GNE, Roche and Adaptimmune agree as follows:

ARTICLE 1 DEFINITIONS

Capitalized terms used in this Agreement, whether used in the singular or plural, shall have the meanings set forth below, unless otherwise specifically indicated herein.

1.1 " $\alpha\beta$ Allogeneic T-Cell" means an Allogeneic T-Cell that incorporates an $\alpha\beta$ Receptor, whether such $\alpha\beta$ Receptor is incorporated into such Allogeneic T-Cell prior to or after differentiation from an iPS Cell.

- 1.2 " $\alpha\beta$ Receptor" means a Receptor that is a transmembrane heterodimer consisting of alpha and beta chains that recognizes one (1) or more Antigens when bound to MHC class 1 or MHC class 2 molecules (including HLA), including any such Receptor that recognizes one (1) or more Antigens of a Collaboration Target.
- 1.3 "Accounting Standard" means, with respect to each Party, either: (a) International Financial Reporting Standards (" IFRS"); or (b) United States generally accepted accounting principles (" GAAP"), in either case, which standards or principles (as applicable) are used at the applicable time, and as consistently applied, by the applicable Party.
- 1.4 "Acquired Party Family" means in the case of a Change of Control of a Party or its Affiliate, (a) such entity subject to the Change of Control and each Affiliate of such entity existing immediately prior to the Change of Control transaction and (b) any subsidiaries of such entity and its Affiliates in clause (a), whether then existing or thereafter created .
- 1.5 "Acquiring Entity" means, in the case of a Change of Control of a Party or its Affiliate, the successor in interest, resulting entity, assignee or purchaser, as applicable, of such Party or such Affiliate.
- 1.6 "Acquiring Entity Family" means in the case of a Change of Control of a Party or its Affiliate, the Acquiring Entity and its Affiliates existing immediately prior to the closing of the Change of Control transaction together with any future Affiliates of such Party or such Affiliate (but excluding the Acquired Party Family).
- 1.7 "Active Research Program" means, [***].
- 1.8 "Adaptimmune Cell Therapy Platform" means Adaptimmune's proprietary cell therapy platform used for the genetic editing of T-Cells (including iPS Cell-derived T-Cells), affinity engineering of $\alpha\beta$ Receptors of T-cells (including iPS Cell-derived T-Cells, but not affinity engineering of GNE-Provided $\alpha\beta$ Receptors), characterisation and testing of iPS Cells, including the insertion sites used by Adaptimmune to edit iPS Cells, constructs for enhancing the safety, persistence, efficacy or affinity of T-Cells or T-Cell $\alpha\beta$ Receptors (other than any GNE-Provided $\alpha\beta$ Receptor), integration and expression of Receptors (other than any GNE-Provided $\alpha\beta$ Receptor) and Cell Therapy products.
- 1.9 "Adaptimmune Differentiation Platform" means Adaptimmune's proprietary cell differentiation process to differentiate an Allogeneic T-Cell from an iPS Cell, including the process described in <u>Exhibit A</u> (Adaptimmune Differentiation Platform) to the extent such process is proprietary to Adaptimmune.
- 1.10 "Adaptimmune iPS Cell Line" means any iPS Cell Line that is Controlled by Adaptimmune and used in the Research Program, including each iPS Cell Line to which Adaptimmune has rights under the Existing Upstream License Agreements.
- 1.11 "Adaptimmune Licensed IP" means Adaptimmune Licensed Know-How and Adaptimmune Licensed Patent Rights.

- 1.12 "Adaptimmune Licensed Know-How" means all rights in Know-How Controlled by Adaptimmune or its Affiliates as of the Effective Date or thereafter that are necessary or reasonably useful for the use, Research, Development, making, having made, sale, offering for sale, importation, exportation, Manufacture or Commercialization of any Licensed Product, [***].
- 1.13 "Adaptimmune Licensed Patent Rights" means: (a) all Patents set forth in Exhibit C (Adaptimmune Licensed Patent Rights); (b) any and all Patents (other than those described in clause (a)) that are Controlled by Adaptimmune or its Affiliates as of the Effective Date or during the Research Term that are necessary or reasonably useful for, the use, Research, Development, making, having made, sale, offering for sale, importation, exportation, Manufacture or Commercialization of any Licensed Product; and (c) any Patents that are Controlled by Adaptimmune as of the Effective Date or during the Term that Cover the Adaptimmune Differentiation Platform and that are necessary or reasonably useful for, the use, Research, Development, making, having made, sale, offering for sale, importation, exportation, Manufacture or Commercialization of any Licensed Product.
- 1.14 "Adaptimmune Off-the-Shelf Royalty Patent Rights" means all Patents set forth on Exhibit D (Adaptimmune Royalty Patent Rights) under the header "Adaptimmune Off-the-Shelf Royalty Patent Rights".
- 1.15 "Adaptimmune Personalised Royalty Patent Right" means (a) all Patents within Adaptimmune Off-the-Shelf Royalty Patent Rights and (b) all Patents that (i) Cover the [***] and (ii) are filed prior to the expiration of, or have a priority date that is during, the Research Term, including all Patents within [***]. Upon coming into existence, such Patents described in clause (b) of the preceding sentence shall automatically be added to Exhibit D (Adaptimmune Royalty Patent Rights) under the header "Adaptimmune Personalised Royalty Patent Rights".
- 1.16 "**Adaptimmune Platform**" means the Adaptimmune Cell Therapy Platform and the Adaptimmune Differentiation Platform.
- 1.17 "Adaptimmune Royalty Patent Rights" means the Adaptimmune Off-the-Shelf Royalty Patent Rights and the Adaptimmune Personalised Royalty Patent Rights.
- 1.18 "Affiliate" means any person that, directly or indirectly (through one or more intermediaries) controls, is controlled by, or is under common control with a Party. For purposes of this Section 1.18 ("Affiliate"), "control" means: (a) the direct or indirect ownership of fifty percent (50%) or more of the voting stock or other voting interests or interest in the profits of the Party; or (b) the ability to otherwise control or direct the decisions of board of directors or equivalent governing body thereof. Notwithstanding the foregoing, none of the following entities shall be deemed to be Affiliates of GNE for the purposes of this Agreement, unless GNE provides written notice to Adaptimmune of GNE's desire to include any such entity (and/or any of such entity's subsidiaries) as Affiliate(s) of GNE and as of the date of provision of such written notice the entity (and/or any of such entity's subsidiaries) named in such notice shall thereafter be deemed an Affiliate(s) of GNE: [***].

1.19 "Allowable Expenses" means, with respect to a Collaboration Off-the-Shelf T-Cell Therapy Directed To a Collaboration Target for which Adaptimmune timely exercised its Opt-In, the following costs and expenses paid or accrued [***]:

```
1.19.1 [***];
1.19.2 [***];
1.19.3 [***];
1.19.4 [***];
1.19.5 [***]; and
1.19.6 [***].
```

- "Annual Net Sales" means, (a) in the case of an Off-the-Shelf Net Sales Milestone Payment pursuant to Section 10.5.1 (Collaboration Off-the-Shelf T-Cell Therapy Net Sales Milestones) or royalty payment pursuant to Section 10.7.1 (Royalty Payments for Collaboration Off-the-Shelf T-Cell Therapy), the total Net Sales by GNE, its Affiliates and its Sublicensees in the applicable territor(ies) for Collaboration Off-the-Shelf T-Cell Therapies Directed To the applicable Collaboration Target in a particular calendar year, or (b) in the case of a Personalised Net Sales Milestone Payment pursuant to Section 10.5.2 (Collaboration Personalised T-Cell Therapy Net Sales Milestones) or royalty payment pursuant to Section 10.7.2 (Royalty Payments for Collaboration Personalised T-Cell Therapy), the total Net Sales by GNE, its Affiliates and its Sublicensees in the applicable territor(ies) for Collaboration Personalised T-Cell Therapies in a particular calendar year.
- 1.21 "Antigen" means a peptide or protein (or any fragment or epitope thereof) against which the immune system may produce an adaptive immune response.
- 1.22 "Available Target" means each Target that is not an Excluded Target as of the time of Nomination in accordance with Section 3.6 (Nomination).
- 1.23 "Business Day" means any day other than: (a) a Saturday or Sunday or any day on which commercial banks in San Francisco, California or London, England are authorized or required by applicable law to remain closed; or (b) December 26 through December 31.
- 1.24 "Cell Line" means an established cell culture developed from a single cell that will proliferate given appropriate medium and space.
- 1.25 "Cell Therapy" means the administration of living cells to a patient for treatment of a disease or condition.

- 1.26 "Change of Control" means, with respect to a Party: (a) that a Third Party acquires directly or indirectly the beneficial ownership of any voting securities of such Party, or if the percentage ownership of such Third Party in the voting securities of such Party is increased through stock redemption, cancellation or other recapitalization, and immediately after such acquisition or increase such Third Party is, directly or indirectly, the beneficial owner of outstanding voting securities representing fifty percent (50%) or more of the total voting power of all of the then outstanding voting securities of such Party; (b) a merger (whether by contract, by statute or by operation of law), consolidation, recapitalization or reorganization of such Party is consummated; (c) a liquidation or dissolution of such Party or any direct or indirect parent of such Party. excluding in the case of (b) or (c), any such transaction in which stockholders or equity holders of such Party immediately prior to such transaction beneficially own, directly or indirectly, more than fifty percent (50%) of the voting securities of the surviving entity (or its parent entity) immediately following such transaction; or (d) the sale or disposition to a Third Party of all or substantially all of such Party's assets relating to this Agreement. For purposes of this definition, "beneficial ownership" shall have the meaning accorded in the US Securities Exchange Act of 1934 and the rules of the US Securities and Exchange Commission as may be updated from time to time. Notwithstanding the foregoing, a transaction solely to change the domicile or legal form of a Party shall not constitute a Change of Control.
- 1.27 "Clinical Trial" means a Phase I Clinical Trial, Phase II Clinical Trial (including for avoidance of any doubt a phase Ib or phase IIb clinical trial), Pivotal/Registrational Trial, or Phase III Clinical Trial, or any other equivalent, combined or other trial in which any product is administered to a human subject.
- 1.28 "CMO" means any Third Party contract manufacturer.
- 1.29 "Collaboration IP" means Collaboration Know-How and Collaboration Patents.
- 1.30 "Collaboration iPS Cell Line" means each iPS Cell Line made in the Research Program.
- 1.31 "Collaboration Know-How" means any Know-How arising out of the performance of activities by either Party under this Agreement (including activities under the Collaboration Off-the-Shelf Exclusive License and Collaboration Personalised Exclusive License).
- 1.32 "Collaboration Off-the-Shelf T-Cell Therapy" means, on a Collaboration Target-by-Collaboration Target basis, a T-Cell Therapy consisting of $\alpha\beta$ Allogeneic T-Cells that (a) are engineered (including by the insertion of Receptor) or differentiated from an Adaptimmune iPS Cell Line, and (b) incorporate one or more GNE-Provided Off-the-Shelf $\alpha\beta$ Receptors (but no other Receptors), which GNE-Provided Off-the-Shelf $\alpha\beta$ Receptors are Directed To such Collaboration Target, including any Functionally Distinct Collaboration Off-the-Shelf T-Cell Therapy that is Directed To such Collaboration Target .
- 1.33 "Collaboration Patent" means any Patent that Covers Collaboration Know-How.
- 1.34 "Collaboration Personalised T-Cell Therapy" means a Personalised T-Cell Therapy consisting of $\alpha\beta$ Allogeneic T-Cells that (a) are engineered (including insertion of Receptor) or differentiated from an Adaptimmune iPS Cell Line under this Agreement, and (b) incorporate only

one or more GNE-Provided Personalised $\alpha\beta$ Receptors, including any Functionally Distinct Collaboration Personalised T-Cell Therapy .

- 1.35 "Collaboration Target" means each (a) Initial Collaboration Target, (b) Additional Collaboration Target following Nomination of such Target by GNE in accordance with Section 3.6 (Nomination) and (c) Available Target that replaces a Replaced Collaboration Target in accordance with Section 3.4.2 (Collaboration Target Substitution). For clarity, a Replaced Collaboration Target is not a Collaboration Target.
- 1.36 "Commercialization" or "Commercialize" means marketing, promoting, detailing, distributing, importing, exporting, offering for sale or selling a product, including medical affairs activities, regulatory activities directed to obtaining pricing and reimbursement approvals, price calculations and related reporting to governmental authorities, and interacting with Regulatory Authorities with respect to the foregoing. For clarity, as used in this Agreement, "Commercialization" includes Manufacturing a product. When used as a verb, "Commercialize" means to engage in Commercialization activities.
- 1.37 "Commercially Reasonable Efforts" means [***].
- 1.38 "Committee" means the JRC, JDC, JPC and JIPC as applicable.
- 1.39 "Compulsory Sublicense" means a sublicense granted to a Third Party, through the order, decree or grant of a governmental authority having competent jurisdiction, authorizing such Third Party to Manufacture, Develop, Commercialize, use, sale, offer for sale, import or export a Licensed Product in any country in the Territory with a royalty rate lower than the applicable royalty rate set forth in Section 10.7 (Royalties) as adjusted pursuant to Section 10.8 (Royalty Reductions).
- 1.40 "Compulsory Sublicensee" means a Third Party that was granted a Compulsory Sublicense. For clarity, a Compulsory Sublicensee is not a Sublicensee.
- "Confidential Information" means proprietary Know-How (of whatever kind and in whatever form or medium, including copies thereof), information within tangible materials or other deliverables: (a) disclosed by or on behalf of a Party or its Affiliates or Sublicensees in connection with this Agreement, whether prior to or during the Term and whether disclosed orally, electronically, by observation or in writing; or (b) created by, or on behalf of, either Party or its Affiliates and provided to the other Party or its Affiliates, or created jointly by the Parties, in the course of this Agreement. For the avoidance of doubt, "Confidential Information" includes: (i) Know-How regarding such Party's Research, Development plans, Clinical Trial designs, preclinical and clinical data, technology, products, business information or objectives and other information of the type that is customarily considered to be confidential information by entities engaged in activities that are substantially similar to the activities being engaged in by the Parties pursuant to this Agreement; and (ii) non-public information which a Party treats as trade secret in accordance with its normal business practices or which would be capable of attracting protection as a trade secret in accordance with relevant laws and on a country by country basis [***].

- 1.42 "Control" or "Controlled by" means, subject to Section 7.3 (Third Party IP Licenses), (as an adjective or as a verb including conjugations and variations such as "Controls" " Controlled" or "Controlling") (a) with respect to Intellectual Property, the possession by a Party of the ability to grant a license or sublicense of such Intellectual Property (whether through ownership or license (other than a license granted from one Party to the other Party under this Agreement)) without violating the terms of any agreement or arrangement between such Party and any Third Party, and (b) with respect to proprietary materials, the possession by a Party of the ability to grant access to such proprietary materials (other than a license or other right granted from one Party to the other Party under this Agreement) to the other Party as provided herein without violating the terms of any agreement or arrangement between such Party and any Third Party at the time such access is granted hereunder. Notwithstanding anything to the contrary in this Agreement, in the event of a Change of Control of a Party, then, whether or not this Agreement is assigned to the Acquiring Entity, any Intellectual Property rights owned or controlled by the Acquiring Entity Family shall not be deemed to be Controlled by such Party after the effective date of such Change of Control transaction for the purposes of this Agreement, except to the extent any such Intellectual Property rights are (i) developed, acquired or otherwise Controlled by the Acquiring Entity Family pursuant to or in connection with a license or other agreement between the Acquiring Entity or any of its Affiliates, on the one hand, and Adaptimmune or any of its Affiliates, on the other hand, entered into prior to such Change of Control, but only to the extent such license or other agreement permits the granting of a license, sublicense, access, right to use or release of such Intellectual Property by Adaptimmune or its Affiliates to GNE without breaching the terms of such license or such other agreement, or (ii) used by such Party or its Affiliates (including Affiliates following such Change of Control) in the Research, Development, Manufacture, or Commercialization of Licensed Products after such Change of Control transaction.
- 1.43 "Covers" means (as an adjective or as a verb including conjugations and variations such as "Covered," "Coverage" or "Covering"), with respect to a particular subject matter at issue and a relevant Patent, that [***]. The determination of whether a particular subject matter is Covered by a particular Patent shall be made on a country-by-country basis.
- 1.44 "Development" or "Develop" means for a given product, any activity directed to obtaining or expanding Marketing Approval, including all preclinical and clinical drug or biologic product development activities, including: the conduct of Clinical Trials, cell line development, Master Cell Bank generation, test method development and stability testing, toxicology, formulation and delivery system development, process development, pre-clinical and clinical supply, Manufacturing scale-up, development-stage Manufacturing, quality assurance/quality control procedure development and performance with respect to clinical materials, statistical analysis and report writing and clinical studies, regulatory affairs with respect to the foregoing. When used as a verb, "Develop" means to engage in Development.
- 1.45 "**Development Costs**" means, with respect to a Collaboration Off-the-Shelf T-Cell Therapy directed to a Collaboration Target for which Adaptimmune has timely exercised its Opt-In, [***]:
- 1.45.1 [***];

1.45.2	[***];
1.45.3	[***];
1.45.4	[***];
1.45.5	[***];
1.45.6	[***]; and
1.45.7	[***].
[***].	

"Directed To" means, with regard to a Target (including an Antigen of such Target and further including an HLA-presented Antigen) and a Cell Therapy containing an $\alpha\beta$ Receptor, that such $\alpha\beta$ Receptor binds directly to such Target (including an Antigen of such Target and further including an HLA-presented Antigen), and such binding causes pharmacologically relevant activity. When required grammatically, the defined term "Directed To" may be separated and will have the same meaning set forth above; *e.g.*, when discussing Targets to which a compound or product is Directed.

1.47 "Discontinuation Event" means [***].

- 1.48 "Effective Date" means the first (1st) Business Day immediately following the date on which the Parties have actual knowledge that all applicable waiting periods under the HSR Act with respect to the transactions contemplated under this Agreement have expired or have been terminated. Upon the request of either Party, the Parties shall memorialize the Effective Date, as defined in the immediately preceding sentence, in a written document for the records.
- 1.49 "EU" means the then-current member states of the European Union; provided that, for the purposes of this Agreement all references herein to the European Union or EU shall be construed to mean the then-current member states of the European Union together with the United Kingdom.
- 1.50 "Excluded Target" means any Target at the time of Adaptimmune's receipt of GNE's applicable Nomination (a) for which Adaptimmune has Developed a T-Cell Therapy that is Directed To such Target and for which Adaptimmune has received a Marketing Approval; (b) that is the subject of an Active Research Program; (c) which is listed on Schedule 1.50 (Excluded Targets); or (d) for which Adaptimmune has licensed, or agreed to license, or is in *bona fide* negotiations to license to a Third Party the right to Develop or Commercialize T-Cell Therapy(ies) Directed To such Target under a then-binding written agreement or term sheet, in each case at the time of GNE's Nomination as further de scribed in Section 3.6.1 (Proposed Targets).
- 1.51 "Existing Upstream License Agreements" means the agreements identified on Schedule 1.51 (Existing Upstream License Agreements), pursuant to which Adaptimmune in-licenses or otherwise acquires Control of Adaptimmune Licensed IP.

- 1.52 "FDA" means the United States Food and Drug Administration, or any successor entity thereto performing similar functions.
- 1.53 "Field" means the treatment, prevention, palliation or diagnosis of all Indications, diseases, disorders and conditions.
- 1.54 **"First Commercial Sale"** means, with respect to a particular Licensed Product in a given country, [***].
- 1.55 [***].
- 1.56 "**Fixed SG&A**" means the amount calculated by multiplying the applicable Fixed SG&A Percentage by the applicable Collaboration Off-the-Shelf T-Cell Therapy's Net Sales amount in the US.
- 1.57 **"Fixed SG&A Percentage"** means on a Collaboration Off-the-Shelf T-Cell Therapy-by-Collaboration Off-the-Shelf T-Cell Therapy basis:
 - (i) [***];
 - (ii) [***]; and
 - (iii) [***].
- 1.58 "FTE" means, with respect to a person, the equivalent of the work of one individual full time (whether provided by a single individual full time or multiple individuals part-time) for one (1) calendar year (consisting of in general a total of [***] per calendar year). Overtime and work on weekends, holidays and the like will not be counted with any multiplier (e.g., time-and-a-half or double time) toward the number of hours that are used to calculate the FTE contribution.
- 1.59 **"FTE Costs"** [***].
- 1.60 "Fully Burdened Cost" means [***].
- 1.61 "Functionally Distinct" means: [***]. Where any iPS Cell Line is generated during the performance of the Research Plan, the JRC shall determine whether any iPS Cell Line is a Functionally Distinct Collaboration OTS Cell Line or Functionally Distinct Collaboration Personalised Cell Line. Outside of the performance of the Research Plan, whether any iPS Cell Line is a Functionally Distinct Collaboration OTS Cell Line or Functionally Distinct Collaboration Personalised Cell Line shall be determined in accordance with Section 2.6 (Dissolution of the JPC and JRC; Functionally Distinct Determinations).
- 1.62 "Functionally Distinct Collaboration Off-the-Shelf Therapy" means that the applicable Collaboration OTS Cell Line of a Collaboration Off-the-Shelf T-Cell Therapy that is Directed To a Collaboration Target is Functionally Distinct from all other Collaboration OTS Cell Lines of a Collaboration Off-the-Shelf T-Cell Therapy that is Directed To the same Collaboration Target.

- 1.63 "Functionally Distinct Collaboration Personalised T-Cell Therapy" means the applicable Collaboration Personalised Cell Line of a given Collaboration Personalised T-Cell Therapy is Functionally Distinct from all other Collaboration Personalised Cell Lines of another Collaboration Personalised T-Cell Therapy.
- 1.64 "Global Development Plan" means, with respect to a Collaboration Off-the-Shelf T-Cell Therapy Directed To a Collaboration Target for which Adaptimmune has timely exercised its Opt-In, a written development plan that is prepared and delivered to Adaptimmune in accordance with, and is subject to, <u>Article 6</u> (Adaptimmune Opt-In Right) of this Agreement. Without limiting the generality of the foregoing, a Global Development Plan will include [***].
- 1.65 "GMP" means current good manufacturing practices and regulations applicable to the Manufacture of Cell Therapy products that are promulgated by any Regulatory Authority, including as promulgated under and in accordance with (a) the US Federal Food, Drug and Cosmetic Act, Title 21 of the US Code of Federal Regulations, Parts 210, 211, 600, 601 and 610, (b) relevant EU legislation, including European Directive 2003/94/EC or national implementations of that Directive, (c) relevant guidelines, including the EU Guidelines for Good Manufacturing Practices for Medicinal Products (Eudralex Vol. 4 and Annexes thereto), (d) International Conference on Harmonisation Good Manufacturing Practice Guide for Active Pharmaceuticals Ingredients, applicable US FDA guidance documents, and (e) any analogous set of regulations, guidelines or standards as defined, from time to time, by any relevant Regulatory Authority applicable to any Party's Development, Manufacture or Commercialization of Cell Therapy products in each case as in effect as of the date such manufacturing for such Cell Therapy product are or were conducted.
- 1.66 "GNE Licensed IP" means the GNE Licensed Know-How and the GNE Licensed Patent Rights.
- 1.67 "GNE Licensed Know-How" means all rights in Know-How Controlled by GNE as of the Effective Date or during the Research Term that is necessary for Adaptimmune to perform its obligations under the Research Plan.
- 1.68 "GNE Licensed Patent Rights" means all Patents Controlled by GNE as of the Effective Date or during the Research Term that are necessary for Adaptimmune to perform its obligations under the Research Plan.
- 1.69 [***].
- 1.70 "GNE-Provided $\alpha\beta$ Receptor" means any $\alpha\beta$ Receptor provided by or on behalf of GNE for use under this Agreement and any modification or derivatives thereto, including GNE-Provided Off-the-Shelf $\alpha\beta$ Receptors and GNE-Provided Personalised $\alpha\beta$ Receptors.
- 1.71 "GNE-Provided Off-the-Shelf $\alpha\beta$ Receptor" means any GNE-Provided $\alpha\beta$ Receptor that is Directed To a Collaboration Target and any modification or derivatives thereto.

- 1.72 "GNE-Provided Personalised $\alpha\beta$ Receptor" means any GNE-Provided $\alpha\beta$ Receptor that is isolated from a patient for use in a Personalised T-Cell Therapy and any modification or derivatives thereto.
- 1.73 **"Governmental Required Consents"** means, with respect to a Party, compliance by such Party with, and filings by such Party under, the HSR Act.
- 1.74 "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended from time to time, and any comparable applicable law in jurisdictions outside the US related to the approval of transactions similar to those contemplated under this Agreement.
- 1.75 "HSR Clearance Date" means the expiration or termination of (a) all applicable waiting periods and requests for information (and any extensions thereof) under the HSR Act and (b) any agreements with the US Federal Trade Commission or the Antitrust Division of the US Department of Justice not to consummate the transactions contemplated by this Agreement.
- 1.76 "HSR Filing" means: (a) filings by the Parties with the US Federal Trade Commission and the Antitrust Division of the US Department of Justice of a Notification and Report Form for Certain Mergers and Acquisitions (as that term is defined in the HSR Act) with respect to the matters set forth in this Agreement, together with all required documentary attachments thereto; or (b) equivalent filings with relevant foreign authorities.
- 1.77 "IND" means an investigational new drug application filed with the FDA pursuant to 21 CFR Part 312 before the commencement of Clinical Trials, or any comparable filing with any relevant Regulatory Authority in any other jurisdiction, including any clinical trial application submitted to a Regulatory Authority to gain authorization to perform Clinical Trials.
- 1.78 "IND Acceptance" means, with respect to (a) an IND submitted by GNE, its Affiliates, or its or their Sublicensees to a Regulatory Authority or (b) an IND submitted by Adaptimmune or its Affiliates to a Regulatory Authority for the purposes of Section 10.3.2(a) (Research Milestone Payments), in each case ((a) and (b)) the earlier of (i) receipt by such Party or its Affiliate, or in the case of (a), a Sublicensee of GNE or GNE's Affiliate, of written confirmation from such Regulatory Authority that Clinical Trials may proceed under such IND, and (ii) expiration of the applicable waiting period after which Clinical Trials may proceed under such IND.
- 1.79 "Indication" means a disease (a) for which a Licensed Product is indicated for treatment, and (b) that is described in the Licensed Product label as required by the Regulatory Approval granted by the applicable Regulatory Authority. For purposes of determining whether an Indication for a Licensed Product is distinct from another Indication, an Indication ("New Indication") is distinct from an existing Indication ("Existing Indication") if such Licensed Product could not be lawfully promoted for the treatment of the New Indication under the Regulatory Approval and label for the Existing Indication. For clarity, label extensions (including front-line, metastatic, adjuvant, etc.) shall not be deemed to be separate Indications.
- 1.80 "Initial Collaboration Target" means each of the Targets listed on Schedule 1.80 (Initial Collaboration Targets).

- 1.81 "Initial Genetic Edits" means genetic edits generating the following iPS Cell Line phenotype: [***]
- 1.82 "**Initiation**" means, with respect to a Clinical Trial, the dosing of the first patient with the drug or product (or placebo) that is the subject of such Clinical Trial.
- 1.83 "In-License Payments" means [***].
- 1.84 "Intellectual Property" means all (a) Patents, including Patent applications, (b) trademarks, service marks, trade dress, trade names, logos and corporate names and registrations and applications for registration thereof together with all of the goodwill associated therewith, (c) copyrights (registered or unregistered) and copyrightable works and registrations and applications for registration thereof, (d) rights in Know-How and other confidential information (including, without limitation, ideas, formulas, compositions, inventions ([***]), manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, proposals, technical data, copyrightable works, financial and marketing plans and customer and supplier lists and information, and (e) other intellectual property rights.
- 1.85 "iPS Cell" means an induced human pluripotent stem cell.
- 1.86 "iPS Cell Line" means a Cell Line made, generated, engineered or created from an iPS Cell.
- 1.87 "Know-How" means all non-public information, inventions (whether or not patentable), improvements, practices, formula, trade secrets, techniques, methods, procedures, knowledge, results, test data (including pharmacological, toxicological, pharmacokinetic and pre-clinical and clinical information and test data, related reports, structure-activity relationship data and statistical analysis), analytical and quality control data, protocols, processes, models, designs, and other information regarding discovery, Development, marketing, pricing, distribution, cost, sales and Manufacturing. Know-How shall not include any Patents.
- 1.88 "**Knowledge**" means, as it pertains to Adaptimmune, the actual knowledge of the members of its [***].
- 1.89 "**Launch Costs**" means, with respect to a Collaboration Off-the-Shelf T-Cell Therapy Directed To a Collaboration Target for which Adaptimmune has timely exercised its Opt-In, [***]:
- 1.89.1 [***]; 1.89.2 [***]; 1.89.3 [***]; 1.89.4 [***]; 1.89.5 [***];

1.89.6	[***];
1.89.7	[***];
1.89.8	[***];
1.89.9	[***];
1.89.10	[***];
1.89.11	[***];
1.89.12	[***];
1.89.13	[***];
1.89.14	[***];
1.89.15	[***]

1.89.16 any other costs, losses and expenses of GNE or any of its Affiliates or Sublicensees that are specifically identifiable or reasonably allocable (in accordance with the applicable Accounting Standard) to the launch of such Collaboration Off-the-Shelf T-Cell Therapy in the US.

[***].

- 1.90 "**Licensed Product**" means a Collaboration Off-the-Shelf T-Cell Therapy or a Collaboration Personalised T-Cell Therapy, individually or collectively as the context may require.
- 1.91 "Manufacture", "Manufacturing" or "Manufactured" means, with respect to a given product, the receipt, handling and storage of active pharmaceutical ingredients specific to such product, drug substance or drug product, medical devices and other materials required for the manufacture of the product or for its administration or distribution, the manufacturing, processing, packaging and labelling (excluding the development of packaging and labelling components for Regulatory Approval), holding (including storage of goods in manufacturing but excluding distribution), quality assurance and quality control testing (including release) for such product.
- 1.92 "Manufacturing Costs" means, with respect to a Collaboration Off-the-Shelf T-Cell Therapy directed to a Collaboration Target for which Adaptimmune has timely exercised its Opt-In:
- 1.92.1 GNE, Affiliate or Sublicensee Directly Manufactures. [***]
- 1.92.2 GNE, Affiliate or Sublicensee Uses CMO to Manufacture. [***].
- 1.93 "Marketing Approval" means all approvals, licenses, registrations or authorizations of any federal, state or local regulatory agency, department, bureau or other governmental entity,

necessary for the Manufacturing, use, storage, import, transport and sale of the applicable Licensed Product in a particular country or regulatory jurisdiction. [***].

- "Marketing Approval Application" means a BLA, sBLA, NDA, sNDA and any equivalent thereof in the United States or any other country or jurisdiction in the Territory. As used herein: "BLA" means a Biologics License Application and amendments thereto filed pursuant to the requirements of the FDA, as defined in 21 C.F.R. § 600 et seq., for FDA approval of a Licensed Product and "sBLA" means a supplemental BLA; and "NDA" means a New Drug Application and amendments thereto filed pursuant to the requirements of the FDA, as defined in 21 C.F.R. § 314 et seq., for FDA approval of a Licensed Product and "sNDA" means a supplemental NDA.
- 1.95 "Marketing Studies" mean human Clinical Trials of a Licensed Product conducted following Initiation of a Pivotal/Registrational Trial for such Licensed Product that are not required for receipt of Regulatory Approval (whether such human Clinical Trial is conducted prior to or after receipt of such Regulatory Approval) and is not a Post-Approval Study, but that may be useful in support of the post-Regulatory Approval exploitation of such Licensed Product.
- 1.96 "**Master Cell Bank**" means, with respect to a Cell Line, an aliquot of a single pool of cells of such Cell Line that has been prepared from a selected cell clone of such Cell Line under defined conditions, dispensed into multiple containers and stored under defined conditions.
- 1.97 "Medical Affairs" means, for the purposes of [***], the coordination of medical information requests and field based medical scientific liaisons by or on behalf of GNE or any of its Affiliates or Sublicensees with respect to a Collaboration Off-the-Shelf T-Cell Therapy Directed To a Collaboration Target for which Adaptimmune timely exercised its Opt-In, including activities of medical scientific liaisons, activities involving key opinion leaders, and the provision of medical information services with respect to such Collaboration Off-the-Shelf T-Cell Therapy.
- 1.98 "Neoantigen" means a mutated Antigen arising in a tumor cell.
- 1.99 "Net Sales" means, with respect to a given Licensed Product in a given period on or after the First Commercial Sale in a country [***].
- 1.99.1 [***]. 1.99.2 [***]. [***]:
 - (a) [***].
 - (b) [***].
 - (i) [***].
 - (ii) [***].

- (iii) [***]
- (iv) [***].
- (v) [***].
- 1.100 "Off-the-Shelf T-Cell Therapy" means a Cell Therapy (a) consisting of any Allogeneic T-Cell engineered to express one (1) or more $\alpha\beta$ Receptors that are Directed To a Target (including an Antigen of such Target and further including an HLA-presented Antigen, a Shared Neoantigen or Tumor Associated Antigen), and (b) that is not a Personalised T-Cell Therapy.
- 1.101 "**Opt-In Term**" means, with respect to a given Opt-In by Adaptimmune with respect to a Collaboration Target, the period of time from the Opt-In Effective Date until the Opt-Out Effective Date.
- 1.102 "Out-of-Pocket Costs" [***].
- 1.103 "Patent(s)" means any and all patents and patent applications and any patents issuing therefrom or claiming priority to, worldwide, together with any extensions (including Patent Term Extensions and SPCs) and renewals thereof, reissues, re-examinations, substitutions, confirmation patents, registration patents, invention certificates, patents of addition, renewals, divisionals, continuations, and continuations-in-part of any of the foregoing.
- 1.104 "Permitted Use Cell Line" means [***].
- 1.105 "**Person**" means any individual, partnership, joint venture, limited liability company, corporation, firm, trust, association, unincorporated organization, governmental authority or any other entity not specifically listed herein.
- 1.106 "Personalised T-Cell Therapy" means a Cell Therapy consisting of any Allogeneic T-Cell engineered to express one (1) or more $\alpha\beta$ Receptors that are isolated from the patient to whom such Cell Therapy is administered, including wherein such Receptor(s) are modified after such isolation from, and prior to such administration to, such patient.
- 1.107 "Phase I Clinical Trial" means a human clinical trial that provides for the first introduction into humans of a product for the purpose of, among other things, determining product safety, tolerability, immunogenicity, pharmacological activity or pharmacokinetics, as more fully defined in 21 C.F.R. § 312.21(a) (or the foreign equivalent thereof).
- 1.108 "Phase II Clinical Trial" means a randomized, placebo or active controlled human clinical trial, designed for the evaluation of the efficacy of such product for a particular Indication in the target patient population and a determination of the common side effects and risks associated with the product in the dosage range to be prescribed, and otherwise consistent with 21 C.F.R. §312.21(b) or its foreign equivalents.
- 1.109 "Phase III Clinical Trial" means a human clinical trial, the principal purpose of which is to demonstrate clinically and statistically the efficacy and safety of a product for one or more

Indications in order to obtain Marketing Approval of such product for such Indication(s), as further defined in 21 C.F.R. §312.21 or a similar clinical study in a country other than the United States.

- 1.110 "Pivotal/Registrational Trial" means, with respect to any Licensed Product, (a) a human clinical trial, the principal purpose of which is to demonstrate clinically and statistically the efficacy and safety of a Licensed Product for one or more Indications in order to obtain Marketing Approval of such Licensed Product for such Indication(s), as further defined in 21 C.F.R. §312.21; (b) a human clinical trial of a product on a sufficient number of subjects that, prior to commencement of the trial, satisfies both of the following (both (i) and (ii)): (i) such trial is designed to establish that a Licensed Product has an acceptable safety and efficacy profile for its intended use, and to determine warnings, precautions, and adverse reactions that are associated with such Licensed Product in the dosage range to be prescribed, which trial is intended to support Marketing Approval of such Licensed Product; and (ii) such trial is designed to be a registration trial sufficient to support the filing of Marketing Approval Application for such Licensed Product in the applicable jurisdiction, as evidenced by (A) an agreement with or statement from the FDA or the EMA on a "Special Protocol Assessment" or equivalent, (B) other guidance or minutes issued by the FDA or EMA, for such registration trial, or (C) the equivalent agreement with the applicable Regulatory Authority in such jurisdiction; or (c) a human clinical trial that GNE determines (in communications with the FDA or the equivalent Regulatory Authority in the applicable jurisdiction) is intended to support the filing of a Marketing Approval Application for any Licensed Product in the US or the EU, and such trial will be deemed a Pivotal/Registrational Trial from the date of such determination.
- 1.111 "Post-Approval Study" means a human clinical trial or other nonclinical studies, whether required by a Regulatory Authority or not, of a Licensed Product initiated in a country after receipt of Regulatory Approval for such Licensed Product in such country.
- 1.112 "Prosecution and Maintenance" or "Prosecute and Maintain", with respect to a particular Patent, means all activities associated with the preparation, filing (including any election under the Unitary Patent Convention), prosecution and maintenance of such Patent (and patent application(s) derived from such P atent), as well as re-examinations, reissues, applications for patent term adjustments and extensions, supplementary protection certificates and the like with respect to that Patent, together with the conduct of interferences, derivation proceedings, the defence of oppositions, defence of inter partes review ("IPR") and other similar proceedings with respect to that Patent.
- 1.113 "Receptor" means any protein, molecule or moiety (including a TCR) that, when included, inserted into, appended to, or present in any T-Cell, is [***].
- 1.114 "Regulatory Approval" means the technical, medical and scientific licenses, registrations, authorizations and approvals required for marketing or use of a Licensed Product (including approvals of, BLAs, INDs, pre-and post-approvals, and labelling approvals and any supplements and amendments to any of such approvals) of any national, supra-national, regional, state or local regulatory agency, department, bureau, commission, council or other governmental entity, necessary for the Development, Manufacture, distribution, marketing, promotion, offer for sale, use, import, export or sale of Licensed Products in a regulatory jurisdiction.

- 1.115 "Regulatory Authority" means the FDA (or any successor agency) or any equivalent agency thereof in jurisdictions outside of the US.
- 1.116 "Regulatory Materials" means the regulatory registrations, applications, Regulatory Approvals or other submissions made to or with any Regulatory Authority necessary for the Research, Development (including the conduct of Clinical Trials), Manufacture, or Commercialization of a Licensed Product in a regulatory jurisdiction, together with all related correspondence to or from any Regulatory Authority, including all drug master file(s) (if any).
- 1.117 "Research" means any pre-clinical research activities (including Cell Line research and development activities).
- 1.118 "Royalty Term" means:
- 1.118.1 **Collaboration Off-the-Shelf T-Cell Therapy.** On a Collaboration Off-the-Shelf T-Cell Therapy-by-Collaboration Off-the-Shelf T-Cell Therapy and country-by-country basis [***].
- 1.118.2 **Collaboration Personalised T-Cell Therapy.** On a Collaboration Personalised T-Cell Therapy-by-Collaboration Personalised T-Cell Therapy and country-by-country basis [***].
- 1.119 "Sales" means, on a given Licensed Product by GNE or any of its Affiliates in a given period [***].

[***]:

- 1.119.1 [***];
- 1.119.2 [***];
- 1.119.3 [***];
- 1.119.4 [***]
- 1.119.5 [***].
- 1.120 [***].
- 1.121 "**Segregate**" means, with respect to a Competing Product, to segregate the Research, Development, Manufacture and Commercialization activities relating to such Competing Product from the Research, Development, Manufacture or Commercialization activities with respect to any Licensed Products, including ensuring that: [***].
- 1.122 "Shared Neoantigen" means any Neoantigen that [***].
- 1.123 "Specifications" means [***], as attached hereto as Schedule 1.123 (Specifications).

- 1.124 "**Target**" means a protein or biological molecule from a human or virus from which an Antigen is derived (including all peptides derived from that protein, all forms of such protein or biological molecule).
- 1.125 "T-Cell" means a lymphocyte with detectable surface expression of CD3, and one or both of CD4 and CD8, in each case from an endogenous genetic loci.
- 1.126 "**Territory**" means all the countries of the world.
- 1.127 "**Third Party**" means any entity other than Adaptimmune or GNE or an Affiliate of either.
- 1.128 "**Tumor Associated Antigen**" means either (a) a wild-type human Antigen that is over-expressed or selectively expressed in a human tumor cell or (b) an Antigen arising from non-human proteins such as viral sequences that is expressed in a human tumor cell, in each case of (a) or (b), that is not a Neoantigen.
- 1.129 "US" means the United States of America and its territories and possessions.
- 1.130 "US Net Profits and Net Losses" means, with respect to a Collaboration Off-the-Shelf T-Cell Therapy Directed To a Collaboration Target for which Adaptimmune timely exercised its Opt-In, [***]
- 1.131 "Valid Claim" means, with respect to a particular country:
- 1.131.1 **Issued and Unexpired Patents.** A claim of any issued and unexpired Patent (but excluding patent applications) whose validity, enforceability, or patentability has not been affected by any of the following: (i) irretrievable lapse, abandonment, revocation, dedication to the public or disclaimer [***]; or (ii) a holding, finding, or decision of invalidity, unenforceability, or non-patentability by a court, government authority, national or regional patent office, or other appropriate body that has competent jurisdiction, such holding, finding, or decision being final and unappealable or unappealed within the time allowed for appeal; and
- 1.131.2 **Pending Patent Application.** A claim of a pending patent application that was filed and is being prosecuted in good faith and which has not been (i) pending for more than [***] years from the date of the filing of the first non-provisional application (or an equivalent filing outside of the US) containing such claim or (ii) affected by any of the following: (A) irretrievable lapse, abandonment or revocation; or (B) a holding, finding, or decision of invalidity, unenforceability, or non-patentability by a court, government authority, national or regional patent office, or other appropriate body that has competent jurisdiction, such holding, finding, or decision being final and unappealable or unappealed within the time allowed for appeal.
- 1.132 "VAT" means, in the EU, value added tax calculated in accordance with Council Directive 2006/112/EC and, in a jurisdiction outside the EU, any equivalent tax.
- 1.133 "Working Cell Bank" means a vialed collection of serially subcultivated cells that are derived from a Master Cell Bank used to establish seed cultures of the Cell Line, including to initiate a manufacturing process using such Cell Line.

Additional Definitions

Defined Term	Section
"Adaptimmune"	<u>Preamble</u>
" Adaptimmune Cell Line Agreements"	<u>Section 15.2.13</u>
" Adaptimmune Notice of GNE Discontinuation Event"	Section 17.5.2
" Adaptimmune Platform Improvement IP"	<u>Section 12.1.2</u>
"Adaptimmune Platform IP"	<u>Section 12.1.1</u>
"Adaptimmune Prosecuted Patents"	<u>Section 12.3.3</u>
" Adaptimmune Step-In Right"	<u>Section 12.3.4</u>
" Adaptimmune Third Party IP License"	<u>Section 7.3.2(a)</u>
"Adaptimmune-to-GNE Grantback License"	<u>Section 7.1.1(d)</u>
"Additional Collaboration Target"	<u>Section 3.4.1</u>
"Additional Collaboration Target Designation Fee"	<u>Section 10.3.3</u>
" Additional Research Term"	<u>Section 3.3.1</u>
"Agreement"	<u>Preamble</u>
"Alliance Manager"	Section 2.7
"Allogeneic T-Cells"	<u>Section 3.5.1</u>
"Allogeneic T-Cell Lines"	<u>Section 3.5.1</u>
"Allogeneic T-Cell Milestone"	Section 10.3.2
"Approved Subcontractors"	Section 3.8.1
" Arbitrator"	<u>Section 11.7.5</u>
" Assigning Party"	<u>Section 12.2.3</u>
"Background IP"	<u>Section 12.1.3</u>
"Binding Budget Year"	<u>Section 6.1.3(b)</u>
L	l

Defined Term	Section
"Biosimilar"	<u>Section 10.8.2</u>
"BLA"	Section 1.94
"Cell Line Creation Date"	<u>Section 10.8.7(c)</u>
"Change of Control Notice"	Section 9.2.2
"[***]"	Section 1.18
"Clinical Candidate"	<u>Section 10.3.2</u>
"[***]"	<u>Section 8.3.3</u>
"[***]"	<u>Section 8.3.3</u>
"Collaboration Off-the-Shelf Exclusive License"	Section 7.1.1(b)
"Collaboration OTS Cell Line"	<u>Section 8.3.1</u>
"Collaboration OTS MCB"	<u>Section 8.3.1</u>
"Collaboration Personalised Exclusive License"	<u>Section 7.1.1(c)</u>
"Collaboration Personalised Cell Line"	<u>Section 8.3.2</u>
"Collaboration Personalised MCB"	<u>Section 8.3.2</u>
"Collaboration Target Substitution Right"	<u>Section 3.4.2</u>
"Combination"	Section 1.99.2(b)
"Competing Product"	<u>Section 9.2.1</u>
"control"	Section 1.18
"CPA Firm"	Section 11.7.2
"Create Act"	<u>Section 12.2.5</u>
"Development Cost Share"	Section 6.1.4
" Disclosing Party"	<u>Section 14.6.3</u>
"Disclosure"	Section 14.1

Defined Term	Section
"Disposition Transaction"	<u>Section 10.10</u>
"Dispute"	Section 19.1
"Enforcement"	Section 12.4.3
"Exclusivity Requirements"	Section 9.1
"Execution Date"	<u>Preamble</u>
"Executives"	Section 2.9
"Exercise Notice"	Section 17.6.6(c)
"Existing Indication"	Section 1.79
" First IND"	<u>Section 6.1.2(a)</u>
"First Three Collaboration Off-the-Shelf T-Cell Therapies"	Section 9.1.2
"First Three Collaboration Personalised T-Cell Therapies"	Section 9.1.2
"First Personalised Cell Line"	Section 1.61
"First OTS Cell Line"	Section 1.61
"[***]"	Section 1.18
"[***]"	Section 1.18
"Force Majeure"	Section 20.7
" GAAP"	Section 1.3
"German WHT Requirement"	<u>Section 11.6.2</u>
"GITA"	<u>Section 11.6.2</u>
"Global Development Budget"	Section 1.64
"GNE"	<u>Preamble</u>
"GNE In-Licensed Receptor Agreement"	<u>Section 17.6.6(g)</u>
"GNE Platform Improvement IP"	<u>Section 12.1.4</u>

Defined Term	Section
"GNE Proprietary Platform Methods"	Section 12.1.5
"GNE Prosecuted Patents"	<u>Section 12.3.4</u>
"GNE Receptor IP"	<u>Section 12.1.6</u>
"GNE Step-In Right"	<u>Section 12.3.3</u>
"GNE-Incurred Third Party Payment Obligations"	<u>Section 7.3.2(a)</u>
"GNE-Modified OTS Cell Line"	Section 10.8.7(a)
"GNE-Modified Personalised Cell Line"	Section 10.8.7(a)
"GNE-to-Adaptimmune Grantback License"	Section 7.2.1(b)
" IFRS"	Section 1.3
"Indemnitee"	Section 16.2
"Indemnitor"	Section 16.2
"Infringement"	Section 12.4.1
"Initial Research Term"	<u>Section 3.3.1</u>
"Intent to File Notice"	Section 6.1.2(a)
"IPR"	<u>Section 1.112</u>
"iPSC Genetic Edit Milestone"	Section 10.3.2(b)
"JDC"	<u>Section 2.4.1</u>
"JIPC"	Section 2.3
"JPC"	<u>Section 2.2.1</u>
"JRC"	Section 2.1.1
[***]	<u>Section 12.1.7</u>
[***]	Section 12.1.7
" Launch Cost Share"	Section 6.1.4
Launen Cost Share	<u>Section 6.1.4</u>

Defined Term	Section
" Losses"	Section 16.1.1
"NDA"	Section 1.94
" Negotiation Period"	Section 17.6.6(d)
"Net Sales Report"	Section 11.2
" New Budget"	Section 6.1.8(d)
"New Indication"	Section 1.79
" Nomination"	Section 3.6.1
" Non-Disclosing Party"	<u>Section 14.6.3</u>
" Non-Disclosure Agreement"	Section 13.5
" Notice of GNE Discontinuation Determination"	<u>Section 17.4.2</u>
" Notice of GNE Discontinuation Event"	<u>Section 17.5.1</u>
" Notice of Interest"	Section 17.6.6(a)
"Off-the-Shelf Development Milestone Event"	<u>Section 10.4.1</u>
"Off-the-Shelf Development Milestone Payment"	Section 10.4.1
"Off-the-Shelf Exclusivity Term"	<u>Section 9.1.1</u>
"Off-the-Shelf Net Sales Milestone Event"	<u>Section 10.5.1</u>
"Off-the-Shelf Net Sales Milestone Payment"	Section 10.5.1
"Old Budget"	Section 6.1.8(d)
"Ongoing Activities"	Section 17.6.6(b)
"Opposition Proceeding"	<u>Section 12.4.2</u>
"Opt-In"	Section 6.1.1
" Opt-In Data Package"	Section 6.1.2(a)
"Opt-In Effective Date"	Section 6.1.8(b)

Defined Term	Section
" Opt-In Notice"	Section 6.1.2(b)
" Opt-Out"	Section 6.1.8(a)
" Opt-Out Notice"	Section 6.1.8(a)
"Other Collaboration IP"	<u>Section 12.1.8</u>
" Outstanding Deferred Amount"	Section 6.1.5
"Overlapping Improvement IP"	Section 12.1.9
" Owning Party"	<u>Section 12.2.3</u>
"Party"	<u>Preamble</u>
" Party Vote"	<u>Section 2.5.7(c)</u>
" Patent Term Extensions"	Section 12.9
"[***]"	Section 7.2.1(c)
"Personalised Development Milestone Event"	<u>Section 10.4.2</u>
"Personalised Development Milestone Payment"	Section 10.4.2
"Personalised Exclusivity Term"	<u>Section 9.1.2</u>
"Personalised Net Sales Milestone Event"	<u>Section 10.5.2</u>
"Personalised Net Sales Milestone Payment"	<u>Section 10.5.2</u>
" Pharmacovigilance Agreement"	Section 4.6
"Phase I Materials"	<u>Section 8.1.1</u>
"Phase I Material Supply Agreement"	<u>Section 8.1.2</u>
"Progress Report"	Section 4.3
"Project Co-Leader"	<u>Section 2.2.1</u>
"Proposed Target"	Section 3.6.1
" Quality Agreement"	Section 8.1.3

Defined Term	Section
"Relevant Collaboration Target Information Package"	Section 17.6.6(b)
"Relevant Collaboration Target IP"	Section 17.6.6(d)(i)
"Replaced Collaboration Target"	Section 3.4.2
" Research Plan"	Section 3.2
"Research Program"	Section 3.1
" Research Program Materials"	Section 8.1.1
"Research Term"	Section 3.3.1
"Research Term Extension Fee"	Section 10.2
"Reversion License"	Section 17.6.6(d)(i)
"Review Meeting"	<u>Section 17.6.6(b)</u>
"Roche"	<u>Preamble</u>
" Rules"	Section 19.2.1
"[***]"	<u>Section 12.1.7</u>
"sBLA"	Section 1.94
"Second Personalised Cell Line"	Section 1.61
"Second OTS Cell Line"	Section 1.61
"Sharing Percentages"	Section 6.1.4
"sNDA"	Section 1.94
"Sole Prosecuted Patent"	Section 12.3.1
"[***]"	Section 1.18
"Special Independent Reviewer"	<u>Section 3.6.2(b)</u>
"SPCs"	Section 12.9
" Sub-Committee(s)"	Section 2.2.3

Defined Term	Section
"Sublicense"	Section 7.1.4(a)
"Sublicensee"	Section 7.1.4(a)
"Target Nomination Request"	<u>Section 3.6.1</u>
"TCR"	Background
"T-Cell Therapy"	<u>Background</u>
"Tech Transfer Plan"	Section 8.4
"Term"	Section 17.1
"Third Party Claims"	<u>Section 16.1.1</u>
"Third Party Infringement Claim"	<u>Section 12.5.1</u>
"Third Party IP"	<u>Section 7.3.1</u>
"Third Party IP License"	<u>Section 10.8.1</u>
"Third Party IP Notice"	<u>Section 7.3.1</u>
"Title 11"	Section 17.3
"Transition Agreement"	Section 17.6.6(d)
"US-only Development Costs"	<u>Section 6.1.4(b)</u>
"US Co-Promotion Agreement"	<u>Section 6.1.7</u>
"US Co-Promotion Option"	<u>Section 6.1.7</u>
"US Launch Plan and Budget"	<u>Section 6.1.3(a)</u>
"US Net Profits and Net Losses Share"	<u>Section 6.1.1</u>
"Washout Period"	Section 10.8.7(d)

ARTICLE 2 GOVERNANCE

2.1. **Joint Research Committee.**

- 2.1.1. **Joint Research Committee Formation and Composition**. As soon as reasonably possible and in any event within [***] days after the Effective Date, Adaptimmune and GNE shall establish a joint research committee (the "**JRC**") to provide a forum for the Parties to address any issues arising from the Research Plan and oversee the conduct of the Research Program. The JRC shall be composed of [***] representatives designated by each Party (and the Parties need not have the same number of representatives). Representatives must be appropriate for the tasks then being undertaken and the stage of research, in terms of their seniority, availability, function in their respective organizations, training and experience. Each Party shall designate one of its representatives as its primary JRC contact. Each Party may replace its representatives from time to time by informing the other Party's Alliance Manager in writing (which may be by email); *provided*, that if a Party's representative is unable to attend a meeting, such Party may designate an alternate to attend such meeting by informing the other Party's Alliance Manager in writing (which may be by email) in advance and following submission of such written notification the alternate will be entitled to perform the functions of such representative. The Alliance Managers may attend meetings of the JRC but shall have no right to vote on any decisions of the Committee.
- 2.1.2. **JRC Responsibilities**. In addition to its overall responsibility for overseeing the Research Program, the JRC shall, in particular:
- (a) work with the Project Co-Leaders to coordinate all material research activities performed by each Party and monitor progress of the research activities of the Parties hereunder;
 - (b) review and approve amendments to the Research Plan as proposed by the JPC;
 - (c) review and approve the allocation of responsibility for the Research Program;
- (d) review and approve the research communication and publication strategy as developed by the JPC;
- (e) work to resolve any technical disputes, controversy or claim related to the matters and authority of the JRC;
- (f) determine whether any iPS Cell Line generated during the performance of the Research Plan is [***];
 - (g) perform such other functions as specified in this Agreement; and

(h) perform such other functions as appropriate to further the purposes of this Agreement as agreed by the Parties in writing.

2.2. **Joint Project Committees**.

- 2.2.1. **Formation and Composition**. As soon as reasonably possible and in any event within [***] days after the Effective Date, the Parties shall establish an initial joint project committee (the "JPC") to oversee the performance of the Research Plan. The JRC may establish one or more JPCs in addition to the initial JPC. Each JPC shall be composed of representatives designated by each Party. Representatives must be appropriate for the tasks then being undertaken and the stage of research, in terms of their seniority, availability, function in their respective organizations, training and experience. For each JPC, each Party shall designate one of its representatives as its primary JPC contact (each, a "**Project Co-Leader**"). Each Party may replace its representatives from time to time by informing the other Party in writing (which may be by email); *provided*, that if a Party's representative is unable to attend a meeting, such Party may designate a knowledgeable alternate to attend such meeting and perform the functions of such representative. Each JPC shall be subject to the oversight, review and approval of the JRC, as the case may be.
- 2.2.2. **JPC Responsibilities**. In addition to its overall responsibility for overseeing the performance of the Research Plan, the JPC shall, in particular:
- (a) Approve minor changes to the Research Plan, specifications, timelines or activities assigned to each of the Parties under the Research Plan;
- (b) prepare draft amendments (as needed) to the Research Plan, and submit such draft amended Research Plan to the JRC for approval, as applicable;
- (c) develop a communication and publication plan for publications and public presentations related to the Research Program and submit such plans to the JRC for approval, and implement such approved plan;
- (d) discuss and attempt to resolve any disputed matters related to the research collaboration before referring such matters to the JRC, as the case may be; and
- (e) perform such other functions as agreed to by the JRC or as specified in this Agreement.
- 2.2.3. **Sub-Committees**. From time to time, the JPC may also establish and delegate duties to sub-committees on an "as-needed" basis to oversee particular projects or activities and facilitate collaboration between the Parties, and each such sub-committee shall be constituted and shall operate as the JPC determines ("**Sub-Committee(s)**"). Each such Sub-Committee and its activities shall be subject to the oversight, review and approval of, and shall report to, the JPC. In no event shall the authority of a Sub-Committee exceed that specified for the JPC in this <u>Article 2</u> (Governance).

- 2.3. **JIPC**. On or promptly after the Effective Date, the Parties shall establish a joint intellectual property committee consisting of designated employee representatives from each Party's in-house intellectual property counsel or other designated person appointed by an employee representative from a Party's in-house legal team ("**JIPC**"). The JIPC shall provide a forum for the exchange of information between the Parties in relation to Collaboration IP and any other matter delegated to the JIPC in accordance with <u>Article 12</u> (Intellectual Property; Ownership), and shall [***]. For the avoidance of doubt, the JIPC may [***].
 - 2.3.1. **Role of the JIPC**. The JIPC will be responsible for: [***].

2.4. **Joint Development Committee.**

- 2.4.1. **Formation and Composition**. Following [***], as soon as reasonably possible and in any event within [***] days thereafter, the Parties shall establish a joint development committee (the "**JDC**"), which will serve as a forum for the exchange of information with respect to the Development of the Collaboration Off-the-Shelf T-Cell Therapies Directed To each Collaboration Target for which an Opt-In is in effect. The JDC shall be composed of representatives designated by each Party. Representatives must be appropriate for the tasks then being undertaken in terms of their seniority, availability, function in their respective organizations, training and experience. Each Party shall designate one of its representatives as its primary JDC contact. Each Party may replace its representatives from time to time by informing the other Party's Alliance Manager in writing (which may be by email); *provided*, that if a Party's representative is unable to attend a meeting, such Party may designate an alternate to attend such meeting by informing the other Party's Alliance Manager in writing (which may be by email) in advance and following submission of such written notification the alternate will be entitled to perform the functions of such representative. The Alliance Managers may attend meetings of the JDC.
- 2.4.2. **JDC Responsibilities**. In addition to serving as a forum for the exchange of information and discussion of issues, in each case with respect to the Development of the Collaboration Off-the-Shelf T-Cell Therapies Directed To each Collaboration Target for which an Opt-In is in effect, the JDC will review, exchange and discuss the following information and issues in particular:
 - (a) GNE will provide to the JDC updates [***];
 - (b) GNE will provide the JDC with all updates (or draft updates, as available) [***];
 - (c) GNE will provide the JDC with [***];
- (d) Adaptimmune, through the JDC, may provide its comments to GNE regarding the Development (and, eventually, the US launch) of the applicable Collaboration Off-the-Shelf T-Cell Therapies, which [***].

2.5. **Meetings**.

- 2.5.1. **JRC**. The JRC shall meet [***] (unless otherwise agreed by the Parties) and at such other times as deemed appropriate by the JRC. The presence of at least [***] JRC members designated by each Party shall constitute a quorum at a JRC meeting. The JRC may meet in person or via teleconference or otherwise, in each case as agreed by the JRC, *provided*, that at least one (1) meeting per calendar year shall be held in person, unless otherwise agreed by the Parties or otherwise prevented by any travel restrictions, isolation requirements or other governmental guidance recommending against travel or in-person meetings.
- 2.5.2. **JPC and Sub-Committee(s)**. The JPC shall meet [***] by audio or video teleconference or as otherwise agreed by the JPC. Each Sub-Committee shall meet at least as often as determined by the JPC by audio or video teleconference or as otherwise agreed by such Sub-Committee. The presence of at least one JPC member or Sub-Committee member designated by each Party shall constitute a quorum at a JPC meeting or a Sub-Committee meeting, as applicable.
- 2.5.3. **JIPC**. The JIPC will meet regularly as agreed by the JIPC and may meet in person or by audio or video teleconference or as otherwise agreed by the JIPC. The presence of at least one (1) JIPC member designated by each Party shall constitute a quorum at a JIPC meeting, as applicable. The Parties agree that all Intellectual Property-related discussions conducted by the JIPC are common interest disclosures as described in <u>Section 12.8</u> (Common Interest Disclosures).
- 2.5.4. **JDC**. The JDC members will meet [***] times per year (provided that the Parties may otherwise agree that additional *ad hoc* meeting(s) are required from time to time) and may meet in person or by audio or video teleconference or as otherwise agreed by the JDC. The presence of at least one (1) JDC member designated by each Party shall constitute a quorum at a JDC meeting, as applicable.
- 2.5.5. **Meeting Agendas and Minutes**. Not later than [***] days after the JRC, the initial JPC, the JIPC, and the JDC are formed, the JRC, JPC, JIPC and JDC shall each hold an organizational meeting by video or teleconference to establish their respective operating procedures, including establishment of agendas, and preparation and approvals of minutes. The Parties shall alternate the taking of minutes recording in writing all decisions made, action items assigned or completed, and other appropriate matters for Committees. GNE shall take the first set of minutes for any Committee meeting. Meeting minutes shall be sent to both Parties promptly after a meeting for review, comment and approval by each Party. A decision that is made at a Committee meeting shall be recorded in minutes (unless otherwise agreed by the Parties), and decisions that are made by a Committee outside of a meeting shall be documented in writing and be shown to be clearly agreed by all representatives of the applicable Committee, as relevant.
- 2.5.6. **General**. Employees of each Party other than Committee or Sub-Committee representatives, including, for the avoidance of doubt each Party's Alliance Manager, may attend meetings of such Committee or Sub-Committee as non-voting participants. Additionally, with the consent of the other Party, a Party's consultants and advisors involved in the Research Program may attend meetings of such Committee or Sub-Committee as non-voting observers; *provided*, that such consultants and advisors are under obligations of confidentiality and non-use applicable to the Confidential Information of the other Party as required by Article 13 (Confidentiality) and each Party shall have the right to excuse the other Party's consultants and advisors from a meeting

at any time. Each Party shall be responsible for all of its own expenses of attending or otherwise participating in each Committee and Sub-Committee, including travel and related costs.

2.5.7. **Decision-Making**.

- (a) **JPC**. Each Party will discuss and attempt to resolve any potential or evolving disagreement related to the Research Program through its respective Project Co-Leaders before it is brought before the JPC. With respect to the responsibilities of the JPC, each Party shall [***] on all matters brought before the JPC. The JPC shall operate as to matters within its responsibility by unanimous Party Vote. If the JPC is unable to achieve unanimous Party Vote within [***] after the dispute matter is brought to vote before the JPC or such longer period as the Project Co-Leaders agree, such matter shall be referred to the JRC for resolution.
- (b) **JIPC**. All decisions of the JIPC on matters within its responsibility will be made by unanimous vote, with each Party having [***]. Subject to <u>Section 12.2.2</u> (Inventorship), as applicable, any unresolved disagreement or dispute arising at the JIPC will be resolved as follows: [***].
- (c) JRC. Each Party will discuss and attempt to resolve any potential or evolving disagreement related to the Research Program through the JPC in accordance with Section 2.5.7(a) (JPC) before it is brought before the JRC. Each Party's designees on the JRC shall, [***] on all matters brought before the JRC shall operate as to matters within its responsibility by unanimous Party Vote.
 - (d) **JDC**. [***].
- 2.6. **Dissolution of the JPC and JRC; Functionally Distinct Determinations**. Upon the earlier of expiration or termination of the Research Program or the mutual agreement of the Parties, the JRC or JPC will have no further responsibilities or authority under this Agreement, and the JRC or JPC will be deemed dissolved by the Parties. Thereafter, the Parties shall determine [***]. Following expiration or termination of Research Program, [***].
- 2.7. Alliance Managers. Promptly following the Effective Date, each Party shall designate an individual to act as the primary business contact for such Party for matters related to this Agreement (such Party's "Alliance Manager"), unless another contact is expressly specified in this Agreement or designated by a Committee for a particular purpose. The Alliance Managers shall facilitate the flow of information and collaboration between the Parties and assist in the resolution of potential and pending issues and potential disputes in a timely manner to enable the JRC and the Parties to reach consensus and avert escalation of such issues or potential disputes. Either Party may replace its Alliance Manager at any time by informing the other Party's Alliance Manager in writing (which may be by email). Each Party shall ensure that its Alliance Manager is capable of performing the obligations required of an Alliance Manager under this Agreement.
- 2.8. **Limitations on Authority**. Each Party shall retain the rights, powers, and discretion granted to it under this Agreement, and no such rights, powers, or discretion shall be delegated to or vested in a Committee unless such delegation or vesting of rights is expressly provided for in this Agreement or the Parties expressly so agree in writing. No Committee shall have the power to

amend, modify or waive compliance with this Agreement, which may only be amended or modified, or compliance with which may only be waived, in each case solely as provided in <u>Section 20.11</u> (Amendment; Waiver).

- 2.9. **Escalation**. If the Alliance Managers are unable to assist the JRC in resolving a technical dispute within [***] after the dispute is first referred to the Alliance Managers, or such longer period as the Parties may agree, either Party may elect to submit such issue to the CBO for Adaptimmune (or a person in an equivalent position at Adaptimmune), and a vice president of research or development for GNE. These executives are referred to collectively as the "**Executives**".
- 2.10. **Final Resolution**. In the event that the Executives are unable to resolve a given issue referred to them in accordance with Section 2.9 (Escalation) within [***] after the dispute is first referred to the Executives, then, subject to Section 2.8 (Limitations on Authority), [***] would have final decision-making authority with respect to (a) [***], provided, that in each case [***] would not have the right, by virtue of its final decision-making authority, to (x) require [***] to take any action that would (i) [***]. In the event that the Executives are unable to resolve whether [***] within [***] after the dispute is first referred to the Executives, then, the Parties shall refer such matter to an independent impartial expert for determination pursuant to the procedures set forth in **Schedule 2.10** (Expert Determination of [***]). Notwithstanding the foregoing, each Party must consider the other Party's position in good faith when exercising its final dec ision making authority pursuant to this Section 2.10 (Final Resolution).

ARTICLE 3 RESEARCH PROGRAM

- 3.1. **Research Program**. During the Research Term, GNE and Adaptimmune will collaborate in the conduct of research activities to create Allogeneic T-Cell Lines using Adaptimmune iPS Cell Lines for use by GNE to Research and Develop (a) Collaboration Off-the-Shelf T-Cell Therapies that are Directed To up to five (5) Collaboration Targets, and (b) Collaboration Personalised T-Cell Therapies (collectively the "**Research Program**") in accordance with the Research Plan and as further described in this <u>Article 3</u> (Research Program). Each Party shall comply with all applicable laws, rules and regulations in the conduct of the Research Program. Each Party shall, in performing its obligations under the Research Program, assign responsibilities to those portions of its organization that have the appropriate resources, expertise and responsibility for such obligations. Subject to <u>Section 3.7</u> (FTE Funding for Research Program), each Party shall be responsible for its own costs associated with the activities it conducts under the Research Program.
- 3.2. **Research Plan**. The initial research plan (" **Research Plan**") is attached to this Agreement as Exhibit B (Research Plan). The JRC may amend in writing the Research Plan from time to time in order to determine what Cell Line specifications and modifications are needed during the course of performing the Research Program. The Research Plan shall be reviewed and updated by the JRC (a) following the addition of any Additional Collaboration Target or the replacement of a Replaced Collaboration Target to the Research Program to set forth activities required to create the Collaboration Off-the-Shelf T-Cell Therapies Directed To such Additional Collaboration Target, and (b) to set forth activities to be conducted to create any Functionally Distinct

Collaboration Off-the-Shelf T-Cell Therapy or Functionally Distinct Collaboration Personalised T-Cell Therapy. Each Party will use Commercially Reasonable Efforts to carry out the activities allocated to such Party under the Research Plan.

3.3. Research Term.

- 3.3.1. **Initial Research Term**. The Research Program shall be carried out during the initial eight (8) year period following the Effective Date ("**Initial Research Term**"), provided that the Initial Research Term may be extended by GNE for [***] additional [***] year periods (as described in <u>Section 3.3.2</u> (Research Term Extension)) by payment of the Research Term Extension Fee (as described in <u>Section 10.2</u> (Research Term Extension Fee)) for each such additional [***] year period (each such [***] year time period an "**Additional Research Term**" and each, collectively with the Initial Research Term, the "**Research Term**").
- 3.3.2. **Research Term Extension**. GNE shall have the right to extend the Initial Research Term by up to [***] Additional Research Terms by delivery of at least [***] months written notice to Adaptimmune prior to the expiration of the Initial Research Term or first Additional Research Term, as applicable, and payment of the Research Term Extension Fee in accordance with <u>Section 10.2</u> (Research Term Extension Fee).
- 3.3.3. **Research Term Wind Down**. GNE shall have the right to extend the Research Term by up [***] months by delivery of at least [***] months written notice to Adaptimmune prior to the expiration of the Initial Research Term or Additional Research Term, as applicable, solely to wind down any planned activities pursuant to the Research Plan that have not been completed or are not anticipated to be completed as of the date the Initial Research Term or Additional Research Term, as applicable, was originally scheduled to expire. The Parties shall meet promptly after notice of any such notice to coordinate wind-down efforts.

3.4. Conduct of Research Program.

3.4.1. Collaboration Targets. As of the Effective Date, the Parties will begin developing Collaboration Off-the-Shelf T-Cell Therapies that are Directed To the [***] Initial Collaboration Targets. GNE shall have the right to nominate Proposed Targets to be designated as additional Collaboration Targets for a maximum of [***] such additional Collaboration Targets (each such additional Collaboration Target an "Additional Collaboration Target") for which Collaboration Off-the-Shelf T-Cell Therapies would be developed as part of the Research Program, for a maximum of five (5) Collaboration Targets at any time under this Agreement. To exercise its right to nominate an Additional Collaboration Target, GNE shall provide notice to Adaptimmune's Alliance Manager in accordance with Section 3.6 (Nomination); provided, however, that any such notice must be delivered prior to the expiration of the Research Term . If, following the process described in Section 3.6 (Nomination), such Proposed Target becomes an Additional Collaboration Target, then GNE would pay Adaptimmune the Additional Collaboration Target Designation Fee described in Section 10.3.3 (Additional Collaboration Targets). GNE shall have the right to substitute Collaboration Targets in accordance with Section 3.4.2 (Collaboration Target Substitution).

3.4.2. Collaboration Target Substitution. On a Collaboration Target-by-Collaboration Target basis, during the period of time beginning (a) [***], GNE shall have the right, for any reason and at no cost to GNE, to substitute such Collaboration Target for an Available Target (such right the "Collaboration Target Substitution Right") (each such Collaboration Target that is replaced by an Available Target thereafter a "Replaced Collaboration Target") by delivery of a Nomination notice to Adaptimmune's Alliance Manager in accordance with Section 3.6 (Nomination), after which such Replaced Collaboration Target shall thereafter not be a Collaboration Target; provided, however, that there will in no event be more than five (5) Collaboration Targets in total under this Agreement at any time during the Research Term.

3.5. Cell Line Development.

3.5.1. **Allogeneic T-Cells**. In the performance of the Research Program during the Research Term, the Parties anticipate developing Cell Lines that are differentiated from Adaptimmune iPS Cell Lines that may serve as the basis of T-Cell Therapies (such differentiated iPS Cells " **Allogeneic T-Cells**" and such Cell Lines " **Allogeneic T-Cell Lines**", including progeny and clones of each). Adaptimmune will use Commercially Reasonable Efforts to create, engineer, and differentiate [***] Allogeneic T-Cells that include the JRC approved edits and meet the JRC approved specifications for use in Collaboration Off-the-Shelf T-Cell Therapies, and [***] Allogeneic T-Cells that include the JRC approved edits and meet the JRC approved specifications for use in Collaboration Personalised T-Cell Therapies. [***].

3.5.2. [***]

3.6. **Nomination**.

Proposed Targets. In the event that GNE wishes to (a) select a Target as an Additional Collaboration Target in accordance with Section 3.4.1 (Collaboration Targets), or (b) exercise its Collaboration Target Substitution Right under Section 3.4.2 (Collaboration Target Substitution), then in each case GNE shall provide written notice to Adaptimmune's Alliance Manager (each such notice a " Target Nomination Request"), such Target Nomination Request to include the identity of such Target (each a " Proposed Target") and which of the GNE rights described in (a)- (b) above GNE is exercising. Adaptimmune's Alliance Manager shall maintain the identity of the Targets within such Target Nomination Request as strictly confidential and shall be responsible for determining whether the Proposed Target is an Available Target or an Excluded Target . If such Proposed Target is an Excluded Target, then Adaptimmune's Alliance Manager will notify GNE's Alliance Manager of such promptly [***] and GNE shall not be entitled to nominate such Target . If the Proposed Target is an Available Target, then Adaptimmune's Alliance Manager will notify GNE's Alliance Manager of such promptly ([***]) and GNE shall have a further [***] days from date of receipt by GNE of the confirmation of availability by Adaptimmune of such Proposed Target to exercise its applicable right described in (a)- (b) above with respect to such Available Target by provision of written notice to Adaptimmune identifying the Target and which of the GNE rights described in (a)- (b) above GNE is exercising (each such notice a "Nomination"). Upon Nomination, such Available Target will become a Collaboration Target for purposes of this Agreement.

3.6.2. Information on Excluded Target; Special Independent Reviewer.

(a) Where Adaptimmune's Alliance Manager notifies GNE that a Proposed Target is an Excluded Target, GNE may request further details as to why such Target is an Excluded Target, and in such event Adaptimmune shall provide reasonable further details to GNE's Alliance Manager (together with supporting documentation, if reasonably requested by GNE's Alliance Manager) on a timely basis, or to a Special Independent Reviewer, at Adaptimmune's election.

3.7. [***].

3.8. **Subcontractors**.

3.8.1. **Third Party Subcontracting.** GNE may subcontract portions of its work under the Research Program to Affiliates or Third Parties; *provided*, that such subcontract is consistent with the terms and conditions of this Agreement. Adaptimmune may subcontract portions of its work under the Research Program to Affiliates and to the Third Parties listed on **Schedule 3.8** (as such list may be amended from time to time by mutual agreement) (" **Approved Subcontractors**"); *provided further*, that in each case such subcontract is consistent with the terms and conditions of this Agreement. Except for the Approved Subcontractors, Adaptimmune may not subcontract any portion of its work under the Research Program to any Third Parties without GNE's prior written consent, such consent not to be unreasonably withheld, conditioned or delayed; *provided*, that if GNE does not object to a subcontractor proposed by Adaptimmune within [***] days of receipt of notice of such proposed subcontractor, GNE will have deemed to have consented to Adaptimmune's use of such subcontractor.

3.8.2. New Subcontractors. [***].

3.8.3. **Confidential Information.** Notwithstanding the foregoing or any other provision in this Agreement, to the extent that any disclosure of Confidential Information of a Party to a subcontractor includes trade secret information specifically identified by such Party as a trade secret in writing in advance to the other Party (*e.g.*, information relating to the Adaptimmune Platform or GNE Proprietary Platform Methods specifically identified as a trade secret in writing in advance to the other Party), the Party owning such trade secret information shall be entitled to request additional protection measures from any Third Party subcontractor, including entry into a direct agreement of confidentiality with such Third Party subcontractor. Each Party shall remain responsible (at its cost) for and shall ensure that each of such Party's subcontractors comply with the terms and conditions of this Agreement, to the extent applicable to such subcontractor.

3.9. **Reports; Records**.

3.9.1. **Research Program Progress Reports**. Each Party shall reasonably keep the other Party informed of its activities under the Research Program and shall provide to the other Party's representatives on the JRC regular summary updates at each meeting. If reasonably

necessary for a Party to perform its work under the Research Program, that Party may request that the other Party provide more detailed information and data regarding the updates it earlier provided, and the other Party shall promptly provide the requesting Party with information and data as is reasonably available and reasonably necessary to conduct the Research Program, and such other information as the Parties agree. Subject to Section 13.2 (Exclusions Regarding Confidential Information), all such reports, information and data provided by a Party shall be considered the providing Party's Confidential Information.

- 3.9.2. **Research Records**. Each Party shall maintain records of the Research Program (or cause such records to be maintained) in sufficient detail and in good scientific manner as will properly reflect all work done and results achieved by or on behalf of such Party in the performance of the Research Program. All laboratory notebooks shall be maintained for no less than the term of any Patent issuing therefrom. All other records shall be maintained by each Party during the Term in accordance with such Party's standard policies for retention of such records. All such records of a Party shall be considered such Party's Confidential Information.
- 3.10. **Research Efforts**. The Parties shall use Commercially Reasonable Efforts to conduct their respective tasks under the Research Program.

ARTICLE 4 DEVELOPMENT

- 4.1. **General**. GNE shall have the sole right and authority to Develop all Licensed Products in the Territory.
- 4.2. **Development Efforts**.
- 4.2.1. **Collaboration Off-the-Shelf T-Cell Therapy.** GNE shall use Commercially Reasonable Efforts to Develop at [***] Collaboration Off-the-Shelf T-Cell Therapy Directed To each Collaboration Target in the [***].
- 4.2.2. **Collaboration Personalised T-Cell Therapy.** GNE shall use Commercially Reasonable Efforts to Develop at [***] Collaboration Personalised T-Cell Therapy in the [***].
- 4.3. **Progress Reports**. GNE shall, for a given Licensed Product that is not a Collaboration Off-the-Shelf T-Cell Therapy Directed To a Collaboration Target for which an Opt-In is in effect, provide to Adaptimmune a written report ("**Progress Report**"), [***] per calendar year by no later than [***] during the period beginning from the earlier of (a) the disbandment of the JRC or (b) completion of all activities under the Research Program with respect to such Licensed Product, and ending upon the achievement of all potential Development Milestone Events with respect to such Licensed Product. Each Progress Report for a given Licensed Product will contain [***]. Where there has been no change from any previous report, GNE may simply confirm the same rather than providing a further written report. All Progress Reports, and the information and data contained therein, shall be considered GNE's Confidential Information. Following provision of any Progress Report by GNE, [***], Adaptimmune may [***].

- 4.4. **Development Funding**. GNE shall be responsible for paying all Development Costs incurred by GNE with respect to (a) each Collaboration Off-the-Shelf T-Cell Therapy that is Directed To a Collaboration Target for which Adaptimmune has not exercised its Opt-In, and (b) all Collaboration Personalised T-Cell Therapies.
- 4.5. **Regulatory**. GNE shall have the sole right and responsibility to prepare and submit regulatory documentation to all Regulatory Authorities in the Territory for all Licensed Products . GNE shall be the sponsor of all Clinical Trials conducted for the Licensed Products and as such, have the sole right to conduct all interactions with Regulatory Authorities with respect to all Licensed Products. Adaptimmune shall support GNE, as may be reasonably necessary, in preparing, submitting and obtaining such regulatory documentation, and in the activities in support thereof, including providing information, documents or other materials (a) [***] that is necessary for GNE to obtain or maintain Regulatory Approvals of such applicable Collaboration Off-the-Shelf T-Cell Therapy or Collaboration Personalised T-Cell Therapy. Additionally, at GNE's election GNE may provide Adaptimmune with copies of filings of Regulatory Materials for Licensed Products and request Adaptimmune's review and comment on such materials. Following completion of the Research Program, Adaptimmune shall continue to provide reasonable assistance to GNE in the provision of information described in this Section 4.5 (Regulatory), to the extent required by any Regulatory Authority, and shall use Commercially Reasonable Efforts to provide such information and assistance within any reasonable timeframes requested by GNE.
- 4.6. **Safety Data Exchange**. Beginning on or about the date on which the first Licensed Product becomes the subject of a Clinical Trial, to the extent required by either Party, the Parties will negotiate and enter into a pharmacovigilance agreement that defines the Parties' responsibilities and obligations with respect to the procedures and timeframes for compliance with applicable law pertaining to safety reporting for such Licensed Product (" **Pharmacovigilance Agreement**"). [***].
- 4.7. [***].

ARTICLE 5 COMMERCIALIZATION

5.1. **General**. Subject to Adaptimmune's right to co-promote a Collaboration Off-the-Shelf T-Cell Therapy directed to a Collaboration Target for which Adaptimmune has timely exercised its Opt-In in accordance with Section 6.1 (Adaptimmune Opt-In) and as further described in, and in accordance with, the US Co-Promotion Agreement, GNE shall have the sole right and authority to Commercialize all Licensed Products in the Territory.

5.2. Commercialization Efforts.

5.2.1. **Collaboration Off-the-Shelf T-Cell Therapy**. GNE shall use Commercially Reasonable Efforts to (a) seek Marketing Approval for [***] Collaboration Off-the-Shelf T-Cell Therapy Directed To each Collaboration Target in the [***]; and (b) if obtained, maintain such Marketing Approval and Commercialize the relevant Collaboration Off-the-Shelf T-Cell Therapy subject to the terms and conditions of such Marketing Approval.

- 5.2.2. **Collaboration Personalised T-Cell Therapy.** GNE shall use Commercially Reasonable Efforts to (a) seek Marketing Approval for [***] Collaboration Personalised T-Cell Therapy in the [***]; and (b) if obtained, maintain such Marketing Approval and Commercialize the relevant Collaboration Personalised T-Cell Therapy subject to the terms and conditions of such Marketing Approval.
- 5.3. **Licensed Products**. Prior to and following Regulatory Approval of each Licensed Product, GNE (itself or through its Affiliates or Sublicensees) shall be solely responsible for and control all Commercialization activities with respect to such Licensed Product subject to Commercialization activities assigned to be performed under the US Co-Promotion Agreement.

ARTICLE 6 ADAPTIMMUNE OPT-IN RIGHT

6.1. **Adaptimmune Opt-In**.

6.1.1. **Opt-In**. Subject to and in accordance with the remainder of this <u>Article 6</u> (Adaptimmune Opt-In Right), GNE hereby grants to Adaptimmune an exclusive option, exercisable in Adaptimmune's sole discretion [***], to fund a percentage of Development Costs and Launch Costs for Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target and, in exchange, share the US Net Profits and Net Losses (" **US Net Profits and Net Losses Share**") with respect to such Collaboration Off-the-Shelf T-Cell Therapies in accordance with <u>Section 10.9</u> (Economic Effect of Opt-In; Profit-Share).

6.1.2. **Opt-In Exercise**.

- (a) For each Collaboration Target, GNE shall provide Adaptimmune written notice of GNE's intent to file the first IND in the US (or the first equivalent filing in any other country, whichever occurs earlier) in relation to the first Collaboration Off-the-Shelf T-Cell Therapy that is Directed To such Collaboration Target (such notice, an "Intent to File Notice" and, such IND (or the equivalent thereof), the "First IND"). Within [***] after GNE provides the Intent to File Notice for a given Collaboration Target, GNE shall also make available to Adaptimmune the Opt-In Data Package for such Collaboration Target, which may be made available in a virtual data room. [***].
- (b) Adaptimmune shall [***] days from the later of delivery of such Intent to File Notice or delivery of such complete Opt-In Data Package to provide written notice to GNE that Adaptimmune wishes to exercise the Opt-In for such Collaboration Target (such notice, an " **Opt-In Notice**").
- (c) If an Opt-In is timely exercised for a given Collaboration Target, the Development Cost Share, Launch Cost Share and US Net Profits and Net Losses Share with respect to such Collaboration Target shall take effect beginning from the date of filing of the First IND for such Collaboration Target (the "Opt-In Effective Date") and continue for the Opt-In Term.

- 6.1.3. **Global Development Plan and Global Development Budget**. With respect to each Collaboration Target for which an Opt-In is in effect, <u>Section 6.1.3</u> (Global Development Plan and Global Development Budget) shall apply.
- (a) [***] of each calendar year within the Opt-In Term for such Collaboration Target, GNE shall provide to the JDC:
- (i) a copy of the final Global Development Plan and Global Development Budget [***]; and
- (ii) if a summary description of US launch activities and a budget therefor for such Collaboration Target are not included the Global Development Plan and Global Development Budget for a given year, a copy of the final US Launch Plan and Budget for the [***].
- (b) Global Development Plans and Global Development Budgets provided by GNE shall be prepared in good faith but not be binding on the Parties, except that the budget included in each final Global Development Budget provided by GNE under Section 6.1.3(a) (Global Development Plan and Global Development Budget) for the [***].
 - (c) [***].
- (d) Without limiting the generality of <u>Section 6.1.3(c)</u> (Global Development Plan and Global Development Budget), GNE shall have the right to make changes to a current or future calendar year Global Development Plan or Global Development Budget at any time[***].
 - (e) [***].
- 6.1.4. **Development and Launch Cost Sharing.** With respect to each Collaboration Target for which an Opt-In is in effect, the Parties shall share the Development Costs (the "**Development Cost Share**") and Launch Costs ("**Launch Cost Share**") for the Development and launch of all Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target in the following proportions (such percentages, the "**Sharing Percentages**"):
 - (a) [***]
 - (b) [***].
 - (c) [***].
- 6.1.5. **Annual Budget Overruns;** [***]. With respect to each Collaboration Target for which an Opt-In is in effect, <u>Section 6.1.5</u> (Annual Budget Overruns; [***]) shall apply. For any calendar year within the applicable Opt-In Term, (i) [***].
- 6.1.6. **Reports; Payment**. With respect to each Collaboration Target for which an Opt-In is in effect, GNE will provide Adaptimmune with a written report within [***] days after the end of each calendar quarter [***] of all Collaboration Off-the-Shelf T-Cell Therapies that are

Directed To such Collaboration Target in the just-ended calendar quarter. Each such report will specify [***] during such calendar quarter and shall include an invoice for payment due from Adaptimmune for such calendar quarter, which will be due [***] days after delivery of such invoice, [***]. At the request of Adaptimmune, GNE will provide additional reasonable supporting documentation to support its calculations of the invoiced amount.

- 6.1.7. US Co-Promotion Agreement. With respect to each Collaboration Target for which an Opt-In is in effect, GNE will provide Adaptimmune with written notice of GNE's intent to commence US launch activities with respect to the first Collaboration Off-the-Shelf T-Cell Therapy that is Directed To such Collaboration Target by no later than [***] prior to the anticipated commencement of such activities. During the [***] days following its receipt of such notice, Adaptimmune shall have the right to exercise an option to co-detail in the US with GNE all Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target, pursuant to the US Co-Promotion Agreement, by delivery of written notice to GNE (such option, the "US Co-Promotion Option"). If the US Co-Promotion Option is timely exercised, the Parties shall negotiate in good faith the terms of an agreement regarding Adaptimmune's co-detailing activities in the US for the Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target (" US Co-Promotion Agreement"). In negotiating the appropriate terms of any US Co-Promotion Agreement, including the allocation of obligations, roles and responsibilities for the US codetailing of the relevant Collaboration Off-the-Shelf T-Cell Therapies, the Parties shall take into account, among other things, [***]. For a given US Co-Promotion Option, if the Parties are unable to agree on the codetailing terms of the US Co-Promotion Agreement within [***] of Adaptimmune's exercise thereof, then either Party may request that the impasse be resolved through "baseball" arbitration in accordance with the procedure set forth on Schedule 6.1.7 (Baseball-Style Arbitration), wherein each Party will provide the arbitrator with a draft US Co-Promotion Agreement and the arbitrator will be instructed to select the US Co-Promotion Agreement that is the most commercially reasonable and balanced of the two draft US Co-Promotion Agreements, that limits the scope of the activities to be performed thereunder to detailing activities only (and not other elements of Commercialization) and that does not impact or restrict any of GNE's decision-making authority under this Agreement. [***].
- 6.1.8. **Opt-Out**. With respect to each Collaboration Target for which an Opt-In is in effect, Section 6.1.8 (Opt-Out) shall apply.
- (a) Adaptimmune shall have the right, in its sole discretion, to opt out of further participation in both the Development and co-detailing of all of the Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target (" **Opt-Out**") at any time by providing written notice to GNE of Adaptimmune's decision to opt out (such notice, the " **Opt-Out Notice**").
- (b) Except as modified in the circumstances described in <u>Section 6.1.8(c)</u> (Opt-Out), the Opt-Out shall take effect as follows (such date, "**Opt-Out Effective Date**"): [***]. Except as modified in the circumstances described in <u>Section 6.1.8(d)</u> (Opt-Out), Adaptimmune shall remain responsible for its Sharing Percentage of Development Cost Share and Launch Cost Share incurred prior to the Opt-Out Effective Date.

- (c) If at any time Adaptimmune fails to timely pay its Sharing Percentage of Development Cost Share or Launch Cost Share in accordance with <u>Section 6.1.6</u> (Reports; Payment) with respect to such Collaboration Target, and such failure to pay is not cured in accordance with <u>Section 17.2</u> (Termination by Either Party for Material Breach), then Adaptimmune shall have been deemed to exercise the Opt-Out with respect to such Collaboration Target, effective as of the last day of the cure period.
 - (d) [***]
- (e) In the event that Adaptimmune exercises the Opt-Out by giving an Opt-Out Notice, then, as of the Opt-Out Effective Date (and on a going-forward, non-retroactive basis), all Collaboration Off-the-Shelf T-Cell Therapies Directed To such Collaboration Target:
- (i) will no longer be subject to the Off-the-Shelf Development Milestone Payments in Section 10.4.1(b) (Adaptimmune Opt-In) and will instead again be subject to the Off-the-Shelf Development Milestone Payments in Section 10.4.1(a) (No Adaptimmune Opt-In);
- (ii) will no longer be subject to the Off-the-Shelf Net Sales Milestone Payments in Section 10.5.1(b) (Adaptimmune Opt-In) and will instead again be subject to the Off-the-Shelf Net Sales Milestone Payments in Section 10.5.1(a) (No Adaptimmune Opt-In);
- (iii) will no longer be subject to the royalties described in Section 10.7.1(b) (Adaptimmune Opt-In) and instead GNE shall pay Adaptimmune the royalties set forth in Section 10.7.1(a) (No Adaptimmune Opt-In) with the rates therein [***] on Annual Net Sales of Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target that occur during the applicable Royalty Term in the Territory [***]; and
- (iv) will no longer be subject to the US Net Profits and Net Losses under <u>Section</u> 10.9 (Economic Effect of Opt-In; Profit-Share).
- (f) In the event that Adap timmune is deemed to have Opted-Out under Section 6.1.8(c) (Opt-Out), then, as of the Opt-Out Effective Date (and on a going-forward, non-retroactive basis), all Collaboration Off-the-Shelf T-Cell Therapies Directed To such Collaboration Target:
- (i) will no longer be subject to the Off-the-Shelf Development Milestone Payments in Section 10.4.1(b) (Adaptimmune Opt-In) and will instead again be subject to the Off-the-Shelf Development Milestone Payments in Section 10.4.1(a) (No Adaptimmune Opt-In);
- (ii) will no longer be subject to the Off-the-Shelf Net Sales Milestone Payments in Section 10.5.1(b) (Adaptimmune Opt-In) and will instead again be subject to the Off-the-Shelf Net Sales Milestone Payments in Section 10.5.1(a) (No Adaptimmune Opt-In);
- (iii) will no longer be subject to the royalties described in <u>Section 10.7.1(b)</u> (Adaptimmune Opt-In) and instead GNE shall pay Adaptimmune the royalty rates set forth in <u>Section 10.7.1(a)</u> (No Adaptimmune Opt-In). [***]; and

(iv) will no longer be subject to the US Net Profits and Net Losses under <u>Section</u> 10.9 (Economic Effect of Opt-In; Profit-Share).

ARTICLE 7 LICENSES

7.1. License from Adaptimmune to GNE.

- 7.1.1. **License Grants**. Subject to the terms and conditions of this Agreement (including Section 7.1.3 (GNE Covenants and Rights)), Adaptimmune hereby grants to GNE:
- (a) a royalty-free, non-transferable (subject to <u>Section 20.4</u> (Assignment)), worldwide, sublicensable, non-exclusive license under Adaptimmune's Background IP and all Collaboration IP owned by Adaptimmune to the extent necessary for GNE to conduct the Research Program.
- (b) on a Collaboration Target-by-Collaboration Target basis, an exclusive (even as to Adaptimmune, except as necessary to perform its obligations and exercise its rights under this Agreement), transferrable (pursuant to Section 20.4 (Assignment)), sublicensable (pursuant to Section 7.1.4 (Sublicenses and Subcontractors)) license, under the Adaptimmune Licensed IP, to Research, Develop, Manufacture, have Manufactured, Commercialize, make, have made, use, sell, offer for sale, import and export the Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target in the Field in the Territory (the "Collaboration Off-the-Shelf Exclusive License").
- (c) an exclusive (even as to Adaptimmune, except as necessary to perform its obligations and exercise its rights under this Agreement), transferrable (pursuant to Section 20.4 (Assignment)), sublicensable (pursuant to Section 7.1.4 (Sublicenses and Subcontractors)) license, under the Adaptimmune Licensed IP, to Research, Develop, Manufacture, have Manufactured, Commercialize, make, have made, use, sell, offer for sale, import and export Collaboration Personalised T-Cell Therapies in the Field in the Territory (the "Collaboration Personalised Exclusive License").
- (d) a perpetual, irrevocable, non-exclusive, royalty-free, fully-paid up, freely transferrable and freely sublicensable license under all of Adaptimmune's interest in, to and under Adaptimmune Platform Improvement IP [***], in each case for any use (" Adaptimmune-to-GNE Grantback License").
- 7.1.2. **Activities**. The Collaboration Off-the-Shelf Exclusive License and Collaboration Personalised Exclusive License include, but are not limited to, the following activities (for clarity, only during or after the Research Term):
 - (a) [***]
 - (b) [***].
 - (c) [***].

7.1.3. **GNE Covenants and Rights**.

- (a) Solely during the Research Term, GNE shall not [***]. For clarity, the foregoing covenant shall be in force and effect during the Research Term and shall end upon the expiration or earlier termination of the Research Term.
- (b) Notwithstanding the foregoing, upon (i) mutual written agreement by the Parties, the restrictions described in <u>Section 7.1.3(a)</u> (GNE Covenants and Rights) shall not apply with respect to the mutually agreed upon activities, or (ii) [***]. If GNE exercises its rights under this <u>Section 7.1.3(b)</u> (GNE Covenants and Rights), then GNE may request that, [***].
- (c) Notwithstanding anything to the contrary in this Agreement, including this <u>Section 7.1.3</u> (GNE Covenants and Rights), nothing shall prohibit, prevent, or limit GNE from using in the manufacturing, making or having made of any other product (i) any part(s) of the Manufacturing process for any Collaboration Off-the-Shelf T-Cell Therapy or Collaboration Personalised T-Cell Therapy that it owns or Controls and (ii) any improvements it subsequently makes to such Manufacturing processes.

7.1.4. Sublicenses and Subcontractors.

- (a) **Sublicenses**. GNE shall have the right to sublicense the rights granted under <u>Sections 7.1.1(a)</u> (License Grants), 7.1.1(b) (License Grants), and 7.1.1(c) (License Grants) (" **Sublicense**") to its Affiliates or Third Parties (each a " **Sublicense**"), *provided* that such sublicense is consistent with the terms and conditions of this Agreement, and *provided* further that GNE shall remain responsible for such Affiliate's or Third Party's compliance with all obligations under this Agreement applicable to such Affiliate or Third Party. [***]. For clarity, no grant of any sublicense to a Third Party or an Affiliate shall relieve GNE of its obligations hereunder. [***] "Sublicensee" excludes any "Compulsory Sublicensee."
- (b) **Subcontractors**. Outside of the Research Program, GNE shall have the right to enter into subcontracts with Third Parties and Affiliates acting by or for the benefit of GNE with respect to the activities authorized under this Agreement; *provided*, that in each instance such subcontract is consistent with the terms and conditions of this Agreement. GNE shall remain responsible (at its cost) for and shall ensure that each subcontractor complies with the terms and conditions of this Agreement, to the extent applicable to such subcontractor.
- (c) **Trade Secrets**. To the extent that any disclosure of Adaptimmune's Confidential Information to any Third Party Sublicensee or subcontractor includes any of Adaptimmune's trade secret information specifically identified as a trade secret by Adaptimmune in writing in advance to GNE (*e.g.*, Adaptimmune's Confidential Information related to the Adaptimmune Platform specifically identified as a trade secret by Adaptimmune in writing in advance to GNE), GNE shall notify Adaptimmune of such requirement to disclose prior to such disclosure and Adaptimmune shall be entitled to request additional protection measures from any such Third Party Sublicensee or subcontractor, including entry into a direct agreement of confidentiality with such Sublicensee or subcontractor.

7.2. License from GNE to Adaptimmune.

- 7.2.1. **License Grants** . Subject to the terms and conditions of this Agreement (including Section 7.2.2 (Adaptimmune Covenants and Rights)), GNE hereby grants to Adaptimmune:
- (a) a royalty-free, non-transferable (subject to Section 20.4 (Assignment)), worldwide, non-sublicensable, non-exclusive license under GNE Licensed IP to the extent necessary for Adaptimmune to conduct the Research Program.
- (b) a perpetual, irrevocable, non-exclusive, royalty-free, fully-paid up, freely transferrable and freely sublicensable license under all of GNE's interest in, to and under GNE Platform Improvement IP [***], in each case for any use ("GNE-to-Adaptimmune Grantback License").
- (c) a perpetual, irrevocable, non-exclusive, royalty-free, fully-paid up, sublicensable license under GNE Process Know-How to [***]. Adaptimmune shall have the right to sublicense the rights granted in this Section to its Affiliates and Third Parties; *provided*, that such sublicense is consistent with the terms and conditions of this Agreement; *provided*, further, that such Affiliate and Third Party shall comply with the obligations that apply to Adaptimmune under Article 9 (Exclusivity) (to the extent applicable to the cell therapies being researched, developed or commercialized by such Affiliate or Third Party); and *provided*, further, that Adaptimmune shall remain responsible for such Affiliate's or Third Party's compliance with all obligations under this Agreement applicable to such Affiliate or Third Party, including those under Article 9 (Exclusivity).
- 7.2.2. **Adaptimmune Covenants and Rights**. Adaptimmune shall have the right to use GNE-Provided $\alpha\beta$ Receptors and any other materials provided by GNE under this Agreement solely for the purposes of performing the activities under the Research Program and not for any other purpose, [***].
- 7.2.3. **Excluded GNE Affiliate Assets**. Notwithstanding anything to the contrary in this $\underline{\text{Section}}$ $\underline{7.1}$ (License from GNE to Adaptimmune) or elsewhere in this Agreement, no licenses or rights are granted to Adaptimmune under any information, data, proprietary materials and/or other intellectual property rights, whether or not patentable, that are Controlled by any of the following entities, unless and until such entity is deemed an Affiliate of GNE in accordance with $\underline{\text{Section 1.18}}$ (" Affiliate"): [***], and each of their respective subsidiaries.

7.3. Third Party IP Licenses.

7.3.1. **Notice**. If at any time during the Term, either Party reasonably determines that a license under any Third Party Intellectual Property may be necessary or useful for the Development, Manufacture or Commercialization of any Licensed Product ("**Third Party IP**"), and a Third Party Infringement Claim has not occurred with respect to such Third Party Intellectual Property, then such Party will promptly notify the other Party in writing via the JIPC ("**Third Party IP Notice**"). The JIPC shall then discuss and determine whether any changes need to be made to the Research Plan (if the Research Term has not then expired or earlier terminated) to avoid any potential infringement of such Third Party IP or whether one or both of the Parties should negotiate to obtain a license under or otherwise acquire rights to such Third Party IP. For

clarity, if a Third Party Infringement Claim has occurred with respect to such Third Party IP, then <u>Section 12.5</u> (Third Party Infringement Claims) shall apply. [***].

7.3.2. **Right to Obtain**.

- (a) IP Relevant to Adaptimmune Platform. To the extent such Third Party IP specifically relates to the use of the Adaptimmune Platform, [***] will have the first right to obtain a license under such Third Party IP [***] (" [***] Third Party IP License"). [***] shall provide [***] with the opportunity to review drafts of and provide comments with respect to such [***] Third Party IP License when [***] becomes aware that [***] would incur payment obligations to such Third Party as a direct result of [***]'s exercise of such Third Party IP under the licenses granted to [***] under this Agreement ("[***]") and shall consider such comments in good faith. [***].
- (b) Other Third Party Licenses. Subject to Section 7.3.2(a) (IP Relevant to Adaptimmune Platform), [***] shall have the right to obtain licenses to all Third Party IP and shall have [***]. To the extent such Third Party IP is necessary for [***] performance of its activities under the Research Program, [***] will use commercially reasonable efforts to obtain a right under such license for [***] to sublicense such rights to [***] for use in [***] performance of the Research Program. For clarity, [***] shall also have the right to itself obtain licenses to such Third Party IP.
- 7.4. **No Additional Licenses**. Except as expressly provided in this Agreement, nothing in this Agreement shall grant either Party any right, title or interest in and to the Know-How, Patents or other Intellectual Property rights of the other Party (either expressly or by implication or estoppel).
- 7.5. **GNE and Roche.** As used in this <u>Article 7</u> (Licenses), the terms "GNE" and "Parties" shall include Roche.
- 7.6. **Pre-Existing License Terms**. Use of iPS Cell Lines provided by Adaptimmune under the Research Plan are subject to the terms and conditions from the Existing Upstream License Agreements that are set out or specifically referenced in **Schedule 7.6** (Pre-Existing License Terms). GNE agrees to comply with such terms and conditions to the extent applicable to GNE. [***].
- 7.7. [***].

ARTICLE 8 MANUFACTURING, SUPPLY AND TECHNOLOGY TRANSFER

8.1. **Phase I Supply by Adaptimmune.**

8.1.1. **Phase I Materials**. Adaptimmune shall Manufacture and supply all Allogeneic T-Cells and Collaboration Off-the-Shelf T-Cell Therapies, in each case developed solely or jointly by Adaptimmune during the Research Program and required for the performance of the Research Program (" **Research Program Materials**") and the first Phase I Clinical Trial of each Collaboration Off-the-Shelf T-Cell Therapy (" **Phase I Materials**"). Adaptimmune shall conduct

all its respective Manufacturing and supply activities hereunder in compliance with all laws, rules and regulations, including GMP, applicable to its performance of such Manufacturing and supply activities. Adaptimmune shall, in performing its Manufacturing and supply obligations, assign responsibilities to those portions of its organization that have the appropriate resources, expertise and responsibility for such obligations.

- 8.1.2. **Phase I Supply Agreement**. For each Collaboration Off-the-Shelf T-Cell Therapy, the Parties shall mutually agree upon and enter a supply agreement and a quality agreement for such Manufacture and supply of Phase I Materials by Adaptimmune ("**Phase I Material Supply Agreement**"), such Phase I Material Supply Agreement to be agreed upon and entered into by [***] applicable to the commencement of Clinical Trials for such Collaboration Off-the-Shelf T-Cell Therapy. [***].
- 8.1.3. **Quality Agreement**. The Parties shall enter into a quality agreement for clinical Manufacture and supply of Phase I Materials by Adaptimmune by no later than the date of execution of the Phase I Material Supply Agreement. Such quality agreement shall contain detailed provisions setting forth the technical and quality requirements for the Manufacture and supply of Phase I Materials in accordance with the specifications therefor and laws, rules and regulations, including compliance requirements, and disposition of Phase I Material, in each case as applicable to Adaptimmune's Manufacture and supply of Phase I Materials ("**Quality Agreement**"). GNE shall have the right, at its own cost, to conduct an initial audit prior to execution of the Quality Agreement and subsequent audits, of any facility at which the Manufacture of the Phase I Materials to be provided under the Phase I Material Supply Agreement will be performed. Any initial audit will be arranged at a time reasonably convenient to Adaptimmune and within normal business hours of Adaptimmune . GNE shall provide a minimum of [***] days' notice ahead of such initial audit . The subsequent audits will be performed in accordance with the terms of the Quality Agreement. Each audit shall not exceed [***] days and shall be limited to the facility at which Manufacture of Phase I Material is performed.

8.2. **GNE Supply**.

- 8.2.1. Off-the Shelf Phase I Supply by GNE. Notwithstanding Section 8.1 (Phase I Supply by Adaptimmune), if GNE provides written notice to Adaptimmune prior to entry into any Phase I Material Supply Agreement for the Manufacture and supply of a given Collaboration Off-the-Shelf T-Cell Therapy, and in any event no later than [***] prior to the anticipated filing of the first IND applicable to the commencement of Clinical Trials for any Collaboration Off-the-Shelf T-Cell Therapy, that GNE has determined that GNE shall instead Manufacture and supply all Phase I Materials for such Collaboration Off-the-Shelf T-Cell Therapy, then GNE shall have the right to undertake such Manufacturing and Adaptimmune shall, after the delivery of such written notice, no longer have such obligation to Manufacture and supply such Collaboration Off-the-Shelf T-Cell Therapy.
- 8.2.2. **Clinical and Commercial Supply**. Subject to Adaptimmune's obligation to Manufacture and supply described in <u>Section 8.1</u> (Phase I Supply by Adaptimmune), GNE shall be solely responsible for clinical and commercial supply, and Manufacturing, of all Licensed Products in the Territory.

- 8.3. **Provision of Cell Lines**. If more vials of any Working Cell Bank are required for activities under this Agreement than are provided to GNE pursuant to this <u>Section 8.3</u> (Provision of Cell Lines), the Parties shall discuss in good faith the provision of additional supply of such materials by Adaptimmune.
 - 8.3.1. Collaboration OTS Cell Banks. [***].
 - 8.3.2. Collaboration Personalised Cell Banks. [***].
 - 8.3.3. Collaboration iPS Cell Lines. [***].
- 8.4. Technology Transfer. For each (a) Collaboration Off-the-Shelf T-Cell Therapy developed under the Research Program for which GNE provides notice that GNE elects to Manufacture such Collaboration Off-the-Shelf T-Cell Therapy pursuant to Section 8.2.1 (Off-the-Shelf Phase I Supply by GNE), Adaptimmune shall, at GNE's request, transfer to GNE or any CMO (acting on behalf of GNE) reasonably acceptable to Adaptimmune the differentiation process used to create the applicable Collaboration Off-the-Shelf T-Cell Therapy from the applicable Collaboration OTS MCB and each analytical method and assay (e.g., IPC, release, characterizations, and stability studies) used in or reasonably required for the Manufacture and release of such Collaboration Off-the-Shelf T-Cell Therapy, (b) Collaboration Off-the-Shelf T-Cell Therapy (other than such Collaboration Off-the-Shelf T-Cell Therapy described in clause (a) above) developed under the Research Program, Adaptimmune shall, prior to commencement of the first Phase II clinical trial for such Collaboration Off-the-Shelf T-Cell Therapy (or such other time as mutually agreed by the Parties), transfer to GNE or any CMO (acting on behalf of GNE) reasonably acceptable to Adaptimmune the differentiation process used to create the applicable Collaboration Off-the-Shelf T-Cell Therapy from the applicable Collaboration OTS MCB and each analytical method and assay (e.g., IPC, release, characterizations, and stability studies) used in or reasonably required for the Manufacture and release of such Collaboration Off-the-Shelf T-Cell Therapy, and (c) Collaboration Personalised T-Cell Therapy under the Research Program, Adaptimmune shall, prior to commencement of the first Phase I clinical trial for the applicable Collaboration Personalised T-Cell Therapy (or by such other time as mutually agreed by the Parties), transfer to GNE or any CMO (acting on behalf of GNE) reasonably acceptable to Adaptimmune the process for Manufacture of the αβ Allogeneic T-Cells used in the Manufacture and release of such Collaboration Personalised T-Cell Therapy, in each case ((a), (b) and (c)) at [***]. Such technology transfer and its respective timing would be further detailed in a technology transfer plan mutually agreed between the Parties ("Tech Transfer Plan"). Such Tech Transfer Plan shall include acceptance criteria and an end date and shall include, at a minimum, such information, assistance and materials reasonably necessary to enable GNE to Manufacture (i) the applicable αβ Allogeneic T-Cells required for Manufacture of any Collaboration Personalised T-Cell Therapy, or (ii) the applicable Collaboration Off-the-Shelf T-Cell Therapy.

ARTICLE 9 EXCLUSIVITY

9.1. **Adaptimmune Exclusivity Requirements**. Adaptimmune hereby agrees to the following restrictions (collectively the "Exclusivity Requirements"):

- Off-the-Shelf T-Cell Therapy. During the Off-the-Shelf Exclusivity Term, to the extent 9.1.1. permitted by applicable laws, and on a Collaboration Target-by-Collaboration Target basis, Adaptimmune shall not itself, or through or with any of its Affiliates, directly or indirectly, Research, Develop or Commercialize, or authorize, enable, or license, directly or indirectly, any Third Party to Research, Develop or Commercialize any Cell Therapy that contains at least one αβ Receptor that is Directed To any Collaboration Target. " Off-the-Shelf Exclusivity Term" means, on a Collaboration Target-by-Collaboration Target basis, the period of time beginning on the Effective Date and ending: (a) [***]. Notwithstanding the foregoing, on a Collaboration Target-by-Collaboration Target basis, if [***] then (A) the Exclusivity Requirements in this Section 9.1.1 (Off-the-Shelf T-Cell Therapy) would cease to apply in relation to such Collaboration Target, and (B) GNE's obligations under Section 4.3 (Progress Reports) would cease to apply in relation to such Collaboration Target. For clarity, inclusion of $\alpha\beta$ Receptor(s) in a Cell Therapy, where such αβ Receptor(s) have been engineered. Researched or Developed to be Directed To a Target that is not a Collaboration Target (as long as such αβ Receptor(s) were not originally isolated from the patient to whom the therapy is administered), shall not be prohibited by the exclusivity commitment in this paragraph.
- 9.1.2. **Personalised T-Cell Therapy**. During the Personalised Exclusivity Term, to the extent permitted under applicable laws, Adaptimmune shall not itself, or through or with any of its Affiliates, directly or indirectly, Research, Develop or Commercialize, or authorize, enable, or license any Third Party, directly or indirectly, to Research, Develop or Commercialize, any Cell Therapy Developed or Manufactured from an iPS Cell and that contains at least one $\alpha\beta$ Receptor isolated from the patient to whom such therapy is administered, or any $\alpha\beta$ Receptor that is modified after such isolation from, and prior to such administration to, such patient. "**Personalised Exclusivity Term**" means the period of time beginning on the Effective Date and ending [***]. Notwithstanding the foregoing, if [***] then (A) the Exclusivity Requirements in this Section 9.1.2 (Personalised T-Cell Therapy) would cease to apply in relation to Collaboration Personalised T-Cell Therapies, and (B) GNE's obligations under Section 4.3 (Progress Reports) would cease to apply in relation to all Collaboration Personalised T-Cell Therapies.

For clarity, the restrictions set out in <u>Section 9.1.1</u> (Off-the-Shelf T-Cell Therapy) or <u>Section 9.1.2</u> (Personalised T-Cell Therapy) shall not prevent Adaptimmune from Researching, Developing, Manufacturing or Commercializing (a) autologous Cell Therapies save where such Cell Therapies are Directed To a Collaboration Target; (b) Cell Therapies Manufactured or Developed from an iPS Cell and comprising Receptors engineered to be Directed To Targets other than Collaboration Targets; or (c) [***].

9.2. **Change of Control.**

9.2.1. **Segregation**. Notwithstanding anything in <u>Section 9.1</u> (Adaptimmune Exclusivity Requirements) to the contrary, if (a) Adaptimmune undergoes a Change of Control, and (b) on the date of the closing of such Change of Control or at any time thereafter, the Acquiring Entity or Acquiring Entity Family are Researching, Developing, Manufacturing or Commercializing a Cell Therapy that would be subject to the restrictions described in <u>Section 9.1</u> (Adaptimmune Exclusivity Requirements) if conducted by Adaptimmune (including any such Cell Therapy that [***]) (such Cell Therapy a "**Competing Product**"), then Adaptimmune will not be

in breach of <u>Section 9.1</u> (Adaptimmune Exclusivity Requirements) as a result of such Change of Control or the continuation or start of such activities by such Acquiring Entity or Acquiring Entity Family Segregates such Competing Product.

9.2.2. Segregation of Adaptimmune Collaboration Activities. Adaptimmune shall notify GNE in writing as soon as possible after Adaptimmune announces publicly any information regarding any proposed Change of Control of Adaptimmune (or if the Change of Control will not be publicly announced, then no later than [***] after the closing of the Change of Control transaction). GNE will have the right to request that Adaptimmune Segregate its performance of the Research Program from other activities of the Acquiring Entity or Acquiring Entity Family by delivery of written notice ("Change of Control Notice") to Adaptimmune within [***] of the effective date of such Change of Control. Upon delivery of the Change of Control Notice, Adaptimmune will use reasonable efforts to ensure that its activities under the Research Program are Segregated from those of the Acquiring Entity or Acquiring Entity Family and will confirm that such Segregation has occurred in writing to GNE.

ARTICLE 10 FINANCIAL TERMS

- 10.1. **Upfront Payment**. In consideration of the rights granted by Adaptimmune under <u>Article 7</u> (Licenses) of this Agreement, GNE shall pay to Adaptimmune a non-refundable, non-creditable upfront payment in the amount of One Hundred Fifty Million US Dollars (\$150,000,000). Such payment shall be made within [***] days after receipt of invoice from Adaptimmune, which invoice shall be given after the Effective Date.
- 10.2. **Research Term Extension Fee**. If GNE exercises its right to extend the Initial Research Term by an Additional Research Term, or to extend the first Additional Research Term by an Additional Research Term (as further described in Section 3.3.2 (Research Term Extension)), GNE shall pay Adaptimmune a non-refundable, non-creditable payment of [***] ("**Research Term Extension Fee**") for each such Additional Research Term, such payment to be made by GNE to Adaptimmune within [***] days after receipt of invoice from Adaptimmune for such Research Term Extension Fee.

10.3. [***] and Research Milestone Payments.

10.3.1. [***] **Payments**. GNE shall pay Adaptimmune the following non-refundable, non-creditable payments in consideration for the [***] following achievement of the corresponding event within [***] days after receipt of invoice from Adaptimmune for such payment following such event:

Event	Payment Amount (in US Dollars)
[***]	[***]

Event	Payment Amount (in US Dollars)
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

10.3.2. **Research Milestone Payments**. GNE shall pay Adaptimmune the following non-refundable, non-creditable Research milestone payments following achievement of the corresponding event within [***] days after receipt of invoice from Adaptimmune for such payment following such event:

Event	Payment Amount (in US Dollars)
[***]	[***]
[***]	[***]

For purposes of this <u>Section 10.3.2</u> (Research Milestone Payments):

- (a) [***].
- (b) [***].

10.3.3. Additional Collaboration Targets. As described in Section 3.4.1 (Collaboration Targets), GNE shall have the right to designate up to [***] Additional Collaboration Targets in addition to the [***] Initial Collaboration Targets. For each of these Additional Collaboration Targets, GNE shall pay Adaptimmune a non-refundable, non-creditable, payment of [***] (each an "Additional Collaboration Target Designation Fee") in consideration for the grant of the rights for such Additional Collaboration Targets, such payment to be made by GNE to Adaptimmune within [***] days after receipt of invoice (which shall not be given to GNE prior to when the nominated Target is deemed an Additional Collaboration Target in accordance with Section 3.6 (Nomination)) from Adaptimmune for such Additional Collaboration Target Designation Fee. For clarity, no Additional Collaboration Target Designation Fee shall be due if any Collaboration Target is substituted in accordance with Section 3.4.2 (Collaboration Target Substitution).

10.4. **Development Event Payments**.

- 10.4.1. Collaboration Off-the-Shelf T-Cell Therapy Development Milestones. Subject to Section 10.8.7 (Sunset of Certain Payment Terms), on a Collaboration Target-by-Collaboration Target basis, GNE shall pay Adaptimmune the Development milestone payments set forth in Schedule 10.4.1(a) and Schedule 10.4.1(b) (each, an "Off-the-Shelf Development Milestone Payment") following achievement of the corresponding milestone event specified in Schedule 10.4.1(a) and Schedule 10.4.1(b) (each, an "Off-the-Shelf Development Milestone Event").
- (a) **No Adaptimmune Opt-In**. Subject to <u>Section 10.8.7</u> (Sunset of Certain Payment Terms), on a Collaboration Target-by-Collaboration Target basis, with respect to each such Collaboration Target for which Adaptimmune has not exercised its Opt-In, GNE shall pay Adaptimmune the Off-the-Shelf Development Milestone Payments set forth in **Schedule 10.4.1(a)** following achievement of the corresponding Off-the-Shelf Development Milestone Event specified in such Schedule by the applicable Collaboration Off-the-Shelf T-Cell Therapy that is Directed To such Collaboration Target.
- (b) Adaptimmune Opt-In. Subject to Section 10.8.7 (Sunset of Certain Payment Terms), on a Collaboration Target-by-Collaboration Target basis, with respect to each such Collaboration Target for which Adaptimmune has exercised its Opt-In, GNE shall pay Adaptimmune the Off-the-Shelf Development Milestone Payments set forth in Schedule 10.4.1(b) following achievement of the corresponding Off-the-Shelf Development Milestone Event specified in such Schedule by the applicable Collaboration Off-the-Shelf T-Cell Therapy that is Directed To such Collaboration Target.
- (c) Number of Payments Per Collaboration Target. Each Off-the-Shelf Development Milestone Payment in Schedule 10.4.1(a) and Schedule 10.4.1(b), whichever is applicable, shall be paid only once per Collaboration Target regardless of how many times the corresponding Off-the-Shelf Development Milestone Event is achieved by Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target.
- 10.4.2. Collaboration Personalised T-Cell Therapy Development Milestones. Subject to Section 10.8.7 (Sunset of Certain Payment Terms), GNE shall pay Adaptimmune the Development milestone payments set forth in Schedule 10.4.2 (each, a "Personalised Development Milestone Payment") following achievement of the corresponding milestone event specified in such Schedule (each, a "Personalised Development Milestone Event") by the applicable Collaboration Personalised T-Cell Therapy. Each milestone payment would be paid only once regardless of how many times the corresponding milestone event is achieved by Collaboration Personalised T-Cell Therapies.

10.5. Net Sales Event Payments.

10.5.1. **Collaboration Off-the-Shelf T-Cell Therapy Net Sales Milestones**. Subject to <u>Section 10.8.7</u> (Sunset of Certain Payment Terms), on a Collaboration Target-by-Collaboration Target basis, GNE shall pay Adaptimmune the payments described in the tables in this <u>Section 10.5.1</u> (Collaboration Off-the-Shelf T-Cell Therapy Net Sales Milestones) (each, an "**Off-the-**

Shelf Net Sales Milestone Payment") following achievement of the corresponding milestone event for Annual Net Sales of Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target that occur in the applicable portion of the Territory (each, an " **Off-the-Shelf Net Sales Milestone Event**").

(a) **No Adaptimmune Opt-In**. Subject to <u>Section 10.8.7</u> (Sunset of Certain Payment Terms), on a Collaboration Target-by-Collaboration Target basis, with respect to each such Collaboration Target for which Adaptimmune has not exercised its Opt-In, GNE shall pay Adaptimmune the applicable Off-the-Shelf Net Sales Milestone Payment specified in the table below following achievement of the corresponding Off-the-Shelf Net Sales Milestone Event specified in the table below for worldwide Annual Net Sales of all Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target. Each Off-the-Shelf Net Sales Milestone Payment shall be paid only once per Collaboration Target regardless of how many, or how many times, Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target generate, individually or collectively, Annual Net Sales at or beyond the thresholds below, and in no event will GNE pay more than [***] in Off-the-Shelf Net Sales Milestone Payments for each Collaboration Target that is not the subject of an Opt-In.

Off-the-Shelf Net Sales Milestone Event (No Opt-In Exercised)	Off-the-Shelf Net Sales Milestone Payment (in US Dollars)
When worldwide Annual Net Sales for all Collaboration Off-the-Shelf T-Cell Products Directed To the applicable Collaboration Target [***]	\$[***]
When worldwide Annual Net Sales for all Collaboration Off-the-Shelf T-Cell Products Directed To the applicable Collaboration Target [***]	\$[***]
When worldwide Annual Net Sales for all Collaboration Off-the-Shelf T-Cell Products Directed To the applicable Collaboration Target [***]	\$[***]
Total amount payable per Collaboration Target	\$ [***]

(b) Adaptimmune Opt-In. Subject to Section 10.8.7 (Sunset of Certain Payment Terms), on a Collaboration Target-by-Collaboration Target basis, with respect to each such Collaboration Target for which Adaptimmune has timely exercised its Opt-In, GNE shall pay Adaptimmune the applicable Off-the-Shelf Net Sales Milestone Payment specified in the table below following achievement of the corresponding Off-the-Shelf Net Sales Milestone Event specified in the table below for Annual Net Sales of all Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target that occur outside the US. Each Off-the-Shelf Net Sales Milestone Payment shall be paid only once per Collaboration Target regardless of how many, or how many times, Collaboration Off-the-Shelf T-Cell Therapies that are Directed To

such Collaboration Target generate, individually or collectively, Annual Net Sales at or beyond the thresholds below, and in no event will GNE pay more than [***] in Off-the-Shelf Net Sales Milestone Payments for each Collaboration Target that is the subject of an Opt-In.

Off-the-Shelf Net Sales Milestone Event (Opt-In Exercised)	Off-the-Shelf Net Sales Milestone Payment (in US Dollars)
When Annual Net Sales outside the US for all Collaboration Off-the- Shelf T-Cell Products Directed To the applicable Collaboration Target [***]	\$[***]
When Annual Net Sales outside the US for all Collaboration Off-the- Shelf T-Cell Products Directed To the applicable Collaboration Target [***]	\$[***]
When Annual Net Sales outside the US for all Collaboration Off-the- Shelf T-Cell Products Directed To the applicable Collaboration Target [***]	\$[***]
Total amount payable per Collaboration Target	\$[***]

10.5.2. Collaboration Personalised T-Cell Therapy Net Sales Milestones. Subject to Section 10.8.7 (Sunset of Certain Payment Terms), GNE shall pay Adaptimmune the payments described in the table in this Section 10.5.2 (Collaboration Personalised T-Cell Therapy Net Sales Milestones) (each, a "Personalised Net Sales Milestone Payment") following achievement of the corresponding milestone event for Annual Net Sales of Collaboration Personalised T-Cell Therapies described in the table in this Section (each, a "Personalised Net Sales Milestone Event"). Each Personalised Net Sales Milestone Event may be achieved by the world-wide Annual Net Sales of all Collaboration Personalised T-Cell Therapies. Each Personalised Net Sales Milestone Payment shall be paid only once regardless of how many, or how many times, Collaboration Personalised T-Cell Therapies generate, individually or collectively, Annual Net Sales at or beyond the thresholds below, and in no event will GNE pay [***] in Personalised Net Sales Milestone Payments.

Personalised Net Sales Milestone Event	Personalised Net Sales Milestone Payment (in US Dollars)
When worldwide Annual Net Sales for all Collaboration Personalised T-Cell Therapies [***]	\$[***]

Personalised Net Sales Milestone Event	Personalised Net Sales Milestone Payment (in US Dollars)
When worldwide Annual Net Sales for all Collaboration Personalised T-Cell Therapies [***]	\$[***]
When worldwide Annual Net Sales for all Collaboration Personalised T-Cell Therapies [***]	\$[***]
When worldwide Annual Net Sales for all Collaboration Personalised T-Cell Therapies [***]	\$[***]
Total amount payable for achieving all Personalised Net Sales Milestone Events	\$ [***]

10.6. **Notice of Achievement; Payment**. With respect to each research milestone listed in Section 10.3 (Exclusivity and Research Milestone Payments), and each event listed in Section 10.4 (Development Event Payments) and Section 10.5 (Net Sales Event Payments), GNE shall notify Adaptimmune following the achievement of such event within [***] days after the quarter for which such event occurs. On or after Adaptimmune's receipt of such notice of achievement, Adaptimmune shall submit a written invoice to GNE for the corresponding milestone payment. Each such invoice shall specify the applicable milestone event, and, unless otherwise requested by GNE in writing, Adaptimmune shall email such invoices to GNE's Alliance Manager. GNE shall pay Adaptimmune the respective accrued and payable milestone payment within [***] days of receipt of an invoice from Adaptimmune with respect thereto.

10.7. **Royalties**.

10.7.1. Royalty Payments for Collaboration Off-the-Shelf T-Cell Therapy.

(a) **No Adaptimmune Opt-In**. On a Collaboration Target-by-Collaboration Target basis, with respect to each such Collaboration Target for which Adaptimmune has not exercised its Opt-In, GNE shall pay Adaptimmune the following tiered royalties on aggregate worldwide Annual Net Sales of Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target that occur during the applicable Royalty Term in the Territory.

Aggregate Worldwide Annual Net Sales of all Collaboration Off- the-Shelf T-Cell Therapies for a Given Collaboration Target (in US Dollars; no Adaptimmune Opt-In)	Royalty Rate Applicable
[***]	[***]
[***]	[***]

[***]	[***]
-------	-------

(b) Adaptimmune Opt-In. On a Collaboration Target-by-Collaboration Target basis, with respect to each such Collaboration Target for which Adaptimmune has exercised its Opt-In, GNE shall pay Adaptimmune the following tiered royalties on aggregate Annual Net Sales of Collaboration Off-the-Shelf T-Cell Therapies that are Directed To such Collaboration Target that occur during the applicable Royalty Term outside of the US. For clarity, if Adaptimmune has exercised its Opt-In with respect to a Collaboration Target, then GNE shall have no obligation to pay any royalties to Adaptimmune under this Section 10.7.1(b) (Adaptimmune Opt-In) on Net Sales of any Collaboration Off-the-Shelf T-Cell Therapies Directed To such Collaboration Target that occur within the US and instead will share US Net Profits and Net Losses in accordance with Section 10.9 (Economic Effect of Opt-In; Profit-Share).

Ex-US Annual Net Sales of all Collaboration Off-the-Shelf T-Cell Therapies for a Given Collaboration Target (in US Dollars; Adaptimmune Opt-In)	Royalty Rate Applicable
[***]	[***]
[***]	[***]
[***]	[***]

10.7.2. **Royalty Payments for Collaboration Personalised T-Cell Therapy**. GNE shall pay Adaptimmune the following tiered royalties on aggregate worldwide Annual Net Sales of all Collaboration Personalised T-Cell Therapies that occur during the applicable Royalty Term in the Territory.

Aggregate Worldwide Annual Net Sales of Collaboration Personalised T-Cell Therapies (in US Dollars)	Royalty Rate Applicable
[***]	[***]
[***]	[***]
[***]	[***]

10.8. **Royalty Reductions**.

10.8.1. **Payments For Third Party IP**. GNE may offset against any royalty payment due and payable by GNE to Adaptimmune under Section 10.7 (Royalties) with respect to such Licensed Product by [***] (or an Affiliate or Sublicensee) pursuant to a license under Third Party IP obtained after the Effective Date in accordance with Section 7.3 (Third Party IP Licenses) (each a " **Third Party IP License**"); provided, [***].

- 10.8.2. **Biosimilar**. On a country-by-country and Licensed Product-by-Licensed Product basis, following the First Commercial Sale of a Biosimilar to a Licensed Product in a country, if Net Sales of the applicable Licensed Product in such country [***] the royalties due and payable by GNE for such Licensed Product under Section 10.7 (Royalties) [***] in such country. As used herein, "**Biosimilar**" means with respect to a Licensed Product, any drug, biological product or Cell Therapy that (a) [***]; provided, in each instance under clauses (a)- (c), which Biosimilar is sold by a Third Party that is not a licensee or Sublicensee of GNE (or any of its Affiliates) and that has not otherwise been authorized, directly or indirectly, by GNE (or any of its Affiliates) to market and sell such product.
- 10.8.3. **Single Royalty**. No more than one royalty payment shall be due under <u>Section 10.7</u> (Royalties) with respect to a sale of a particular Licensed Product. For the avoidance of doubt, multiple royalties shall not be payable because the sale of a particular Licensed Product is Covered by more than one (1) Valid Claim of the Adaptimmune Royalty Patent Rights in the country in which such Licensed Product is sold.
- 10.8.4. **No Valid Claims**. Subject to <u>Section 10.8.5</u> (Royalty Floor), on a Licensed Product-by-Licensed Product and country-by-country basis, if such Licensed Product is no longer Covered by a Valid Claim of (a) [***] the applicable royalty rates for Net Sales with respect to such Licensed Product pursuant to <u>Section 10.7</u> (Royalties) in such country will be [***].
- 10.8.5. **Royalty Floor** . On a calendar quarter-by-calendar quarter, country-by-country, and Licensed Product-by-Licensed Product basis, the [***].
- 10.8.6. **Rights Following Expiration of Royalty Term**. On a country-by-country and Licensed Product-by-Licensed Product basis, upon expiry of its payment obligation hereunder with respect to a Licensed Product in a country, all of the licenses granted to GNE under this Agreement, including in Section 7.1.1 (License Grants), shall be perpetual, non-exclusive, irrevocable, fully paid-up, and royalty-free in respect of that Licensed Product in that country.

10.8.7. Sunset of Certain Payment Terms.

- (a) Collaboration Off-the-Shelf T-Cell Therapy. Notwithstanding anything to the contrary in this Agreement, if GNE creates a Cell Line that is Functionally Distinct from all then-existing Collaboration OTS Cell Lines (each such Cell Line "GNE-Modified OTS Cell Line") and incorporates an $\alpha\beta$ Allogeneic T-Cell that is engineered or differentiated from such GNE-Modified OTS Cell Line into a Collaboration Off-the-Shelf T-Cell Therapy, then the following shall apply:
- (i) If such GNE-Modified OTS Cell Line has a Cell Line Creation Date that occurs (A) [***], then the Off-the-Shelf Development Milestone Payments for any Off-the-Shelf Development Milestone Event achieved by any Collaboration Off-the-Shelf T-Cell Therapy that contains such GNE-Modified OTS Cell Line, to the extent payable per the terms of Section 10.4.1(a) (No Adaptimmune Opt-In) or Section 10.4.1(b) (Adaptimmune Opt-In), shall be [***]; and (B) during the [***], then the Off-the-Shelf Development Milestone Payments for any Off-the-Shelf Development Milestone Event achieved by any Collaboration Off-the-Shelf T-Cell

Therapy that contains such GNE-Modified OTS Cell Line, to the extent payable per the terms of <u>Section 10.4.1(a)</u> (No Adaptimmune Opt-In) or <u>Section 10.4.1(b)</u> (Adaptimmune Opt-In), [***].

- (ii) If such GNE-Modified OTS Cell Line has a Cell Line Creation Date that [***], then GNE shall [***] for any Off-the-Shelf Development Milestone Event subsequently achieved by any Collaboration Off-the-Shelf T-Cell Therapy that contains such GNE-Modified OTS Cell Line
- (iii) For clarity, (A) this <u>Section 10.8.7(a)</u> (Collaboration Off-the-Shelf T-Cell Therapy) does not apply to any Collaboration Off-the-Shelf T-Cell Therapy containing a Cell Line created by the Parties during the Research Term, including all GNE-Modified OTS Cell Lines created by GNE during the Research Term, and (B) the [***] described in this <u>Section 10.8.7(a)</u> (Collaboration Off-the-Shelf T-Cell Therapy) shall not apply to any other payments described in <u>Article 10</u> (Financial Terms) of this Agreement, including royalties.
- (b) **Collaboration Personalised T-Cell Therapy**. Notwithstanding anything to the contrary in this Agreement, if GNE creates a Cell Line that is Functionally Distinct from all then-existing Collaboration Personalised Cell Lines (each such Cell Line a "**GNE-Modified Personalised Cell Line**") and incorporates an $\alpha\beta$ Allogeneic T-Cell that is engineered or differentiated from such GNE-Modified Personalised Cell Line into a Collaboration Personalised T-Cell Therapy, the following shall apply:
- (i) If such GNE-Modified Personalised Cell Line has a Cell Line Creation Date that occurs (A) [***], then the Personalised Development Milestone Payments for any Personalised Development Milestone Event achieved by any Collaboration Personalised T-Cell Therapy that contains such GNE-Modified Personalised Cell Line, to the extent payable per the terms of Section 10.4.2 (Collaboration Personalised T-Cell Therapy Development Milestones), [***]; and (B) during the [***], then the Personalised Development Milestone Payments for any Personalised Development Milestone Event achieved by any Collaboration Personalised T-Cell Therapy that contains such GNE-Modified Personalised Cell Line, to the extent payable per the terms of Section 10.4.2 (Collaboration Personalised T-Cell Therapy Development Milestones), [***].
- (ii) If such GNE-Modified Personalised Cell Line has a Cell Line Creation Date that occurs [***], then GNE shall [***] Personalised Development Milestone Payments for any Personalised Development Milestone Event achieved by any Collaboration Personalised T-Cell Therapy that contains such GNE-Modified Personalised Cell Line.
- (iii) For clarity, (A) this <u>Section 10.8.7(b)</u> (Collaboration Personalised T-Cell Therapy) does not apply to any Collaboration Personalised T-Cell Therapy containing a Cell Line created by the Parties during the Research Term, including all GNE-Modified Personalised Cell Lines created by GNE during the Research Term, and (B) the reductions described in this <u>Section 10.8.7(b)</u> (Collaboration Personalised T-Cell Therapy) shall not apply to any other payments described in <u>Article 10</u> (Financial Terms) of this Agreement, including royalties.
 - (c) "Cell Line Creation Date" means the date on which [***].

(d) [***].

- 10.9. **Economic Effect of Opt-In; Profit-Share.** Notwithstanding anything to the contrary in this Agreement, as of the Opt-In Effective Date, (a) Sections 10.4.1(b) (Adaptimmune Opt-In), 10.5.1(b) (Adaptimmune Opt-In) and 10.7.1(b) (Adaptimmune Opt-In) shall apply with respect to the Collaboration Off-the-Shelf T-Cell Therapies that are the subject of the applicable Opt-In, and (b) in lieu of the royalties contemplated in Section 10.7 (Royalties) on US Net Sales of such Collaboration Off-the-Shelf T-Cell Therapies, the following terms will apply to each Collaboration Off-the-Shelf T-Cell Therapy Directed To such Collaboration Target sold in the US during the applicable Opt-In Term:
- US Net Profits and Net Losses Reports and Payments. During the applicable Opt-In Term, US Net Profits and Net Losses with respect to each Collaboration Off-the-Shelf T-Cell Therapy for the applicable Collaboration Target will be shared equally by the Parties on a quarterly basis. Within [***] days after the end of a calendar quarter, GNE will report in writing to Adaptimmune the elements of the US Net Profits and Net Losses calculation for such quarter for the applicable Collaboration Off-the-Shelf T-Cell Therapy that GNE controls, including the Net Sales in the US and Allowable Expenses incurred or accrued by GNE or any of its Affiliates or Sublicensees. Within such [***] day period, Adaptimmune will report in writing to GNE the Allowable Expenses, if any, Adaptimmune incurred or accrued during such calendar quarter for the applicable Collaboration Off-the-Shelf T-Cell Therapy. Each Party's report will specify all expenses included in its Allowable Expenses, identified in reasonable detail. At the request of a Party, the other Party will provide additional reasonable supporting documentation to support its calculations and make its personnel reasonably available during normal working hours to answer questions. Within the [***] after the end of the applicable calendar quarter or [***] after GNE's receipt of Adaptimmune's report, GNE will provide to Adaptimmune a consolidated financial statement setting forth the US Net Profits and Net Losses for each applicable Collaboration Off-the-Shelf T-Cell Therapy for the calendar quarter, and the following remittances will be paid as set forth below after GNE has provided such consolidated financial statement:
- (a) if there is a Net Profit for the applicable Collaboration Off-the-Shelf T-Cell Therapy in such calendar quarter, then GNE will pay to Adaptimmune a reconciling payment amount equal to Adaptimmune's portion of the Net Profit for such Collaboration Off-the-Shelf T-Cell Therapy for such calendar quarter within [***] days after providing the consolidated financial statement to Adaptimmune; or
- (b) if there is a Net Loss for the applicable Collaboration Off-the-Shelf T-Cell Therapy in such calendar quarter, then GNE will invoice Adaptimmune for the amount equal to Adaptimmune's portion of the Net Loss for such Collaboration Off-the-Shelf T-Cell Therapy for such calendar quarter. Payment by Adaptimmune of such amount will be due [***] days after receiving such an invoice from GNE.
- (c) An example of the reconciliation calculation described in this Section 10.9.1 (US Net Profits and Net Losses Reports and Payments) is set forth on Exhibit \underline{E} (Example US Net Profit and Net Loss Calculation in Section 10.9.1).

10.10. Right of Negotiation. [***].

ARTICLE 11 PAYMENT TERMS; REPORTS; AUDITS

- 11.1. **Timing of Royalty Payment**. All royalty payments shall be made within [***] days of the end of each calendar quarter in which the sale was made.
- 11.2. **Royalty Report**. For each calendar quarter for which GNE has an obligation to make royalty payments, such payments shall be accompanied by a report that specifies for such calendar quarter the following information ("**Net Sales Report**"):
 - 11.2.1. [***]; 11.2.2. [***]; 11.2.3. [***] 11.2.4. [***].

If GNE is reporting Net Sales for more than one Licensed Product, the foregoing information shall be reported on a Licensed Product-by-Licensed Product basis . Where no payment is due, GNE shall also deliver a report indicating such.

11.3. **Mode of Payment**. All payments hereunder shall be made in immediately available funds to the account listed below (or such other account as Adaptimmune shall designate before such payment is due):

[***]

11.4. **Currency of Payments**. All payments under this Agreement shall be made in United States dollars, unless otherwise expressly provided in this Agreement. Net Sales not made in United States dollars shall be converted into an amount in United States dollars as follows: (a) with respect to (i) sales by or on behalf of GNE or its Affiliates or (ii) sales by or on behalf of a given Sublicensee (that is not an Affiliate of GNE) if a royalty is paid to GNE or its Affiliates in the currency of such sales of Licensed Product are made, in each case ((i) and (ii)) using GNE's customary and usual conversion procedures in accordance with the applicable Accounting Standard, consistently applied; and (b) with respect to sales by or on behalf of a given Sublicensee (that is not an Affiliate of GNE) if a royalty is paid to GNE or its Affiliates in United States dollars on such sales of Licensed Product that are not made in United States dollars, using the conversion procedures applicable to royalty payments by such Sublicensee to GNE for such sales under the applicable sublicense agreement; *provided* GNE shall provide Adaptimmune a copy of the portion of such sublicense agreement that are relevant to currency conversion, if requested by Adaptimmune, prior to or concurrently with the first Net Sales Report under which such currency conversion from such Sublicensee is utilized. In addition, the conversion procedures used by GNE will be provided to Adaptimmune.

11.5. **Blocked Currency**. If, at any time, legal restrictions prevent GNE (or an Affiliate or Sublicensee) from remitting part or all of royalty payments when due with respect to any country in the Territory where Licensed Products are sold, GNE shall continue to provide Net Sales Reports for such royalty payments. [***].

11.6. **Taxes**.

- 11.6.1. **Taxes.** Each Party shall comply with applicable laws and regulations regarding filing and reporting for income tax purposes. All payments made under this Agreement shall be made free and clear of any and all taxes, duties, levies, fees or other charges, except for withholding taxes and VAT. GNE shall be entitled to deduct from payments made to Adaptimmune under this Agreement the amount of any withholding taxes required to be withheld, including under Section 11.6.2 (German Withholding Tax Requirement), to the extent paid to the appropriate governmental authority on behalf of Adaptimmune (and not refunded or reimbursed). GNE shall deliver to Adaptimmune, upon request and when available, proof of payment of all such withholding taxes. GNE shall provide reasonable assistance to Adaptimmune in seeking any benefits available to Adaptimmune with respect to government tax withholdings by any relevant law, regulation or double tax treaty. All payments made under this Agreement shall be exclusive of VAT (if applicable) and such VAT shall be paid promptly on receipt of a valid VAT invoice.
- 11.6.2. **German Withholding Tax Requirement**. The Parties acknowledge that payments to Adaptimmune with respect to the rights in Germany granted to GNE under this Agreement may be subject to (i) German income tax pursuant to sec . 49 para . 1 German Income Tax Act (" **GITA**") and (ii) withholding tax pursuant to sec . 50a para . 1 GITA (the " **German WHT Requirement**") . Without limiting anything in Section 11.6.1 (Taxes) above, the following shall apply.
- (a) Adaptimmune shall use reasonable efforts to provide GNE with such reasonable information relevant to assess the applicability of and the tax assessment basis for the German WHT Requirement.
- (b) After reasonably taking into account any comments and information received from Adaptimmune, GNE shall use reasonable best efforts to determine (i) whether the German WHT Requirement is applicable on the licenses granted to GNE under this Agreement and (ii) the amount to be withheld and remitted to the competent German tax authority (including the allocation to and calculation of the assessment basis for the withholding).
- (c) Based on the determination made pursuant to Section 11.6.2(b) (German Withholding Tax Requirement), GNE shall remit the withheld amount to the competent German tax authority in due course. With regards to GNE's payment obligations under this Agreement, any amount paid to the German tax authority pursuant to the preceding sentence shall be deemed as payment to Adaptimmune. GNE shall provide reasonable assistance to Adaptimmune in seeking any benefits available to Adaptimmune with respect to government tax withholdings by the competent German tax authorities, including a valid exemption certificate (Freistellungsbescheinigung).

(d) As soon as GNE has received a valid exemption certificate (Freistellungsbescheinigung) issued by a competent German tax authority (upon the application of Adaptimmune) confirming that Adaptimmune is not required to make a withholding pursuant to the German WHT Requirement, GNE shall not be allowed to make any deductions from any payments pursuant to this Section 11.6.2 (German Withholding Tax Requirement) for the time period specified in the exemption certificate.

11.7. **Records; Inspection**.

- 11.7.1. **Records**. GNE agrees to keep and to procure that Affiliates and Sublicensees agree to keep for [***] from the year of creation, records of all sales of Licensed Products, Development Costs, Launch Costs, and Allowable Expenses for each reporting period in which payments are due hereunder, showing sales of Licensed Products and applicable deductions in sufficient detail to enable the report provided under Section 11.2 (Royalty Report), Section 6.1.6 (Reports; Payments), and Section 10.9.1 (US Net Profits and Net Losses Reports and Payments) to be verified. GNE will include substantially similar rights as set forth in this Section 11.7 (Records; Inspection) in any Sublicense agreement.
- 11.7.2. **Audits**. Adaptimmune shall have the right to request that all Net Sales Reports, all reports of Development Costs and Launch Costs provided pursuant to Section 6.1.6 (Reports; Payments), and GNE reports of US Net Profits and Net Losses provided by GNE pursuant to Section 10.9.1 (US Net Profits and Net Losses Reports and Payments) be verified by an independent, certified and internationally recognized public accounting firm selected by Adaptimmune and acceptable to GNE (the "CPA Firm"), such acceptance not to be unreasonably withheld conditioned or delayed. Such right to request a verified report shall: (a) [***]; (b) [***]; and (c) [***]. Subject to Section 11.7.3 (Confidentiality), GNE shall, upon timely request and at least [***] Business Days advance notice from Adaptimmune and at a mutually agreeable time during its regular business hours, make its records available for inspection by such CPA Firm at such place or places where such records are customarily kept, solely to verify the accuracy of the reports provided under Section 11.2 (Royalty Report) and Section 10.9.1 (US Net Profits and Net Losses Reports and Payments) and related payments due under this Agreement. The CPA Firm shall only state factual findings in the audit reports. The final audit report shall be shared with GNE at the same time that it is shared with Adaptimmune.
- 11.7.3. **Confidentiality**. Prior to any audit under Section 11.7.2 (Audits), the CPA Firm shall enter into a written confidentiality agreement with GNE that: (a) limits the CPA Firm's use of GNE's records to the verification purpose described in Section 11.7.2 (Audits); (b) limits the information that the CPA Firm may disclose to Adaptimmune to the numerical summary of payments due and paid; and (c) prohibits the disclosure of any information contained in such records to any Third Party for any purpose (subject to customary exception for legally compelled disclosure). The Parties agree that all information subject to review under Section 11.7.2 (Audits) or provided by the CPA Firm to Adaptimmune is GNE's Confidential Information, and Adaptimmune shall not use any such information for any purpose that is not germane to Section 11.7.2 (Audits).

- 11.7.4. **Underpayment; Overpayment**. After reviewing the CPA Firm's audit report, GNE shall promptly pay any uncontested, understated amounts due to Adaptimmune. Any overpayment made by GNE shall be promptly refunded or fully creditable against amounts payable in subsequent payment periods, at GNE's election. Any audit under Section 11.7.2 (Audits) shall be at Adaptimmune's expense; provided, that GNE shall reimburse reasonable audit fees for a given audit if the results of such audit reveal that GNE underpaid Adaptimmune with respect to royalty payments by [***] or more for the audited period and such audited period includes [***] within a calendar year.
- 11.7.5. **Audit Dispute**. In the event of a dispute with respect to any audit under Section 11.7.4, Adaptimmune and GNE shall work in good faith to resolve the disagreement. If the Parties are unable to reach a mutually acceptable resolution of any such dispute within [***] days, either Party may submit the dispute for resolution to a certified public accounting firm jointly selected by each Party's certified public accountants or to such other Person as the Parties shall mutually agree (the "Arbitrator"). The decision of the Arbitrator shall be final and the costs of such arbitration as well as the initial audit shall be borne between the Parties in such manner as the Arbitrator shall determine. Not later than [***] after such decision and in accordance with such decision, the audited Party shall pay the additional amounts, with interest from the date originally due as provided in Section 11.7.4 (Underpayment; Overpayment), or the auditing Party shall reimburse the excess payments, as applicable.
- 11.7.6. **Adaptimmune Records and Opt-In**. If Adaptimmune exercises its Opt-In right pursuant to <u>Section 6.1</u> (Adaptimmune Opt-In), the provisions of this <u>Section 11.7</u> (Records; Inspection) shall apply, *mutatis mutandis*, to Adaptimmune and its Affiliates and sublicensees with respect to Adaptimmune's Allowable Expenses and the records relating thereto.

ARTICLE 12 INTELLECTUAL PROPERTY; OWNERSHIP

- 12.1. **Definitions and Ownership**. As used herein this <u>Article 12</u> (Intellectual Property; Ownership):
- 12.1.1. "Adaptimmune Platform IP" means any Know-How and Patents specifically relating to the Adaptimmune Platform (including Adaptimmune Background IP specifically related to Adaptimmune's methods for differentiating T-Cells from iPS Cells or specifically related to Adaptimmune's Cell Therapy Platform) owned or Controlled by Adaptimmune as of the Effective Date. Adaptimmune shall retain ownership of Adaptimmune Platform IP.
- 12.1.2. "Adaptimmune Platform Improvement IP" means any Collaboration IP that specifically relates to the Adaptimmune Platform provided by Adaptimmune to GNE under the Agreement, other than Overlapping Improvement IP. [***].
- 12.1.3. "**Background IP**" means, with respect to a Party, all Know-How and Patents that such Party Controls as of the Effective Date and during the Term of this Agreement through

efforts outside the course of activities conducted under this Agreement. Each Party shall retain ownership of its Background IP.

- 12.1.4. "GNE Platform Improvement IP" means Collaboration IP that specifically relates to GNE Proprietary Platform Methods provided by GNE to Adaptimmune under the Agreement, other than Overlapping Improvement IP. [***].
- 12.1.5. "GNE Proprietary Platform Methods" means GNE's proprietary processes or technology for making, engineering or developing Cell Therapies, including [***]. GNE shall retain ownership of GNE Proprietary Platform Methods.
- 12.1.6. "GNE Receptor IP" means Collaboration IP that specifically relates to any GNE-Provided $\alpha\beta$ Receptor or any other $\alpha\beta$ Receptor provided by or on behalf of GNE that is incorporated in any Licensed Product, or is a modification or derivative of any such Receptor, and any improvements thereto. GNE shall own GNE Receptor IP, [***].
 - 12.1.7. [***].
- 12.1.8. "Other Collaboration IP" means Collaboration IP not specifically relating to any of the Intellectual Property definitions described in this <u>Section 12.1</u> (Definitions and Ownership). Ownership of Other Collaboration IP shall follow inventorship in accordance with <u>Section 12.2.2</u> (Inventorship).
- 12.1.9. "**Overlapping Improvement IP**" means Collaboration IP that is specifically related to both the GNE Platform Improvement IP and Adaptimmune Platform Improvement IP. [***].

12.2. Disclosure; Inventorship; Assignment and Cooperation; Joint Ownership.

- 12.2.1. **Disclosure**. During the Term, each Party shall promptly disclose to the other Party any potentially patentable Collaboration IP as set forth in <u>Article 12</u> (Intellectual Property; Ownership) that is discovered, conceived of, or reduced to practice by or for the disclosing Party in the course of the activities performed by or for such Party in connection with this Agreement.
- 12.2.2. **Inventorship**. Inventorship shall, to the extent legally permitted, be determined according to the US patent law. In the event of a dispute between the Parties over inventorship of Collaboration IP or the subject matter or which Party owns Collaboration IP, the Parties shall, notwithstanding anything to the contrary in <u>Article 12</u> (Intellectual Property; Ownership), refer such dispute to a mutually acceptable independent outside patent counsel to determine inventorship and shall use all reasonable efforts to do so in an efficient and expedient manner. The Parties agree that the decision rendered by such independent outside patent counsel shall be the sole, exclusive and binding resolution and remedy between them regarding such dispute, and the Parties shall share equally the fees and expenses of the independent outside patent counsel in resolving such dispute.
- 12.2.3. **Assignment; Cooperation**. Each Party ("**Assigning Party**") hereby assigns to the other Party ("**Owning Party**") all of such Assigning Party's right, title and interest in, to and

under the Intellectual Property described in Section 12.1.2 (" Adaptimmune Platform Improvement IP"), 12.1.4 (" GNE Platform Improvement IP"), 12.1.6 (" GNE Receptor IP"), 12.1.7 ("[***]") that is to be assigned to the Owning Party (and not jointly owned by the Parties). To the extent that any Collaboration IP is [***]. The Parties each hereby grant to one another the rights necessary to accomplish the ownership provisions set forth in this Article 12 (Intellectual Property; Ownership). Each Party shall execute such further documentation as may be necessary or appropriate, and provide reasonable assistance and cooperation, to implement the provisions of this Article 12 (Intellectual Property; Ownership). Each Party shall require all of its employees, Affiliates and any Third Parties working pursuant to this Agreement on its behalf, to assign (or otherwise convey rights) to such Party any Patents and rights in Know-How discovered, conceived or reduced to practice by such employee, Affiliate or Third Party, and to cooperate with such Party in connection with obtaining patent protection therefore.

- 12.2.4. **Joint Ownership**. Each Party shall have the right to use, license and otherwise exploit Collaboration IP that is jointly owned without any restriction or obligation to account to the other Party.
- **CREATE Act**. It is the intention of the Parties that this Agreement is a "joint research 12.2.5. agreement" as that phrase is defined in Public Law 108-53 (the "Create Act"). In the event that either Party to this Agreement intends to overcome a rejection of a claimed invention within the Adaptimmune Licensed IP or Collaboration IP pursuant to the provisions of the Create Act, such Party must first obtain the prior written consent of the other Party. Neither Party will invoke this Agreement as a joint research agreement under the Create Act to overcome such an objection without the prior written consent of the other Party. If the other Party provides such written consent, such Party shall limit any amendment to the specification or statement to the patent office with respect to this Agreement to that which is strictly required by 35 USC § 103(c) and the rules and regulations promulgated thereunder and which is consistent with the terms and conditions of this Agreement (including the scope of the Research Program activities). To the extent that the Parties agree that, in order to overcome a rejection of a claimed invention within the Adaptimmune Licensed IP or Collaboration IP pursuant to the provisions of the Create Act, the filing of a terminal disclaimer is required or advisable, the Parties shall first agree on terms and conditions under which the patent application subject to such terminal disclaimer and the patent or application over which such application is disclaimed shall be jointly enforced, to the extent that the Parties have not previously agreed to such terms and conditions.

12.3. **Patent Prosecution**.

- 12.3.1. **Sole IP**. The Party that solely owns the applicable Intellectual Property described in <u>Section 12.1</u> (Definitions and Ownership) shall, at its sole discretion and expense, have the right (but not the obligation) to Prosecute and Maintain Patents that describe, cover or claim such owned Intellectual Property (each a "**Sole Prosecuted Patent**") as further described in this <u>Section 12.3</u> (Patent Prosecution).
- 12.3.2. **Joint Collaboration IP** . [***] shall at its sole discretion (subject to its obligations under the JIPC) and expense, have the right (but not the obligation) to Prosecute and Maintain Patents within the jointly owned Collaboration IP, save that the Prosecution and

Maintenance of any jointly owned Collaboration IP will be discussed at the JIPC and the strategy for such agreed at the JIPC. Each Party may additionally file divisionals under any jointly owned Collaboration IP where specific to either Party's activities or products as further described in <u>Section 2.3.1</u> (Role of the JIPC) and the Party filing such shall be responsible for such Prosecution and Maintenance, subject in each case to discussion at JIPC.

- 12.3.3. Adaptimmune. [***]. Adaptimmune will provide updates at JIPC meetings in relation to Patents within the Adaptimmune Platform IP, including those listed in Exhibit C (Adaptimmune Licensed Patent Rights) to the extent such Patents continue to Cover any Licensed Products or the Development of any Licensed Products. [***]. GNE will provide all reasonable cooperation and assistance to Adaptimmune at Adaptimmune's reasonable request and at Adaptimmune's expense in Prosecution and Maintenance of the Adaptimmune Prosecuted Patents, including making data, reports, and scientific personnel reasonably available to prepare and prosecute patent applications. [***].
- 12.3.4. **GNE**. GNE shall, at its sole discretion (subject to its obligations under the JIPC) and expense, have the right (but not the obligation) to Prosecute and Maintain [***], subject in the latter case to JIPC involvement. GNE will keep Adaptimmune reasonably informed of the status of the Prosecution and Maintenance of GNE Platform Improvement IP [***]. Adaptimmune will provide all reasonable cooperation and assistance to GNE at GNE's reasonable request and at Adaptimmune's expense in Prosecution and Maintenance of such GNE Prosecuted Patents, including making data, reports, and scientific personnel reasonably available to prepare and prosecute patent applications. [***].

12.4. Enforcement Rights for Infringement by Third Parties.

- 12.4.1. **Notice**. Each Party shall promptly notify, in writing, the other Party upon learning of any actual or suspected infringement or misappropriation of the Adaptimmune Licensed IP or Collaboration IP by the manufacture, commercialization, use, import, export, offer for sale or sale by a Third Party of a product that is competitive with one or more Licensed Products (each an "**Infringement**"), or of any claim of invalidity, unenforceability, or non-infringement of such Adaptimmune Licensed IP or Collaboration IP by a Third Party that is developing, manufacturing, or commercializing a product that is competitive with one or more Licensed Products. At the request of the Party receiving such notice, the other Party shall use commercially reasonable efforts to provide all evidence in its possession pertaining to the actual or suspected Infringement or claim that it can disclose without breach of a pre-existing obligation to a Third Party or waiver of privilege.
- 12.4.2. **Enforcement Actions**. The Parties shall consult (through the JIPC or as otherwise agreed by the Parties) as to potential strategies to terminate suspected or potential Infringement, including by initiating IPRs, post-grant reviews, oppositions, or other actions against a Third Party's Patent that interferes with either the Adaptimmune Platform IP or Collaboration IP (each IPR, post-grant review, opposition or other action, an "**Opposition Proceeding**"), consistent with the overall goals of this Agreement. If the Parties fail to agree on such strategies:

- (a) [***] shall have the (a) sole right, but not the obligation, to seek to abate any actual or suspected Infringement by a Third Party, or to file and control suit or other action or proceeding, including an Opposition Proceeding, against any Third Party for Infringement in its own name and entirely under its own direction and control, in each case under the Patent rights within the [***], including filing suit against any Third Party for Infringement of a Patent claiming a composition of matter, or use thereof, of a GNE-Provided $\alpha\beta$ Receptor (including any modifications thereto) incorporated into a Licensed Product or a cell expressing a GNE-Provided that [***] shall not be permitted to enforce [***] without [***] prior consent, such consent not to be unreasonably withheld, conditioned or delayed. If [***] requests, [***] shall reasonably cooperate with [***] in the planning and execution of any such action to enforce such Patent rights (including the obligation to be named or joined as a party in a lawsuit, as applicable). For avoidance of doubt, [***] will not have the right to bring an infringement action or other proceeding to abate any actual or suspected Infringement by a Third Party without the written consent from [***].
- (b) Notwithstanding the foregoing, [***] shall in good faith consider bringing an appropriate suit or other action against a Third Party Infringement under [***] at [***] request. If [***] declines to bring a suit or other action to abate a Third Party's Infringement at [***] reasonable request, then each claim in the specific Patent that [***] declined to enforce despite such reasonable request shall [***].
- (c) [***] shall have the exclusive right, but not the obligation, to bring a suit or other action or proceeding including an Opposition Proceeding specifically affecting [***] it owns to the extent the action or proceeding is not brought to abate Infringement by a Third Party that is developing, manufacturing, or commercializing a product that is competitive with one or more Licensed Products.
- (d) The non-controlling Party shall cooperate with the Party controlling any such action to abate or enforce (as may be reasonably requested by the controlling Party and at the controlling Party's expense), including, if necessary, by being joined as a party *provided*, that the non-controlling Party shall be indemnified by the controlling Party as to any costs or expenses, and shall have the right to be represented by its own counsel at its own expense. The Party controlling any such action shall keep the other Party updated with respect to any such action, including providing copies of all documents received or filed in connection with any such action.
- 12.4.3. **Settlement**. The Party controlling any such enforcement action described in <u>Section 12.4.2</u> (Enforcement Actions) (an "**Enforcement**"), at its sole discretion, may take reasonable actions to terminate any alleged Infringement without litigation; *provided*, that if any such arrangement would adversely affect the non-controlling Party's rights under this Agreement, then that arrangement is subject to the non-controlling Party's prior written consent. The Party controlling any Enforcement may not settle or consent to an adverse judgment without the express written consent of the non-controlling Party (such consent not to be unreasonably withheld or delayed).
- 12.4.4. **Costs and Expenses** . The Party controlling any Enforcement shall bear all costs and expenses, including but not limited to litigation expenses, related to such Enforcement.

- 12.4.5. **Damages**. Unless otherwise mutually agreed by the Parties, and subject to the respective indemnity obligations of the Parties set forth in <u>Article 16</u> (Indemnification), all damages, amounts received in settlement, judgment or other monetary awards recovered in an Enforcement with respect to activities of the Third Party that occurred prior to the effective date of such award shall be shared as follows:
- (a) first, to reimburse the controlling Party for costs and expenses incurred under <u>Section</u> 12.4.2 (Enforcement Actions);
 - (b) second, any amounts remaining to be allocated as follows: [***].

For the avoidance of doubt, if any settlement results in the granting to the person or entity accused of infringement or misappropriation of a sublicense of any of the Collaboration IP from GNE with running royalties payable on post-settlement sales by the alleged infringer, such alleged infringer shall be deemed to be a Sublicensee of GNE and such royalties on post-settlement sales: (i) shall be subject to all applicable royalty obligations hereunder; and (ii) shall not be subject to this Section 12.4.5 (Damages); provided, that any upfront or event payments or the like shall be deemed monetary awards and subject to Section 12.4.5(b) (Damages). In the event the sublicense agreement includes other Intellectual Property that is not a subject of this Agreement, then unless allocation of payments to Collaboration IP is clear within agreement then GNE shall have the right, in its reasonable discretion, to apportion such upfront, event payments or royalties prorata between such other Intellectual Property and any of the Collaboration IP.

12.5. Third Party Infringement Claims.

- 12.5.1. **Notice.** In the event that a Third Party shall make any claim, give notice, or bring any suit or other *inter partes* proceeding against GNE or Adaptimmune, or any of their respective Affiliates or licensees or customers, for infringement or misappropriation of any Intellectual Property rights with respect to the Research, Development, making, using, selling, offering for sale, import or export of any Licensed Product ("**Third Party Infringement Claim**"), in each case, the Party receiving notice of a Third Party Infringement Claim shall promptly notify the other Party and use commercially reasonable efforts to provide all evidence in its possession pertaining to the claim or suit that it can disclose without breach of a pre-existing obligation to a Third Party or waiver of privilege.
- 12.5.2. **Defence**. The Parties shall consult (through the JIPC or otherwise) as to potential strategies to defend against any Third Party Infringement Claim, including initiating an Opposition Proceeding or by being joined as a Party, in each case consistent with the overall goals of this Agreement. If the Parties fail to agree on such strategies, and subject to the respective indemnity obligations of the Parties set forth in Article 16 (Indemnification), the Parties shall cooperate with each other in all reasonable respects in the defence of any Third Party Infringement Claim or raising of any counterclaim related thereto.
- (a) [***] shall have the first right, but not the obligation, to defend any Third Party Infringement Claim related to the [***], including by initiating an Opposition Proceeding against a Third Party Patent. If [***] does not, within one hundred twenty (120) days of receipt of a notice under Section 12.5.1 (Notice), take steps to defend the Third Party Infringement Claim,

then to the extent that such Third Party Infringement Claim is brought against [***] and impairs [***] ability to make, use or sell the Licensed Products, [***] shall have the right, but not the obligation, to take action, including initiating an Opposition Proceeding, to defend or enforce against such Third Party Infringement Claim; *provided*, that if [***] is diligently pursuing ongoing settlement discussions at the end of such one hundred and twenty (120) day period then [***] shall not be permitted to exercise such right unless such settlement discussions cease without reaching settlement. To the extent that any action that is required under any Third Party Infringement Claim described in this Section 12.5.2(a) (Defence) is brought against [***] and prior to [***] agreeing to [***] taking over the control of such Third Party Infringement Claim, [***] may take all steps reasonably required to defend itself in such claim at its own expense.

- (b) [***] shall have the first right, but not the obligation, to defend or enforce against any Third Party Infringement Claim directed to any Licensed Products, including initiating an Opposition Proceeding against a Third Party Patent. If [***] does not, within one hundred twenty (120) days of receipt of a notice under Section 12.5.1 (Notice), take steps to defend the Third Party Infringement Claim, then solely to the extent that such Third Party Infringement Claim is brought against [***], [***] shall have the right, but not the obligation, to take action, including initiating an Opposition Proceeding, to enforce against such Third Party Infringement Claim; provided, that if [***] is diligently pursuing ongoing settlement discussions at the end of such one hundred and twenty (120) day period then [***] shall not be permitted to exercise such right unless such settlement discussions cease without reaching settlement. To the extent that any action that is required under any Third Party Infringement Claim described in this Section 12.5.2(b) (Defence) is brought against [***] and prior to [***] agreeing to [***] taking over the control of such Third Party Infringement Claim, [***] may take all steps reasonably required to defend itself in such claim at its own expense.
- 12.5.3. **Defence and Counterclaim.** The non-controlling Party shall cooperate with the Party controlling in connection with any such defence and counterclaim (as may be reasonably requested by the controlling Party and at the controlling Party's expense), including, if necessary, by being joined as a party, *provided*, that the non-controlling party shall be indemnified by the controlling party as to any costs or expenses, and shall have the right to be represented by its own counsel at its own expense. The Party controlling any such action shall keep the other Party updated with respect to any such action, including providing copies of all documents received or filed in connection with any such action. Any counterclaim or other similar action by a Party, to the extent such action involves any enforcement of rights under the Collaboration IP or Adaptimmune Platform IP, will be treated as an enforcement action subject to Section 12.4 (Enforcement Rights for Infringement by Third Parties).
- 12.5.4. **Settlement**. If any such defence under <u>Section 12.5.2</u> (Defence) would adversely affect the other Party's rights under this Agreement or impose a financial obligation upon the other Party or grant rights in respect, or affect the validity or enforceability, of the other Party's Patents, then any settlement, consent judgment or other voluntary final disposition of such Third Party Infringement Claim shall not be entered into without the consent of the other Party (such consent not to be unreasonably withheld).

- 12.5.5. **Costs and Expenses**. The Party controlling the defence of any Third Party Infringement Claim shall bear all costs and expenses, including but not limited to litigation expenses, to defend against any Third Party Infringement Claim.
- 12.6. **Trademarks**. GNE shall be free to use and to register in any trademark office worldwide, at its sole cost, any trademark for use with a Licensed Product in its sole discretion. GNE shall own all right, title and interest in and to any such trademark (including any and all claims and causes of action, rights to and claims for damages, restitution and injunctive and other legal and equitable relief for past, present and future infringement, dilution, misappropriation, violation, misuse, breach or default, with the right but no obligation to sue for such legal and equitable relief and to collect, or otherwise recover, any such damages) in its own name during and after the Term.
- 12.7. **Unified Patent Court (Europe)** At any time prior to the end of the "transitional period" as such term is used in Article 83 of the Agreement on a Unified Patent Court between the participating Member States of the European Union, for a given relevant Patent within the Collaboration IP in the EU, GNE may request in writing that Adaptimmune either (a) opt out from the exclusive competence of the Unified Patent Court or (b) if applicable, withdraw a previously-registered opt-out, and Adaptimmune shall notify the Registry, pay any such registry fee and take such other action as may be necessary to effect the opt-out or opt-out withdrawal. Adaptimmune shall reasonably consider such request.
- 12.8. Common Interest Disclosures . The Parties: (a) share a common legal and commercial interest in such disclosure that is subject to such privileges and protections; (b) are or may become joint defendants in proceedings to which the information covered by such protections and privileges relates; (c) intend that such privileges and protections remain intact should either Party become subject to any actual or threatened proceeding to which the disclosing Party's Confidential Information covered by such protections and privileges relates; and (d) intend that after the Effective Date both the receiving Party and the disclosing Party shall have the right to assert such protections and privileges. With regard to any information or opinions disclosed pursuant to this Agreement by one Party to each other regarding Intellectual Property or technology owned by Third Parties, the Parties agree that they have a common legal interest in determining whether, and to what extent, Third Party Intellectual Property rights may affect the conduct of the Research Program or Licensed Products, and have a further common legal interest in defending against any actual or prospective Third Party claims based on allegations of misuse or infringement of Intellectual Property rights relating to the conduct of the Research Program or Licensed Products. Accordingly, the Parties agree that all such information and materials obtained by Adaptimmune and GNE from each other will be used solely for purposes of the Parties' common legal interests with respect to the conduct of this Agreement. All information and materials will be treated as protected by the attorney-client privilege, the work product privilege, and any other privilege or immunity that may otherwise be applicable. By sharing any such information and materials, neither Party intends to waive or limit any privilege or immunity that may apply to the shared information and materials. Neither Party shall have the authority to waive any privilege or immunity on behalf of the other Party without such other Party's prior written consent, nor shall the waiver of privilege or immunity resulting from the conduct of one Party be deemed to apply against any other Party. Notwithstanding the foregoing, neither Party's attorney represents the other Party. Neither Party is waiving, nor shall be deemed to have waived or diminished, any of its attorney work product

protections, attorney-client privileges or similar protections and privileges or the like as a result of disclosing information pursuant to this Agreement or any of its Confidential Information (including Confidential Information related to pending or threatened litigation) to the receiving Party, regardless of whether the disclosing Party has asserted, or is or may be entitled to assert, such privileges and protections.

- 12.9. **Patent Term Extensions**. With respect to Collaboration IP, the Parties shall use Commercially Reasonable Efforts to obtain all available patent term extensions, adjustments or restorations, or supplementary protection certificates ("**SPCs**", and together with patent term extensions, adjustments and restorations, "**Patent Term Extensions**"). [***] shall have the sole right, but not the obligation, to file or request [***] to file any Patent Term Extensions related to Licensed Products at [***] expense. [***] shall execute such authorizations and other documents and take such other actions as may be reasonably requested by [***] to obtain such Patent Term Extensions, including designating [***] as its agent for such purpose as provided in 35 USC § 156. Adaptimmune shall cooperate in all reasonable ways in connection therewith, including, at [***] request, filing such Patent Term Extensions on behalf of [***].
- 12.10. **Patent Listings**. With respect to any filings made to Regulatory Authorities with respect to any Licensed Products, including, as required or allowed in the United States, the FDA's Orange or Purple Book, if applicable, or outside the United States, other international equivalents, GNE will have the sole right to make any such decision whether to list [***], but in all events will comply with applicable law; *provided* that GNE will consider in good faith any timely comments received from or on behalf of Adaptimmune with respect to such filings prior to submission. Upon GNE's request, Adaptimmune will reasonably cooperate in the implementation of GNE's decision made under this Section 12.10 (Patent Listings).

ARTICLE 13 CONFIDENTIALITY

- 13.1. **Non-Use and Non-Disclosure of Confidential Information**. During the Term, and for a period of [***] years thereafter, each Party shall: (a) except to the extent permitted by this Agreement or otherwise agreed to in writing, keep confidential and not disclose to any Third Party any Confidential Information of the other Party; (b) except in connection with activities contemplated by, the exercise of rights permitted by, or in order to further the purposes of this Agreement or otherwise agreed to in writing, not use for any purpose any Confidential Information of the other Party; and (c) take all reasonable precautions to protect the Confidential Information of the other Party (including all precautions a Party employs with respect to its own confidential information of a similar nature and taking reasonable precautions to assure that no unauthorized use or disclosure is made by others to whom access to the Confidential Information of the Party is granted). Notwithstanding the foregoing, to the extent that a Party maintains any Confidential Information as trade secret and expressly notifies the other Party accordingly in writing, the obligations of non-use and confidentiality over such trade secret information shall continue to apply indefinitely, unless and until the exclusions under Section 13.2 (Exclusivity Regarding Confidential Information) apply.
- 13.2. **Exclusions Regarding Confidential Information**. Notwithstanding anything set forth in this Article 13 (Confidentiality) to the contrary, the obligations of Section 13.1 (Non-Use and Non-

Disclosure of Confidential Information) shall not apply to the extent that the Party seeking the benefit of the exclusion can demonstrate that the Confidential Information of the other Party:

- 13.2.1. **Known to Receiving Party.** was already known to the receiving Party, other than under an obligation of confidentiality, at the time of receipt by the receiving Party;
- 13.2.2. **Was Generally Available to Public.** was generally available to the public or otherwise part of the public domain at the time of its receipt by the receiving Party;
- 13.2.3. **Became Generally Available to Public.** became generally available to the public or otherwise part of the public domain after its receipt by the receiving Party other than through any act or omission of the receiving Party in breach of this Agreement;
- 13.2.4. **Received Without Obligation of Confidentiality.** was received by the receiving Party without an obligation of confidentiality from a Third Party having the right to disclose such information without restriction;
- 13.2.5. **Independently Developed.** was independently developed by or for the receiving Party without use of or reference to the Confidential Information of the other Party as evidenced by contemporaneous written records; or
- 13.2.6. **Released.** was released from the restrictions set forth in this Agreement by express prior written consent of the Party.
- 13.3. **Authorized Disclosures of Confidential Information**. Notwithstanding the foregoing, a Party may use and disclose the Confidential Information of the other Party as follows:
- 13.3.1. **Subject to Filing of Agreement.** Subject to <u>Section 14.5</u> (Filing of Agreement), if required by law, rule or governmental regulation, including as may be required in connection with any filings made with, or by the disclosure policies of a major stock exchange; *provided*, that the Party seeking to disclose the Confidential Information of the other Party: (i) use all reasonable efforts to inform the other Party prior to making any such disclosures and cooperate with the other Party in seeking a protective order or other appropriate remedy (including redaction); and (ii) whenever possible, request confidential treatment of such information;
- 13.3.2. **Prosecution and Maintenance of Patents.** To the extent such use and disclosure is reasonably required in the Prosecution and Maintenance of a Patent within the Collaboration IP or Adaptimmune Licensed IP in accordance with this Agreement, *provided* that the Party controlling such Prosecution and Maintenance of a Patent shall notify the other Party (whether through the JIPC, or if the JIPC has been disbanded pursuant to this Agreement, through the Alliance Managers) of any inclusion of such other Party's Confidential Information and work with such Party in good faith to remove such Confidential Information from the disclosure or to amend and restrict the disclosure of such Confidential Information to the extent reasonable. Notwithstanding the foregoing, any inclusion of trade secrets of the other Party that has been specifically identified in writing in the Prosecution and Maintenance of a Patent within the Collaboration IP or Adaptimmune Licensed IP in accordance with this Agreement shall require the written consent of the Party owning such trade secrets before inclusion in any such disclosure;

- 13.3.3. **Regulatory Approval.** As reasonably necessary to obtain or maintain any Regulatory Approval, including to conduct preclinical studies and clinical trials and for pricing approvals, for any Licensed Products, *provided*, that the disclosing Party shall take all reasonable steps to limit disclosure of the Confidential Information outside such regulatory agency and to otherwise maintain the confidentiality of the Confidential Information;
- 13.3.4. **Lawful Action.** To take any lawful action that it deems necessary to protect its interest under, or to enforce compliance with the terms and conditions of, this Agreement; or
- 13.3.5. **Need To Know Basis.** To the extent necessary, to permitted sublicensees, licensees, collaborators, vendors, consultants, agents, attorneys, contractors and clinicians under written agreements of confidentiality at least as restrictive as those set forth in this Agreement, who have a need to know such information in connection with such Party performing its obligations or exercising its rights under this Agreement. Further, the receiving Party may disclose Confidential Information to existing or potential acquirers, merger partners, permitted collaborators, licensees and sources of financing or to professional advisors (*e.g.*, attorneys, accountants and prospective investment bankers) involved in such activities, for the limited purpose of evaluating such transaction, collaboration or license and under appropriate conditions of confidentiality, only to the extent necessary and with the agreement by those permitted individuals to maintain such Confidential Information in strict confidence.
- 13.4. **Terms of this Agreement**. The Parties agree that this Agreement and the terms hereof will be considered Confidential Information of both Parties.
- 13.5. **Termination of Prior Agreements**. As of the Effective Date, as between the Parties, this Agreement supersedes the Mutual Confidentiality Agreement between [***] (" **Non-Disclosure Agreement**") but only insofar as each relates to the subject matter of this Agreement. All "Confidential Information" (as defined in such agreement) exchanged between the Parties thereunder relating to the subject matter of this Agreement shall be deemed Confidential Information hereunder.
- 13.6. **No License**. As between the Parties, Confidential Information disclosed hereunder shall remain the property of the disclosing Party. Disclosure of Confidential Information to the other Party shall not constitute any grant, option or license to the other Party, beyond those licenses expressly granted under <u>Article 7</u> (Licenses), under any patent, trade secret or other rights now or hereinafter held by the disclosing Party.

ARTICLE 14 PUBLICITY; PUBLICATIONS; USE OF NAME

14.1. **Publicity and Disclosures**. GNE hereby agrees to Adaptimmune issuing the press release set forth in **Schedule 14.1** concerning the execution of this Agreement within [***] days after the Execution Date. The text of any other press releases or other public disclosure or announcement concerning this Agreement, the subject matter hereof, information arising from the conduct of activities under this Agreement, Licensed Products, or the research, development or commercial results of Licensed Products hereunder (a "**Disclosure**") shall be addressed pursuant to <u>Section 14.1</u> (Publicity and Disclosures) through <u>Section 14.4</u> (Approved Disclosures). Any such

Disclosure shall not include any financial terms of this transaction other than to the extent allowed pursuant to Section 14.1 (Publicity and Disclosures) through Section 14.4 (Approved Disclosures) or as may otherwise be agreed in writing by the Parties on a case-by-case basis or if the amount of any payment accrued or actually paid to either Party pursuant to this Agreement is required to be disclosed in the financial accounts of either Party, in each case without disclosing the details of any event that gave rise to such payment. The provisions of this Section 14.1 (Publicity and Disclosures) are in addition to the provision of Article 13 (Confidentiality).

- 14.2. **Disclosures by Adaptimmune**. If Adaptimmune desires to make a Disclosure that is not permitted pursuant to <u>Section 14.1</u> (Publicity and Disclosures) or <u>Section 14.4</u> (Approved Disclosures), <u>Section 14.6.1</u> (Prior Written Consent), <u>Section 14.6.3</u> (Non-Disclosing Party Rights) or <u>Section 13.3</u> (Authorized Disclosures of Confidential Information), it shall obtain GNE's prior written approval for the proposed Disclosure, which approval (a) [***].
- 14.3. **Disclosures by GNE.** Subject to <u>Article 13</u> (Confidentiality), other than Disclosures of the terms and conditions of this Agreement, or Disclosures containing Confidential Information of Adaptimmune, Disclosures by GNE shall not be subject to either review or approval by Adaptimmune . If GNE wishes to make any other Disclosure (a) which is not expressly permitted by the terms and conditions of this Agreement, or (b) that contains the Confidential Information of Adaptimmune, then it shall [***].
- 14.4. **Approved Disclosures**. Either Party may make subsequent public disclosure of the contents of any Disclosure which becomes public without the further approval of the Party whose consent was required; *provided*, that such content is not materially altered or changed.
- 14.5. **Filing of Agreement**. Notwithstanding the generality of Section 13.3 (Authorized Disclosures of Confidential Information), if either Party determines that the filing of this Agreement is required to comply with the disclosure requirements of any applicable securities laws or any applicable securities exchange, the filing Party shall, at the request of the other Party, seek confidential treatment for portions of this Agreement from the relevant authority and shall provide such other Party with the opportunity, for no less than [***] days (before the date of the proposed filing), to review and comment on any such proposed filing, and shall thereafter provide reasonable advance notice and opportunity for comment on any subsequent changes to such filing.
- 14.6. **Scientific Publications**. Notwithstanding Section 14.1 (Publicity and Disclosures) through Section 14.4 (Approved Disclosures), both Parties recognize that the publication or disclosure of papers, presentations, abstracts or any other written or oral presentations regarding results of and other information regarding the Licensed Products may be beneficial to both Parties, *provided*, that such publications or presentations are subject to reasonable controls to protect Confidential Information, the patentability of inventions and other commercial considerations. Accordingly, the following shall apply with respect to papers and presentations proposed for disclosure by either Party:
- 14.6.1. **Prior Written Consent.** Adaptimmune shall not disclose any information arising from this Agreement or any Confidential Information of GNE in scientific publications or public presentations without the prior written consent of GNE or unless <u>Section 13.2</u> (Exclusions

Regarding Confidential Information) or <u>Section 13.3</u> (Authorized Disclosures of Confidential Information) apply; *provided*, that, Adaptimmune may publish information on the Adaptimmune Platform to the extent such information (i) does not relate to any Licensed Product or any Collaboration iPS Cell Line and (ii) is not Confidential Information of GNE; *provided*, further, that, Adaptimmune may publish the genetic edits made to any Permitted Use Cell Line to the extent necessary for the identification or description of such Permitted Use Cell Line in connection with its own use of such Permitted Use Cell Line outside of the activities under this Agreement as and to the extent permitted in accordance with <u>Section 3.5.2</u> (Adaptimmune Permitted Uses of Research Program Cell Lines) and licensed in accordance with <u>Section 7.2</u> (License from GNE to Adaptimmune).

- 14.6.2. **GNE Freedom of Publication.** GNE shall be free to publish any paper or presentation proposed for disclosure by GNE, so long as such paper or presentation does not contain any Confidential Information of Adaptimmune. For clarity, GNE shall not be permitted to publish or otherwise disclose any Confidential Information of Adaptimmune except as may be expressly permitted pursuant to Section 13.2 (Exclusions Regarding Confidential Information), Section 13.3 (Authorized Disclosures of Confidential Information) or Section 14.6.3 (Non-Disclosing Party Rights).
- 14.6.3. Non-Disclosing Party Rights. With respect to any paper or presentation proposed for disclosure by: (i) GNE that includes Confidential Information of Adaptimmune or where Adaptimmune's consent is required pursuant to Article 14 (Publicity; Publications; Use of Name); or (ii) Adaptimmune that contains information arising from this Agreement or any Confidential Information of GNE or where GNE's consent is required pursuant to this Article 14 (but excluding any information on the Adaptimmune Platform to the extent such information does not relate to any Licensed Product or any Collaboration iPS Cell Line), (in each case, the "Disclosing Party"), the other Party (the "Non-Disclosing Party") shall have the right to review and approve any such proposed paper or presentation. The Disclosing Party shall submit to the Non-Disclosing Party the proposed publication or presentation (including posters, slides, abstracts, manuscripts and written descriptions of oral presentations) at [***] days for abstracts) prior to the date of submission for publication or the date of presentation, whichever is earlier, of any of such submitted materials. The Non-Disclosing Party shall review such submitted materials and respond to the Disclosing Party as soon as reasonably possible, but in any case within [***] days ([***] days for abstracts) of receipt thereof. At the option of the Non-Disclosing Party in the Non-Disclosing Party's sole discretion, the Disclosing Party shall: (x) delete from such proposed publication or presentation any Confidential Information of the Non-Disclosing Party; or (y) delay the date of such submission for publication or the date of such presentation for a period of time sufficiently long (but in no event longer than [***] days) to permit the Non-Disclosing Party to seek appropriate patent protection. Once a publication has been approved by the Non-Disclosing Party and made public, either Party may make subsequent public disclosure of the contents of such publication without the further approval; provided, that such content is not presented in a form that materially alters the subject matter therein.
- 14.7. **No Right to Use Names**. Except as expressly provided herein, no right, express or implied, is granted by this Agreement to use in any manner the name of "Adaptimmune" or "Genentech"

or any other trade name, symbol, logo or trademark of the other Party or its Affiliates in connection with the performance of this Agreement.

ARTICLE 15 REPRESENTATIONS, WARRANTIES AND COVENANTS

- 15.1. **Mutual Representations and Warranties**. Each Party represents and warrants to the other Party that as of the Execution Date and Effective Date that:
- 15.1.1. **Valid Organization.** It is validly organized under the laws of its jurisdiction of incorporation;
- 15.1.2. **Obtained Necessary Consents and Approvals.** It has obtained all necessary consents, approvals and authorizations of all governmental authorities and other persons or entities required to be obtained by it in connection with this Agreement, subject to obtaining any required clearance of this Agreement under the HSR Act;
- 15.1.3. **Duly Authorized.** The execution, delivery and performance of this Agreement have been duly authorized by all necessary corporate action on its part;
- 15.1.4. **Legal Right.** It has the legal right and power to enter into this Agreement and to fully perform its obligations hereunder;
- 15.1.5. **No Conflicts.** The performance of its obligations under this Agreement will not conflict with such Party's charter documents or any agreement, contract or other arrangement to which such Party is a party; and
- 15.1.6. **Reasonable Commercial Practices.** It follows reasonable commercial practices common in the industry to protect its proprietary and confidential information, including requiring its employees, consultants and agents to be bound in writing by obligations of confidentiality and non-disclosure, and requiring its employees, consultants and agents to assign to it any and all inventions and discoveries discovered by such employees (in each case to the extent permitted by applicable laws), consultants or agents made within the scope of, and during their employment, and only disclosing proprietary and confidential information to Third Parties pursuant to written confidentiality and non-disclosure agreements.
- 15.2. **Adaptimmune Additional Representations and Warranties**. Adaptimmune also represents and warrants to GNE, as of the Execution Date, that:
- 15.2.1. **Adaptimmune Authority to Grant.** Adaptimmune has the full right and authority to grant all of the rights and licenses granted and purported to be granted to GNE hereunder, and neither Adaptimmune nor its Affiliates have granted any right or license, or committed to grant any right or license, to any Third Party relating to any of the Adaptimmune Licensed IP that would conflict with or limit the scope of any of the rights or licenses granted to GNE hereunder;

- 15.2.2. **List of Patents.** Exhibit C (Adaptimmune Licensed Patent Rights) sets forth a complete and accurate list of all Patents (including all pending Patent applications) existing as of the Execution Date that are Controlled by Adaptimmune and that Cover the Adaptimmune Differentiation Platform or any of the activities as contemplated within the Research Plan as of the Execution Date;
- 15.2.3. **Control of Patents.** Adaptimmune Controls all Patents listed in <u>Exhibit C</u> (Adaptimmune Licensed Patent Rights);
- 15.2.4. **Prosecution and Maintenance of Patents.** All issued Patents listed in Exhibit C (Adaptimmune Licensed Patent Rights) have been Prosecuted and Maintained by or on behalf of Adaptimmune in good faith, are in full force and effect and, to the Knowledge of Adaptimmune, are valid and enforceable;
- 15.2.5. **Pending Applications.** The pending applications listed in <u>Exhibit C</u> (Adaptimmune Licensed Patent Rights) are being Prosecuted and Maintained in accordance with applicable law, and, to its Knowledge, Adaptimmune has presented all relevant references, documents and information of which it and the inventors are aware to the relevant patent examiners and patent offices that are required to be so submitted under applicable law;
- 15.2.6. **Existing Upstream License Agreements.** Adaptimmune is either (a) the sole and exclusive owner, or (b) the licensee under the Existing Upstream License Agreements set out in **Schedule 1.51** (Existing Upstream License Agreements), cumulatively (a) and (b) of the Adaptimmune Licensed IP. All Affiliates of Adaptimmune have exclusively licensed or assigned all of their rights, title and interests in and to the Adaptimmune Licensed IP to Adaptimmune. Neither Adaptimmune nor any of its Affiliates has granted any mortgage, pledge, claim, security interest, lien or other charge of any kind on or in the Adaptimmune Licensed IP, and the Adaptimmune Licensed IP is free and clear of any mortgage, pledge, claim, security interest, lien or charge of any kind;
- 15.2.7. **Misappropriation of Trade Secret or Know-How.** To its Knowledge, the Adaptimmune Licensed IP does not include any misappropriated trade secret or other misappropriated Know-How of a Third Party;
- 15.2.8. **Inventors and Ownership.** Adaptimmune and its Affiliates have obtained, from all individuals who are identified as an inventor of an invention claimed in any Adaptimmune Licensed Patent Right, effective written assignments of all ownership rights of such individuals in such Adaptimmune Licensed Patent Right and, to the Knowledge of Adaptimmune, no Person who claims to be an inventor of an invention claimed in an Adaptimmune Licensed Patent Right is not identified as an inventor of such invention in the filed patent documents for such Adaptimmune Licensed Patent Right;
- 15.2.9. **Notice or Written Threat of Litigation.** Adaptimmune has not received any written notice of a claim or written threat of a claim or litigation made by any Person against Adaptimmune or its Affiliates that alleges that any Adaptimmune Licensed Patent Right necessary or useful to any activities contemplated in the Research Plan as of the Execution Date or Covers the Adaptimmune Differentiation Platform is invalid or unenforceable;

- 15.2.10. **Inventions and Government Relation.** The inventions claimed by Patents within the Adaptimmune Platform IP (a) were not conceived, discovered, developed or otherwise made in connection with any research activities funded, in whole or in part, by the federal government of the United States or any agency thereof; (b) are not a "subject invention" as that term is described in 35 USC §201(e); (c) are not otherwise subject to the provisions of the Patent and Trademark Law Amendments Act of 1980, as amended, codified at 35 USC §\$200-212, as amended, or any regulations promulgated pursuant thereto, including in 37 C.F.R. Part 401; and (d) are not the subject of any licenses, options or other rights of any governmental authority, within or outside the United States;
- 15.2.11. **Data Room.** To Adaptimmune's Knowledge, the information included in the electronic data room for the purpose of facilitating the transaction contemplated under this Agreement do not contain any untrue statement(s) of fact;
- 15.2.12. Adaptimmune Cell Line Agreement and Upstream License Agreement. Adaptimmune has provided GNE true, correct and complete copies of each Adaptimmune Cell Line Agreement and Existing Upstream License Agreement (in redacted form as applicable);

```
15.2.13. [***]; and 15.2.14. [***].
```

- 15.3. **GNE Additional Representations and Warranties**. GNE also represents and warrants to Adaptimmune as of the Execution Date that:
- 15.3.1. **Legal Right.** It has the legal right and power to extend the rights and licenses granted to Adaptimmune hereunder.

- 15.4. **Additional Adaptimmune Covenants.** Adaptimmune hereby covenants to GNE beginning on the Execution Date through the remainder of the Term that:
- 15.4.1. **Option, Right, or License to Third Party.** Neither Adaptimmune nor its Affiliates will grant any option, right or license to any Third Party relating to any of the intellectual property rights it Controls (including the Adaptimmune Licensed IP), or otherwise with respect to any Licensed Product, which is inconsistent with, or limits the scope of, any of the rights or licenses granted to GNE hereunder;
- 15.4.2. **Assignment.** Except as otherwise expressly permitted under this Agreement, Adaptimmune will not, and will cause its Affiliates not to assign, transfer, convey, encumber (through a lien, charge, security interest, mortgage or similar encumbrance) or dispose of, or enter into any agreement with any Third Party to assign, transfer, convey, encumber (through a lien, charge, security interest, mortgage or similar encumbrance) or dispose of, any Adaptimmune Licensed IP, any Adaptimmune Background IP or any Collaboration IP owned by Adaptimmune, in each case licensed to GNE hereunder, except to the extent that such assignment, transfer, conveyance, encumbrance, disposition or agreement is not inconsistent with, or does not limit the

scope of, any of the rights or licenses granted to GNE hereunder (by way of example, an encumbrance that was subject to the license would not be inconsistent with, or would limit the scope of, this Agreement);

- 15.4.3. **Debarment.** Neither Adaptimmune nor any of its Affiliates shall Knowingly employ, or otherwise use in any capacity, the services of any Person suspended, proposed for debarment or debarred under United States law, including under 21 USC § 335a, or any foreign equivalent thereof, with respect to the performance of activities hereunder, including to perform Adaptimmune's obligations under the Research Program;
- 15.4.4. **Termination of Licenses.** Adaptimmune shall not either (a) terminate, or (b) breach any Existing Upstream License Agreement or Adaptimmune Third Party IP License in a manner that would permit the counterparty thereto to terminate such Existing Upstream License Agreement or Adaptimmune Third Party IP License or otherwise diminish the scope or exclusivity of the licenses granted to GNE under any Adaptimmune Licensed IP exclusively licensed to GNE hereunder;
- 15.4.5. **New Developments.** Prior to the Effective Date, Adaptimmune will provide GNE an updated letter reflecting any new developments prior to the Effective Date that, if such developments had occurred prior to the Execution Date, would have been required to be disclosed against the representations in Section 15.2 (Adaptimmune Additional Representations and Warranties) pursuant to a disclosure letter; and
- Adaptimmune or its Affiliates under any Adaptimmune Cell Line Agreement or Existing Upstream License Agreement, where termination of such Adaptimmune Cell Line Agreement or Existing Upstream License Agreement or any diminishment of the use of the applicable Cell Line in the Research Program or scope or exclusivity of the licenses granted to GNE under the Adaptimmune Licensed IP exclusively licensed to GNE hereunder is being or could be sought by the counterparty or result from such breach, then Adaptimmune will promptly, but in no event less than three (3) Business Days thereafter, provide written notice thereof to GNE; and if (a) Adaptimmune has been finally determined to have breached such agreement or if Adaptimmune has agreed that it is in breach of such agreement, and (b) Adaptimmune has not cured such breach within the time period such agreement allows for cure, then GNE will have the right (but not the obligation) to: (i) cure such alleged breach to the extent possible; and (ii) offset any amounts paid by GNE to the counterparty of such Adaptimmune Cell Line Agreement to cure such breach against any payments due or that may become due under this Agreement.
- 15.5. **Additional GNE Covenants.** GNE hereby covenants to Adaptimmune beginning on the Execution Date through the remainder of the Term that:
- 15.5.1. **Option, Right, or License to Third Party**. Neither GNE nor its Affiliates will grant any option, right or license to any Third Party under any GNE Licensed IP, any GNE Platform Improvement IP invented solely by Adaptimmune or jointly by both Parties or any GNE Process Know-How, in each case that is inconsistent with, or limits the scope of, any of the rights or licenses granted to Adaptimmune hereunder;

15.5.2. **Assignment.** Except as otherwise expressly permitted under this Agreement, GNE will not, and will cause its Affiliates not to assign, transfer, convey, encumber (through a lien, charge, security interest, mortgage or similar encumbrance) or dispose of, or enter into any agreement with any Third Party to assign, transfer, convey, encumber (through a lien, charge, security interest, mortgage or similar encumbrance) or dispose of, any GNE Licensed IP, any GNE Platform Improvement IP invented solely by Adaptimmune or jointly by both Parties or any GNE Process Know-How, in each case licensed to Adaptimmune hereunder, except to the extent that such assignment, transfer, conveyance, encumbrance, disposition, or agreement is not inconsistent with, or does not limit the scope of, any of the rights or licenses granted to Adaptimmune hereunder (by way of example, an encumbrance that was subject to the license would not be inconsistent with, or would limit the scope of, this Agreement);

15.5.3. [***].

15.6. **Disclaimers**. EXCEPT AS OTHERWISE EXPRESSLY STATED IN THIS AGREEMENT, NEITHER PARTY MAKES ANY REPRESENTATION OR WARRANTY OF ANY KIND WITH RESPECT TO PATENTS, KNOW-HOW, MATERIALS OR CONFIDENTIAL INFORMATION SUPPLIED BY IT TO THE OTHER PARTY HEREUNDER, AND EXPRESSLY DISCLAIMS ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT.

ARTICLE 16 INDEMNIFICATION

16.1. **Indemnification**.

- 16.1.1. **Indemnification of Adaptimmune**. GNE shall indemnify Adaptimmune, its Affiliates and their respective directors, officers, employees, and agents, and defend and save each of them harmless, from and against any and all losses, damages, liabilities, costs, and expenses (including reasonable attorneys' fees and expenses) (collectively, "**Losses**") in connection with any and all suits, investigations, claims, or demands of Third Parties (collectively, "**Third Party Claims**") arising from or occurring as a result of:
 - (a) the breach by GNE of its obligations under this Agreement;
- (b) the breach of any of the warranties or representations made by GNE to Adaptimmune under this Agreement;
- (c) the negligence or wilful misconduct on the part of GNE or its Affiliates or their respective directors, officers, employees, and agents in performing its or their obligations under this Agreement; or
 - (d) [***]; or
 - (e) [***];

except in the case of clauses (a) through (e) above, for those Losses for which Adaptimmune has an obligation to indemnify GNE pursuant to <u>Section 16.1.2</u> (Indemnification of GNE), as to which Losses each Party shall indemnify the other to the extent of their respective liability for the Losses.

- 16.1.2. **Indemnification of GNE**. Adaptimmune shall indemnify GNE, its Affiliates and their respective directors, officers, employees, and agents, and defend and save each of them harmless, from and against any and all Losses in connection with any and all Third Party Claims arising from or occurring as a result of:
 - (a) the breach by Adaptimmune of its obligations under this Agreement;
- (b) the breach of any of the warranties or representations made by Adaptimmune to GNE under this Agreement;
- (c) the negligence or wilful misconduct on the part of Adaptimmune or its Affiliates or its or their respective directors, officers, employees, and agents in performing its obligations under this Agreement; or
 - (d) [***];

except, in the case of clauses (a) through (d) above, for those Losses for which GNE has an obligation to indemnify Adaptimmune pursuant to <u>Section 16.1.1</u> (Indemnification of Adaptimmune), as to which Losses each Party shall indemnify the other to the extent of their respective liability for the Losses.

Procedure. If a Party intends to claim indemnification under this Agreement (the "Indemnitee"), it shall promptly notify the other Party (the "Indemnitor") in writing of such alleged Loss. The Indemnitor shall have the right to control the defence thereof with counsel of its choice as long as such counsel is reasonably acceptable to Indemnitee. Any Indemnitee shall have the right to retain its own counsel at its own expense for any reason; provided, that if the Indemnitee shall have reasonably concluded, based upon a written opinion from outside legal counsel, that there is a conflict of interest between the Indemnitor and the Indemnitee in the defence of such action, in each of which cases the Indemnitor shall pay the reasonable fees and expenses of one law firm serving as counsel for the Indemnitee. The Indemnitee, and its employees and agents, shall reasonably cooperate with the Indemnitor and its legal representatives in the investigation of any Third Party Claims covered by this Agreement. The obligations of this Article 16 (Indemnification) shall not apply to any settlement of any Third Party Claims if such settlement is effected without the consent of both Parties, which shall not be unreasonably withheld or delayed. The failure to deliver written notice to the Indemnitor within a reasonable time after the commencement of any such action, to the extent prejudicial to its ability to defend such action, shall relieve the Indemnitor of any obligation to the Indemnitee under this Section 16.2 (Procedure). It is understood that only GNE and Adaptimmune may claim indemnity under this Agreement (on its own behalf or on behalf of its Indemnitees), and other Indemnitees may not directly claim indemnity hereunder.

16.3. **Insurance**.

- 16.3.1. **Evidence of Insurance.** Within [***] days of signing this Agreement, each Party shall provide the other Party with its certificate of insurance evidencing the insurance coverage set forth in <u>Section 16.3.2</u> (Insurance Coverage). Each Party shall provide to the other Party at least [***] prior written notice of any cancellation, non-renewal or material change in any of such insurance coverage.
- 16.3.2. **Insurance Coverage**. Subject to Section 16.3.4 (Election to Self-Insure), each Party shall obtain and maintain from an insurance company having an A.M. Best's rating of "A-, VII" or better comprehensive general liability insurance customary in the industry for companies of similar size conducting similar business, and in any case sufficient to cover its obligations.

16.3.3. **Product/Clinical Trial Liability Insurance**.

- (a) Commencing not later than thirty (30) days prior to the first use in humans of the first Licensed Product by GNE, its Affiliate or any of its Sublicensees, GNE shall have and maintain such type and amounts of products/clinical trial liability insurance covering the Development, Manufacture, use and sale of Licensed Products as is normal and customary in the industry generally for parties similarly situated, but, in any event, with a minimum combined single limit per occurrence for products/clinical trials liability as follows: (i) [***] for any period during which GNE, its Affiliates or any of its Sublicensees is conducting a clinical trial(s) with any Licensed Product(s); and (ii) [***] for any period during which GNE, its Affiliates or any of its Sublicensees is selling any Licensed Product(s).
- (b) Commencing not later than thirty (30) days prior to the first use in humans of the first Licensed Product by GNE, its Affiliates or any of its Sublicensees in which the applicable Licensed Product will be Manufactured by Adaptimmune, Adaptimmune shall have and maintain such type and amounts of products liability insurance covering the Manufacture, and use of Licensed Products as is normal and customary in the industry generally for parties similarly situated, but, in any event, with a minimum combined single limit per occurrence for products liability of [***] for any period during which GNE, its Affiliates or any of its Sublicensees is conducting a clinical trial for which Adaptimmune is Manufacturing Licensed Product(s).
 - (c) Each of the above insurance policies shall be primary insurance.
- 16.3.4. **Election to Self-Insure**. If either Party is an entity which, together with its Affiliates, has worldwide revenues from [***] per year, the obligations set forth in Sections 16.3.1 (Evidence of Insurance) through 16.3.3 (Product/Clinical Trial Liability Insurance) above shall not apply with respect to such Party, if such Party notifies the other Party in writing that it elects to provide coverage through a commercially reasonable program of self-insurance; provided, that the obligations set forth in Sections 16.3.1 (Evidence of Insurance) through Section 16.3.3 (Product/Clinical Trial Liability Insurance) shall resume with respect to such Party and its Affiliates, or successor-in-interest and its Affiliates, if such program of self-insurance is terminated or discontinued for any reason.

16.4. **Limitation of Damages**. NEITHER PARTY HERETO WILL BE LIABLE FOR INDIRECT, INCIDENTAL, CONSEQUENTIAL, SPECIAL, EXEMPLARY, OR PUNITIVE DAMAGES (INCLUDING LOST PROFITS), IN EACH CASE ARISING FROM OR RELATING TO THIS AGREEMENT, REGARDLESS OF ANY NOTICE OF SUCH DAMAGES, EXCEPT IN RESPECT OF ANY BREACH OF A PARTY'S OBLIGATIONS UNDER <u>ARTICLE 13</u> (CONFIDENTIALITY) OR INDEMNIFICATION OBLIGATIONS UNDER THIS <u>ARTICLE 16</u> (INDEMNIFICATION) FOR CLAIMS OF THIRD PARTIES.

ARTICLE 17 TERM; TERMINATION

- 17.1. **Term**. The term of this Agreement (the "**Term**") shall commence on the Effective Date and, unless sooner terminated as provided in this <u>Article 17</u> (Term; Termination) shall continue in full force and effect, on a country-by-country and Licensed Product-by-Licensed Product basis until there is no remaining royalty payment or other payment obligation in such country with respect to such Licensed Product under <u>Article 10</u> (Financial Terms), at which time this Agreement shall expire with respect to such Licensed Product in such country. The Term shall expire on the date this Agreement has expired in its entirety with respect to all Licensed Products in all countries in the Territory.
- Termination by Either Party for Material Breach. Either Party may terminate this Agreement by written notice to the other Party for any material breach of this Agreement by the other Party (provided, however, that, if such Party's rights under this Agreement that are materially and adversely affected by such material breach are reasonably specific to a given Licensed Product, Collaboration Target, country or region, such Party shall have the right to terminate this Agreement solely with respect to such Licensed Product, Collaboration Target, country or region, and not in its entirety) if, in the case of remediable breach, such material breach is not cured within [***] after the breaching Party receives written notice of such breach from the non-breaching Party; provided further, that if such breach is not capable of being cured [***] period, the cure period shall be extended for such amount of time that the Parties may agree in writing is reasonably necessary to cure such breach (not to exceed an additional [***] unless otherwise agreed by the Parties), so long as: (a) the breaching Party is making diligent efforts to do so; and (b) the Parties agree on an extension within such [***] period. Notwithstanding anything to the contrary herein, if the allegedly breaching Party in good faith either disputes: (i) whether a breach is material or has occurred; or (ii) the alleged failure to cure or remedy such material breach, and provides written notice of that dispute to the other Party within the above time periods, then the matter will be addressed under the dispute resolution provisions in Article 19 (Dispute Resolution), and the notifying Party may not so terminate this Agreement until it has been determined under Article 19 (Dispute Resolution) that the allegedly breaching Party is in material breach of this Agreement, and such breaching Party further fails to cure such breach within [***] (or such longer period as determined by the arbiter of such dispute resolution) after the conclusion of that dispute resolution procedure.
- 17.3. **Termination by Either Party for Insolvency or Bankruptcy**. Either Party may terminate this Agreement effective on written notice to the other Party upon the liquidation, dissolution, winding-up, insolvency, bankruptcy, or filing of any petition therefor, appointment of a receiver,

custodian or trustee, or any other similar proceeding, by or of the other Party where such petition, appointment or similar proceeding is not dismissed or vacated within ninety (90) days. All rights and licenses granted pursuant to this Agreement are, for purposes of Section 365(n) of Title 11 of the United States Code or any foreign equivalents thereof (as used in this Section 17.3 (Termination by Either Party for Insolvency or Bankruptcy), "Title 11"), licenses of rights to "intellectual property" as defined in Title 11. Each Party in its capacity as a licensor hereunder agrees that, in the event of the commencement of bankruptcy proceedings by or against such bankrupt Party under Title 11, (a) the other Party, in its capacity as a licensee of rights under this Agreement, shall retain and may fully exercise all of such licensed rights under this Agreement (including as provided in this Section 17.3 (Termination by Either Party for Insolvency or Bankruptcy)) and all of its rights and elections under Title 11; and (b) the other Party shall be entitled to a complete duplicate of all embodiments of such intellectual property, and such embodiments, if not already in its possession, shall be promptly delivered to the other Party: (i) upon any such commencement of a bankruptcy proceeding, unless the bankrupt Party elects to continue to perform all of its obligations under this Agreement; or (ii) if not delivered under (i), immediately upon the rejection of this Agreement by or on behalf of the bankrupt Party.

17.4. **GNE Termination**.

17.4.1. **Termination for Convenience**. GNE shall also have the right to terminate this Agreement, in its sole discretion, on a Licensed Product-by-Licensed Product, Collaboration Target-by-Collaboration Target, country-by-country, or region-by-region basis, or in its entirety, at any time by providing written notice to Adaptimmune. Such termination will be effective [***] after such notice.

```
17.4.2. [***].
17.5. [***].
17.5.1. [***].
17.5.2. [***].
```

17.6. Effects of Termination.

17.6.1. Accrued Rights and Obligations. Expiration or termination of this Agreement for any reason shall not release either Party from any liability that, as of the effective date of such expiration or termination, had already accrued to the other Party or that is attributable to a period prior to such expiration or termination, nor preclude either Party from pursuing any rights and remedies it may have hereunder or at law or in equity that accrued or are based upon any event occurring prior to the effective date of such expiration or termination, and any such termination shall not be an exclusive remedy for any liability accrued hereunder. Notwithstanding the foregoing, in the event this Agreement is terminated with respect to all Collaboration Personalised T-Cell Therapies and any payment obligation under Section 10.3.1 ([***] Payments) becomes due between the provision of notice of such termination and the effective date of such termination, then, for such payment, GNE shall be obligated to pay Adaptimmune only an amount equal to such milestone payment amount multiplied by the fraction, the numerator of which is the

number of days between the date such payment is due and the effective date of such termination, and the denominator of which is three hundred-sixty-five (365).

17.6.2. **Termination of Licenses**. Subject to <u>Section 10.8.6</u> (Rights Following Expiration of Royalty Term) and <u>Section 17.6.3</u> (Continuation of Sublicenses), upon termination of this Agreement in its entirety, on a Licensed Product-by-Licensed Product, Collaboration Target-by-Collaboration Target, country-by-country, or region-by-region basis, all licenses under this Agreement that are not perpetual and irrevocable and that are affected by such termination (on a Licensed Product-by-Licensed Product, Collaboration Target-by-Collaboration Target, country-by-country or region-by-region basis, as applicable) shall terminate as of the effective date of such termination. By way of example but not limitation, if the licenses to one Collaboration Target in the US are terminated, (a) the licenses to such Collaboration Target in the rest of the world and (b) the licenses to all other Collaboration Targets, in each case ((a) and (b)), shall survive such termination.

17.6.3. Continuation of Sublicenses. [***].

- Return of Confidential Information. Subject to the foregoing, following expiry or any 17.6.4. early termination of this Agreement, the Party that has Confidential Information of the other Party shall destroy (at such Party's written request): (a) all such Confidential Information in its possession as of the effective date of expiration or early termination (with the exception of one copy of such Confidential Information, which may be retained by the legal department of the Party that received such Confidential Information to confirm compliance with the non-use and non-disclosure provisions of this Agreement); and (b) any Confidential Information of the other Party contained in its laboratory notebooks or databases; provided, that (with respect to both clauses (a) and (b)) each Party may retain and continue to use such Confidential Information of the other Party to the extent necessary to exercise any surviving rights, licenses or obligations under this Agreement; provided, further, that (with respect to both clauses (a) and (b)) such Confidential Information of the other Party existing on any backup, back-end, or archiving system, or in electronic files of such Party that are not reasonably accessible, and which cannot be reasonably deleted from such systems or files, may be retained by such Party, provided that confidentiality is maintained in accordance with this Agreement. Following termination of this Agreement, upon the request of a Party, the other Party will confirm in writing its compliance with the obligations under this Section 17.6.4 (Return of Confidential Information).
- 17.6.5. **Inventory at Termination**. Upon termination of this Agreement, GNE, its Affiliates and its Sublicensees shall have the right to sell or otherwise dispose of all inventory of affected Licensed Products in all affected countries then in its stock, subject to the applicable royalty payments due under this Agreement and any other applicable provisions of this Agreement, and Adaptimmune covenants not to sue GNE, its Affiliates or its Sublicensees for infringement under any of the Patents that were licensed by Adaptimmune to GNE immediately prior to such termination with respect to such activities conducted by GNE, its Affiliates or its Sublicensees pursuant to this Section 17.6.5 (Inventory at Termination).
- 17.6.6. **Effects of Certain Terminations in Relation to Opt-In Collaboration Targets.** In the event of termination of this Agreement by Adaptimmune pursuant to Section 17.2

(Termination by Either Party for Material Breach) or <u>Section 17.5</u> (Termination for Discontinuation Event) or by GNE pursuant to <u>Section 17.4</u> (GNE Termination), in addition to the other terms set forth in <u>Section 17.6</u> (Effects of Termination), including those provisions surviving under <u>Section 17.6.7</u> (Survival), upon such termination the following terms of this <u>Section 17.6.6</u> (Effects of Certain Terminations in Relation to Opt-In Collaboration Targets) shall apply with respect to any Relevant Collaboration Target. As used herein, "**Relevant Collaboration Target**" means a Collaboration Target (i) for which an Opt-In is then in effect, and (ii) that is the subject of any of the termination events described in the foregoing sentence.

- (a) For each Relevant Collaboration Target, [***].
- (b) [***]
- (c) [***].
- (d) [***].
 - (i) [***]
 - (ii) [***]
 - (iii) [***]
- (e) [***]
- (f) [***]
- (g) [***].
- (h) [***].

17.6.7. **Survival.** In addition to any provisions specified in this Agreement as surviving under the applicable circumstances, the provisions of <u>Article 1</u> (Definitions), <u>Article 13</u> (Confidentiality), <u>Article 14</u> (Publicity; Publications; Use of Name), <u>Article 15</u> (Representations, Warranties and Covenants), <u>Article 16</u> (Indemnification) (*provided*, that with respect to <u>Article 15</u> (Representations, Warranties and Covenants) and <u>Article 16</u> (Indemnification), only with respect to those claims that arise from the acts or omissions of a Party prior to the effective date of termination or expiration) <u>Article 19</u> (Dispute Resolution) and <u>Article 20</u> (Miscellaneous) and <u>Section 3.5.2</u> (Adaptimmune Permitted Uses of Research Program Cell Lines), <u>Section 12.1</u> (Definitions and Ownership), <u>Section 12.2.2</u> (Inventorship), <u>Section 12.2.3</u> (Assignment; Cooperation), <u>Section 12.2.4</u> (Joint Ownership), <u>Section 12.2.5</u> (CREATE Act), <u>Section 12.3.1</u> (Sole IP), <u>Section 12.3.2</u> (Joint Collaboration IP), <u>Section 12.8</u> (Common Interest Disclosures), <u>Section 17.1</u> (Term) and <u>Section 17.6</u> (Effects of Termination) shall survive any termination or expiration of this Agreement. In addition, <u>Article 10</u> (Financial Terms), <u>Article 11</u> (Payment Terms; Reports; Audits), and <u>Section 12.4.4</u> (Costs and Expenses), <u>Section 12.4.5</u> (Damages), and <u>Section 12.5.5</u> (Costs and Expenses) shall survive with respect to any outstanding unpaid amounts that accrued prior to any termination or expiration of this Agreement.

ARTICLE 18 HSR FILING; TERMINATION UPON HSR DENIAL

If GNE determines that an HSR Filing is necessary, each Party shall, within ten (10) Business Days of the Execution Date (or such later time as may be agreed to in writing by the Parties), file with the United States Federal Trade Commission and the Antitrust Division of the United States Department of Justice, or with equivalent foreign authorities, any HSR Filing required of it under the HSR Act or applicable antitrust or competition laws of other jurisdictions with respect to the transactions contemplated hereby. The Parties shall seek expedited treatment of any HSR Filing unless otherwise agreed by the Parties in writing. Each Party will use reasonable efforts to do, or cause to be done, all things necessary or advisable to, as promptly as practicable, take all actions necessary to make the filings required of such Party or its Affiliates under the HSR Act and obtain the requisite Governmental Required Consents. The Parties shall cooperate with one another to the extent necessary in the preparation of any such HSR Filing. GNE shall be responsible for the filing fees associated with any HSR Filing. Each Party shall be responsible for all of its other own costs and expenses associated with any HSR Filing, including its own attorneys' fees and associated costs and expenses. If the Parties make an HSR Filing under this Agreement, then this Agreement shall terminate: (i) at the election of either Party, immediately upon written notice to the other Party, if the US Federal Trade Commission or the US Department of Justice, or an equivalent foreign authority, seeks a preliminary injunction under the applicable antitrust laws against the Parties to enjoin the transactions contemplated by this Agreement; or (ii) at the election of either Party, immediately upon written notice to the other Party, in the event that the HSR Clearance Date shall not have occurred on or prior to one hundred eighty (180) days after the effective date of the HSR Filing. In the event of such termination, this Agreement shall be of no further force and effect.

ARTICLE 19 DISPUTE RESOLUTION

19.1. **Disputes**. Adaptimmune and GNE recognize that a dispute, controversy or claim of any nature whatsoever arising out of or relating to this Agreement, or the breach, termination or invalidity thereof, (each, a "**Dispute**") may from time to time arise during the Term. Except as otherwise provided in this Agreement (including as otherwise expressly provided with respect to audit disputes in <u>Section 11.7.5</u> (Audit Dispute) and JRC disputes as set forth in <u>Article 2</u> (Governance)), such Disputes between Adaptimmune and GNE will be resolved as recited in this <u>Article 19</u> (Dispute Resolution). In the event of the occurrence of such a Dispute, the Parties shall first refer such Dispute to their respective Alliance Managers for attempted resolution by such Alliance Managers within [***] days after such referral. If such Dispute is not resolved within such [***] day period, either Adaptimmune and GNE may, by written notice to the other, have such Dispute referred to their respective officers designated below, or their respective designees, for attempted resolution within [***] days after such notice is received. Such designated officers are as follows:

For GNE – A Vice President

For Adaptimmune – Chief Executive Officer

In the event the designated officers, or their respective designees, are not able to resolve such dispute within [***] days of such other Party's receipt of such written notice, either Party may initiate the dispute resolution procedures set forth in Section 19.2 (Arbitration).

19.2. **Arbitration**.

- 19.2.1. **Rules**. Except as otherwise expressly provided in this Agreement (including under Section 19.3 (Subject Matter Exclusions)), the Parties agree that any Dispute not resolved internally by the Parties pursuant to Section 19.1 (Disputes) shall be resolved through binding arbitration conducted by the International Chamber of Commerce in accordance with the then prevailing Rules of Arbitration of the International Chamber of Commerce (" **Rules**"), except as modified in this Agreement, applying the substantive law specified in Section 20.1 (Applicable Law).
- 19.2.2. **Arbitrators; Location**. Each Party shall select one (1) arbitrator, and the two (2) arbitrators so selected shall choose a third arbitrator. All three (3) arbitrators shall serve as neutrals and have at least ten (10) years of: (a) dispute resolution experience (including judicial experience) or (b) legal or business experience in the biotech or pharmaceutical industry. In any event, at least one (1) arbitrator shall satisfy the foregoing experience requirement under clause (b). If a Party fails to nominate its arbitrator, or if the Parties' arbitrators cannot agree on the third, the necessary appointments shall be made in accordance with the Rules. Once appointed by a Party, such Party shall have no *ex parte* communication with its appointed arbitrator. The arbitration proceedings shall be conducted in [***]. The arbitration proceedings and all pleadings and written evidence shall be in the English language. Any written evidence originally in another language shall be submitted in English translation accompanied by the original or a true copy thereof.
- 19.2.3. **Procedures; Awards**. Each Party agrees to use reasonable efforts to make all of its current employees available, if reasonably needed, and agrees that the arbitrators may determine any person as necessary. The arbitrators shall be instructed and required to render a written, binding, non-appealable resolution and award on each issue that clearly states the basis upon which such resolution and award is made. The written resolution and award shall be delivered to the Parties as expeditiously as possible, but in no event more than ninety (90) days after conclusion of the hearing, unless otherwise agreed by the Parties. Judgment upon such award may be entered in any competent court or application may be made to any competent court for judicial acceptance of such an award and order for enforcement. Each Party agrees that, except as permitted by Section 16.4 (Limitation of Damages), notwithstanding any provision of applicable law or of this Agreement, it will not request, and the arbitrators shall have no authority to award, punitive or exemplary damages against any Party.
- 19.2.4. **Costs**. The prevailing Party, as determined by the arbitrators, shall be entitled to: (a) its share of fees and expenses of the arbitrators; and (b) its attorneys' fees and associated costs and expenses. In determining which Party "prevailed," the arbitrators shall consider: (i) the significance, including the financial impact, of the claims prevailed upon; and (ii) the scope of claims prevailed upon, in comparison to the total scope of the claims at issue. If the arbitrators determine that, given the scope of the arbitration, neither Party "prevailed," the arbitrators shall

order that the Parties: (x) share equally the fees and expenses of the arbitrators; and (y) bear their own attorneys' fees and associated costs and expenses.

- 19.2.5. **Interim Equitable Relief**. Notwithstanding anything to the contrary in this <u>Section 19.2</u> (Arbitration), in the event that a Party reasonably requires relief on a more expedited basis than would be possible pursuant to the procedure set forth in this <u>Article 19</u> (Dispute Resolution), such Party may seek a temporary injunction or other interim equitable relief in a court of competent jurisdiction pending the ability of the arbitrators to review the decision under this <u>Section 19.2</u> (Arbitration). Such court shall have no jurisdiction or ability to resolve Disputes beyond the specific issue of temporary injunction or other interim equitable relief.
- 19.2.6. **Protective Orders; Arbitrability**. At the request of either Party, the arbitrators shall enter an appropriate protective order to maintain the confidentiality of information produced or exchanged in the course of the arbitration proceedings. The arbitrators shall have the power to decide all questions of arbitrability.
- 19.3. **Subject Matter Exclusions**. Notwithstanding the provisions of Section 19.2 (Arbitration), any Dispute not resolved internally by the Parties pursuant to Section 19.1 (Disputes) that involves the validity or infringement of a Patent Covering a Licensed Product that is issued in: (a) the United States shall be subject to actions before the United States Patent and Trademark Office or submitted exclusively to the federal court located in the jurisdiction of the district where any of the defendants resides; and (b) any other country (or region) shall be brought before an appropriate regulatory or administrative body or court in that country (or region), and the Parties hereby consent to the jurisdiction and venue of such courts and bodies.
- 19.4. **Continued Performance**. Provided that this Agreement has not terminated, the Parties agree to continue performing under this Agreement in accordance with its provisions, pending the final resolution of any Dispute.

ARTICLE 20 MISCELLANEOUS

- 20.1. **Applicable Law**. This Agreement (including the arbitration provisions of <u>Section 19.2</u> (Arbitration)) shall be governed by and interpreted in accordance with the laws of the State of New York, without reference to the principles of conflicts of laws. The United Nations Convention on Contracts for the International Sale of Goods shall not apply to the transactions contemplated by this Agreement.
- 20.2. **Notices**. Except as otherwise expressly provided in this Agreement, any notice required under this Agreement shall be in writing and shall specifically refer to this Agreement. Notices shall be sent via one of the following means and will be effective: (a) on the date of delivery, if delivered in person; or (b) on the date of receipt, if sent by private express courier or by first class certified mail, return receipt requested. Notices shall be sent to the other Party at the addresses set forth below. Either Party may change its addresses for purposes of this Section 20.2 (Notices) by sending written notice to the other Party. Notwithstanding the foregoing, notices required to be provided to a Party's Alliance Manager may be provided solely by email to such Alliance Manager's email address.

If to GNE: [***

with required copies (which shall not constitute notice) to:

[***

If to Adaptimmune: [***]

Assignment. Neither Party may assign or otherwise transfer, in whole or in part, this Agreement without the prior written consent of the non-assigning Party, such approval not to be unreasonably withheld or delayed. Notwithstanding the foregoing, either Party may assign this Agreement to: (a) an Affiliate that is sufficiently capitalized to perform its obligations under this Agreement; or (b) subject to Section 9.2 (Change of Control), any purchaser of all or substantially all of the assets of such Party (and in such event this Agreement must be assigned to such purchaser), or of all of its capital stock, or to any successor corporation or entity resulting from any merger or consolidation of such Party with or into such corporation or entity (whether arising under contract, by statute or at law), or otherwise in connection with a Change of Control of such Party; provided, that the party to which this Agreement is assigned expressly agrees in writing to assume and be bound by all obligations of the assigning Party under this Agreement. Any attempted assignment in contravention of this Section 20.3 (Assignment) shall be null and void. A copy of such written agreement by such assignee shall be provided to the non-assigning Party within ten (10) days of execution of such written agreement. Subject to the foregoing, this Agreement will benefit and bind the Parties' successors and assigns.

20.4. Certain Employees of the Other Party. [***].

20.5. Force Majeure. Neither Party will be held liable to the other Party nor be deemed to have defaulted under or breached this Agreement for failure or delay in achieving any objective, satisfying any condition, or performing any obligation under this Agreement to the extent that such failure or delay is caused by or results from acts or events beyond the reasonable control of such Party, including acts of God, embargoes, war, acts of war (whether war be declared or not), terrorism, insurrections, riots, civil commotions, strikes, lockouts, or other labour disturbances (other than strikes, lockouts, or labour disturbances involving a Party's own employees), government actions, fire, earthquakes, floods, epidemics, pandemics, the spread of infectious diseases, and quarantines (" Force Majeure"). The affected Party will notify the other Party in writing of any Force Majeure circumstances that may affect its performance under this Agreement as soon as reasonably practical, will provide a good faith estimate of the period for which its failure or delay in performance under this Agreement is expected to continue based on currently available information, and will undertake reasonable efforts necessary to mitigate and overcome such Force Majeure circumstances and resume normal performance of its obligations hereunder as soon a reasonably practicable under the circumstances . If the Force Majeure circumstance continues, then the affected Party will update such written notice to the other Party on a weekly basis, or more frequently if requested by the other Party, to provide updated summaries of its mitigation efforts and its estimates of when normal performance under this Agreement will be able to resume.

- 20.6. **Independent Contractors**. The Parties hereto are independent contractors and nothing contained in this Agreement shall be deemed or construed to create a partnership, joint venture, employment, franchise, agency or fiduciary relationship between the Parties.
- 20.7. **Integration**. Except to the extent expressly provided herein, this Agreement constitutes the entire agreement between the Parties relating to the subject matter of this Agreement and supersedes all previous oral and written communications between the Parties with respect to the subject matter of this Agreement (including the Non-Disclosure Agreement and term sheets exchanged by and between Adaptimmune and GNE).
- 20.8. **Amendment; Waiver**. Except as otherwise expressly provided herein, no alteration of or modification to this Agreement shall be effective unless made in writing and executed by an authorized representative of both Parties. No course of dealing or failing of either Party to strictly enforce any term, right or condition of this Agreement in any instance shall be construed as a general waiver or relinquishment of such term, right or condition. The observance of any provision of this Agreement may be waived (either generally or in any given instance and either retroactively or prospectively) only with the written consent of the Party granting such waiver.
- 20.9. **Further Assurance**. Each Party shall, and shall use reasonable efforts to procure that any necessary Third Party shall, promptly execute and deliver such further documents and do such further acts as may be required for the purpose of giving full effect to this Agreement.
- 20.10. **Severability**. The Parties do not intend to violate any public policy or statutory or common law. However, if any sentence, paragraph, clause or combination or part thereof of this Agreement is in violation of any law or is found to be otherwise unenforceable, such sentence, paragraph, clause or combination or part of the same shall be deleted and the remainder of this Agreement shall remain binding, *provided*, that such deletion does not alter the basic purpose and structure of this Agreement.
- 20.11. **No Third Party Rights.** The Parties do not intend that any term of this Agreement should be enforceable by any person who is not a Party.
- 20.12. **Construction**. The Parties mutually acknowledge that they and their attorneys have participated in the negotiation and preparation of this Agreement. Ambiguities, if any, in this Agreement shall not be construed against any Party, irrespective of which Party may be deemed to have drafted this Agreement or authorized the ambiguous provision.
- 20.13. **Actions of Affiliates.** Either Party may exercise its rights (including the licenses granted in <u>Section 7.1.1</u> (License Grants) and <u>Section 7.2.1</u> (License Grants)) or perform its obligations under this Agreement personally or through one or more Affiliates, *provided* that the applicable Party shall nonetheless be primarily liable for the performance of its Affiliates and for any failure by its Affiliates to comply with the restrictions, limitations and obligations set forth in this Agreement.
- 20.14. **Interpretation**. The captions and headings to this Agreement are for convenience only, and are to be of no force or effect in construing or interpreting any of the provisions of this Agreement. Unless context otherwise clearly requires, whenever used in this Agreement: (a) the

words "include" or "including" shall be construed as incorporating "but not limited to" or "without limitation"; (b) the words "hereof," "herein," "hereby" and derivative or similar words refer to this Agreement, including the exhibits; (c) the word "law" or "laws" means any applicable, legally binding statute, ordinance, resolution, regulation, code, guideline, rule, order, decree, judgment, injunction, mandate or other legally binding requirement of a governmental authority (including a court, tribunal, agency, legislative body or other instrumentality of any: (i) government or country or territory; (ii) state, province, county, city or other political subdivision thereof; or (iii) supranational body); (d) all references to the word "will" are interchangeable with the word "shall" and shall be understood to be imperative or mandatory in nature; (e) all references to "sublicensees" shall include all sublicensees of sublicensees through multiple tiers of sublicensing; (f) the singular shall include the plural and vice versa; (g) all references to "dollars" shall mean US dollars; and (h) the word "or" has the inclusive meaning represented by the phrase "and/or". All references to days, months, quarters or years are references to calendar days, calendar months, calendar quarters, or calendar years. Whenever any matter hereunder requires consent or approval, such consent shall not be unreasonably withheld or delayed. If anything in the Research Plan is inconsistent with this Agreement, the terms of this Agreement will control, unless stated otherwise in the Research Plan.

20.15. **Counterparts**. This Agreement may be executed in two or more counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument. For purposes hereof, a facsimile copy, or email with attached pdf copy, of this Agreement, including the signature pages hereto, will be deemed to be an original. Notwithstanding the foregoing, the Parties shall deliver original execution copies of this Agreement to one another as soon as practicable following execution thereof.

[Signature page follows – the rest of this page intentionally left blank.]

IN	WITNESS	WHEREOF,	Adaptimmune,	GNE a	and I	Roche	have	executed	this	Agreement	by	their
resi	pective office	ers hereunto du	ly authorized, on	the Exe	ecutio	on Date	e.					

ADAPTIMMUNE LIMITED	
By: /s/ Adrian Rawcliffe	
Name: Adrian Rawcliffe	
Title: <u>Director and CEO</u>	
GENENTECH, INC.	
By: /s/ Edward Harrington	
Name: Edward Harrington	
Title: <u>CFO</u>	
F. HOFFMANN-LA ROCHE LTD	
By: /s/ Vikas Kabra	By: /s/ Barbara Schroeder
Name: <u>Vikas Kabra</u>	Name: Barbara Schroeder
Title: Global Head Transaction Excellence	Title: _Authorized Signatory
[Signature Page to Genentech-Adaptimmus	ne Strategic Collaboration and License Agreement]

[***]

[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	
[***]	[***]	[***]	

Existing Upstream License Agreements

- 1. [***]
- 2. [***]

Initial Collaboration Targets

[***]

Specifications

[***]

[THIS PAGE HAS BEEN EXCLUDED FROM THIS EXHIBIT BECAUSE IT IS BOTH NOT MATERIAL AND IS THE TYPE THAT THE REGISTRANT TREATS AS PRIVATE OR CONFIDENTIAL]

[***]

Approved Subcontractors

[***]

[***]

Pre-Existing License Terms

[***]

Schedule 8.3.3

[***]

• [***]

• [***]

[***]

Schedule 10.4.1(a)

No Adaptimmune Opt-In

	Off-the-Shelf Development Milestone Event	Off-the-Shelf Development Milestone Payment (US Dollars)
	[***]	
[***]		[***]
[***]		[***]
[***]		[***]
	[***]	
[***]		[***]
[***]		[***]
[***]		[***]
	[***]	
[***]		[***]
[***]		[***]
[***]		[***]
[***]		[***]
[***]		[***]
[***]		[***]
[***]		[***]
[***]		[***]
[***]		[***]

[***].

Schedule 10.4.1(b)

Adaptimmune Opt-In

Off-the-Shelf Development Milestone Event	Off-the-Shelf Development Milestone Payment (US Dollars)
[***]	·
[***]	[***]
[***]	•
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

[***].

Schedule 10.4.2

Collaboration Personalised T-Cell Therapy Development Milestones

	Personalised Development Milestone Event	Milestone Payment (US Dollars)	
	[***]	<u>'</u>	
[***]		[***]	
	[***]		
[***]		[***]	
[***]		[***]	
[***]		[***]	
	[***]		
[***]		[***]	
[***]		[***]	
[***]		[***]	
[***]		[***]	
[***]		[***]	
[***]		[***]	
[***]		[***]	
 [***]		[***]	
[***]		[***]	

[***].

Press Release

[Attached]

EXHIBIT A

Adaptimmune Differentiation Platform

[***]

EXHIBIT B

Research Plan

[***]

EXHIBIT C

Adaptimmune Licensed Patent Rights

Title	Country	Application no.	Filing date	Adaptimmune reference
Methods of Producing	PCT	PCT/EP2020/073396	20 Aug 2020	ADAP-PC10101
Haemogenic Progenitor				
cells from Pluripotent Stem Cells				
Culture Medium for	PCT	PCT/EP2020/073407	20 Aug 2020	ADAP-PC10102
Haematopoietic Induction				
Lentiviral Transduction	PCT	PCT/EP2020/073403	20 Aug 2020	ADAP-PC10099
Methods				
Methods of T Cell	PCT	PCT/EP2020/073332	20 Aug 2020	ADAP-PC10103
Production				
T Cell Production from	PCT	PCT/EP2020/073400	20 Aug 2020	ADAP-PC10100
RAG Inactivated iPSCs				
Improved T Cell	PCT	PCT/GB2021/050909	16 Apr 2021	ADAP-PC10115
Manufacturing Process				
Modified iPSCs	PCT	PCT/GB2021/051125	11 May 2021	ADAP-PC10118
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]

EXHIBIT D

Adaptimmune Royalty Patent Rights

Adaptimmune Off-the-Shelf Royalty Patent Rights

Title	Country	Application no.	Filing date	Adaptimmune reference
Methods of Producing	PCT	PCT/EP2020/073396	20 Aug 2020	ADAP-PC10101
Haemogenic Progenitor				
cells from Pluripotent				
Stem Cells				
Culture Medium for	PCT	PCT/EP2020/073407	20 Aug 2020	ADAP-PC10102
Haematopoietic Induction				
Lentiviral Transduction	PCT	PCT/EP2020/073403	20 Aug 2020	ADAP-PC10099
Methods				
Methods of T Cell	PCT	PCT/EP2020/073332	20 Aug 2020	ADAP-PC10103
Production				
T Cell Production from	PCT	PCT/EP2020/073400	20 Aug 2020	ADAP-PC10100
RAG Inactivated iPSCs				
Improved T Cell	PCT	PCT/GB2021/050909	16 Apr 2021	ADAP-PC10115
Manufacturing Process				
Modified iPSCs	PCT	PCT/GB2021/051125	11 May 2021	ADAP-PC10118
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]

Adaptimmune Personalised Royalty Patent Rights

Title	Country	Application no.	Filing date	Adaptimmune reference
[***]	[***]	[***]	[***]	[***]

EXHIBIT E

Example US Net Profit and Net Loss Calculation in Section 10.9.1

USD	GNE	Adaptimmune	Total
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]

Certification Required by Rule 13a-14(a) or 15d-14(a) under the Securities Exchange Act of 1934 as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Adrian Rawcliffe, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Adaptimmune Therapeutics plc;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)), for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected or is reasonably likely to materially affect the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 4, 2021

/s/ Adrian Rawcliffe

Adrian Rawcliffe

Chief Executive Officer

Certification Required by Rule 13a-14(a) or 15d-14(a) under the Securities Exchange Act of 1934 as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Gavin Wood, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Adaptimmune Therapeutics plc;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)), for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected or is reasonably likely to materially affect the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 4, 2021

/s/ Gavin Wood

Gavin Wood

Chief Financial Officer

Section 906 Certificate

Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), I, Adrian Rawcliffe, Chief Executive Officer of Adaptimmune Therapeutics plc, a public limited company incorporated under English law (the "Company"), hereby certify, to my knowledge, that:

- The Company's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2021, to which this Certification is attached as Exhibit 32.1 (the "Quarterly Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Quarterly Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 4, 2021

/s/ Adrian Rawcliffe Adrian Rawcliffe Chief Executive Officer

Section 906 Certificate

Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), I, Gavin Wood, Chief Financial Officer of Adaptimmune Therapeutics plc, a public limited company incorporated under English law (the "Company"), hereby certify, to my knowledge, that:

- The Company's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2021, to which this Certification is attached as Exhibit 32.2 (the "Quarterly Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Quarterly Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 4, 2021

/s/ Gavin Wood

Gavin Wood

Chief Financial Officer